DATA VALIDATION AND RELIABILITY CALCULATIONS

IN DIGITAL PROTECTION SYSTEMS

by

Gustavo Enrique Gonzalez-Urdaneta, Ing. Elec.

A Thesis submitted for the Degree of Doctor of Philosophy in Engineering

Department of Electrical Engineering, Imperial College of Science and Technology, University of London.

September 1976

ABSTRACT

A desirable characteristic in a decision-making process based on any type of measurements is that all data required had been previously validated. Data validation techniques are analysed and a new algorithm is proposed for the detection in real time of bad data in a digital protection system. A simple estimation routine based on the redundancy of the measurement takes account of missing data to form a reliable relaying data base in a highvoltage substation.

Computer methods for reliability evaluation in digital protection systems are presented. The methods are, however, general and can also be used for reliability evaluation of other systems. The minimal cut-set approach to system reliability is presented and three new algorithms are proposed for the minimal cut-set generation process. All are based on the concepts of graph theory. The algorithms enumerate the set of minimal cuts and provide system reliability bounds for nonseparable, separable and acyclic directed graphs. The node failure problem is analysed and a branchnode cut-set algorithm is proposed which allows the analysis of node outage influence on the failure rates and outage durations of each node.

The random nature of component reliability data demands computation methods to take into account the uncertainty of component parameters. Two approaches for confidence limits of the probability and frequency of system failure are presented for a repairable system represented by its minimal cut-sets. The first based on a moment method gives the variance and approximate confidence limits for both reliability indices. The second approach applies Monte Carlo simulation to obtain approximate system reliability limits at any confidence level. Both approaches are also presented for the probability of failure of non-repairable systems. A sensitivity analysis of the component reliability parameters in a digital protection system is evaluated using the repairable approach.

para

Claudia y Adriana

quienes me han dado su amor, abnegacion y estimulo en todo momento.

ACKNOWLEDGEMENTS

This research work was conducted under the supervision of Dr. B.J. Cory, B.Sc.(Eng.), D.Sc., A.C.G.I., C.Eng., F.I.E.E., Reader in Electrical Engineering, Imperial College of Science and Technology, London, whose help and constant encouragement are gratefully acknowledged.

I am grateful to the Compañía Anónima de Administración y Fomento Eléctrico (CADAFE), Venezuela and the British Council for the financial support which made this work possible.

I would like to thank all my colleagues of the Power System Section, particularly Messrs. E. Horne, A. Ranjbar, A. Traca, S. Molina. and E. Arriola who contributed directly or indirectly to this work.

Finally, I would like to thank the Electrical Engineering Department of the Imperial College of Science and Technology for the opportunity given to lecture in the Digital Protection of Power Systems, Post Experience Course in 1975 and 1976.

TABLE OF CONTENTS

			Page
Titl	.e		l
Abst	ract		2
Ackr	lowledge	ements	4
Tabl	e of Co.	ontents	5
List	; of Sym	bols and abbreviations	11
CHAI	TER I :	INTRODUCTION	14
1.1	Data v	validation	14
1.2	System	ı reliability	15
1.3	Conter	nts of this thesis	17
CHAI	TER II	: RELAYING DATA VALIDATION AND ESTIMATION	21
2.1	Introd	luction	21
2.2	Review	of the literature	21
2.3	Protec	tion statistics	24
2.4	Relayi	ng data validation	27
	2.4.1	Information from the system	27
	2.4.2	Types of error	29
	2.4.3	Problem formulation	29
2.5	Data v	validation and estimation procedure	30
	2.5.1	Digital data validation -	31
	2.5.2	Analogue data validation	31
	2.5.3	Relaying data base estimation	34
2.6	Substa	tion-computer model system	39
	2.6.1	Data acquisition processor	41
	2.6.2	Data system hardware	41
	2.6.3	Intermediate storage	45

CHAP	TER II (Contd)	Page
2.7	Software system	45
	2.7.1 Programming considerations	46
	2.7,2 Computer requirements	47
2.8	Off-line tests	51
	2.8.1 Trials 1	51
	2.8.2 Trials 2	52
	2.8.3 Trials 3	53
2.9	On-line tests	70
	2.9.1 Hardware arrangement	70
	2.9.2 Test program	72
	2.9.3 On-line test results	74
2.10	Final comments	79
	-	•
CHAP	TER III : SYSTEM RELIABILITY ANALYSIS	80
3.1	Introduction	80
3.2	Terminal-pair reliability problem	80
3.3	Event enumeration	81
3.4	Factoring theorem	84
3.5	Tie-set enumeration	87
	3.5.1 Problem formulation	87
	3.5.2 Network representation	88
	3.5.3 Methods of solution	88
3.6	Cut-set enumeration	95
	3.6.1 Problem formulation	95
	3.6.2 Network representation	9 6
	3.6.3 Methods of solution	98
3•7	Selection of the method	102

.

•

6

•

•

..

CHAPI	ER III (Contd)		Page
3.8	Approximations for rel	iability evaluation	103
	3.8.1 Mutually exclus	ive approximation	104
	3.8.2 Independence ap	proximation	105
3.9	The inclusion-exclusio	n method	105
3.10	System reliability ind	ices	108
3.11	Final comments		111
			210
	ER IV : MINIMAL COT-SET	ALGORITHMS	112
4•1 4 0	Introduction	A - Finition	112
4.2	Madelling of complement	aeiinitions	112
4.5	Moderning of complex s	ystems by reliability graphs	110
4•4	Norsensuchie grande	FIX	10
4•2	Nonseparable graphs		121
	4.5.1 Description of	the algorithm	121
	4.5.2 Programming con	siderations	126
	4.5.3 Tests	. · ·	129
4.6	Separable graphs		132
	4.6.1 Definitions and	basic assumptions	136
	4.6.2 Description of	the algorithm	138
	4.6.3 Tests		143
4.7	Acyclic directed graph	S	143
	4.7.1 Basic definitio	ns -	145
	4.7.2 Description of	the algorithm	146
	4.7.3 Tests		152
4.8	Node failure problem		155
4•9	A branch-node cut-set	algorithm	16 2
	4.9.1 Basic definitio	ns and assumptions	16 2
	4.9.2 Description of	the algorithm	164

-

C	HAPTE	R IV (Contd)	Page
		4.9.3 Tests	169
		4.9.4 A practical application	173
4	.10	Conclusion	175
		· · ·	
C	HAPTE	R V : VARIANCE AND APPROXIMATE CONFIDENCE LIMITS FOR THE PROBABILITY AND FREQUENCY OF SYSTEM FAILURE	178
5	. 1	Introduction	178
. 5	. 2	Distribution of the failure and repair models	179
5	i•3 '	The exponential distribution in reliability	181
		5.3.1 Estimate and variance of the parameter	181
		5.3.2 Reliability estimate and variance	182
5	.4	Ninimal cut-set basic formulae	183
5	• 5	Problem formulation	185
5	.6	Review of the literature	186
5	• 7 :	Deterministic approach	19 0
5	.8	Approximate statistical approach - Moment method	196
		5.8.1 Low order moment formulae	197
		5.8.2 Expected value and variance	198
		5.8.3 Confidence limits	203
		5.8.4 Tests : Test 1	204
		5.8.5 Test 2	207
		5.8.6 Test 3	208
		5.8.7 Test 4	208
		5.8.8 Test 5	210
		5.8.9 Test 6	210
5	•9	Approximate confidence limits - Monte Carlo approach	226
•		5.9.1 Non-repairable systems	227
		5.9.2 Algorithmic procedure and an illustrative system	229

-.

•

•

CHAPI	TER V (Contd)	Page
	5.9.3 Comparison with the Moment approach	231
	5.9.4 Repairable systems	236
	5.9.5 Algorithmic procedure and an illustrative system	240
	5.9.6 Comparison with the Moment approach	242
5.10	Distribution of the reliability indices	251
5 .1 1	Conclusion	259
СНАРТ	YER VI : SENSITIVITY ANALYSIS AND RELIABILITY CALCULATIONS IN DIGITAL PROTECTION SYSTEMS	261
6.1	Introduction	261
6.2	Sensitivity coefficients	261
6.3	Application to system reliability indices	262
6.4	Illustrative example	265
6.5	Computer-operated protection reliability	2 66
6.6	A computer-based protection scheme	271
6.7	Alternative configurations - Reliability evaluation	272
.6.8	Reliability analysis in a digital protection system	274
	6.8.1 Failure rate	281
	6.8.2 Repair rate	282
	6.8.3 Coefficient of dispersion	282
	6.8.4 Confidence limit	283
	6.8.5 Sensitivity coefficients	283
6.9	Conclusion	284
CHAFI	ER VII : CONCLUSIONS AND SCOPE FOR FUTURE RESEARCH	288
7.1	Use of data validation techniques in digital relaying systems	288
	7.1.1 Zero data detection	2 88
	7.1.2 Doubtful data identification	288

CHAPTER VII (Contd)	Page
7.1.3 Application in a one-corner mesh substation	289
7.2 Reliability evaluation of complex systems	290
7.3 Minimal cut-set enumeration	290
7.4 Node failure problem in power systems	. 290
7.5 Probability and frequency of system failure	291
7.6 Reliability considerations in the design of digital protection systems	291
7.7 Future work	292
7.8 Original contributions	293
APPENDIX 1 : EDGE LISTING OF GRAPHS	295
APPENDIX 2 : PROOF OF THE MINIMAL CUT-SET ALGORITHMS	302
BEFE	315

~ \

LIST OF ABBREVIATIONS AND SYMBOLS

Abbreviations

CEGB	Central Electricity Generating Board, U.K.
CIGRE	Conference Internationale des Grands Reseaux Electriques a haute tension
Tol	Tolerance
Resc	Residual counter
Σĭ	Summation of currents
Ph R	Phase red
Ph Y	Phase yellow
Ph B	Phase blue
s/c	Samples per cycle
μз	Microseconds
ADC	Analog to digital converter
A/D	Analog to digital
c.b.	Circuit-breaker
N/0	Normally open
N/C	Normally closed
IO	Input-output
DAP	Data acquisition processor
CPP	Control and protection processor
RDB	Relaying data base
DB	Data base
COMS	Complement and swap instruction
f/y	failures per year
h	Hours
m/y	Maintenance per year
MTTF	Mean time to failure
MTBF	Mean time between failures

.

n ^C m	n-combinations of m-elements
n II a _i i=1	$(a_1 \cdot a_2 \cdot a_3 \cdot \dots \cdot a_n)$
U	Logical union
\cap	Logical intersection
\oplus	Ring sum operation
P _s (s,t)	Probability of successful communication between nodes s and t
P _f (s,t)	Probability of failure between nodes s and t
P _{jd}	Steady-state probability of j-component being failed
$^{P}\mathbf{f}$	Steady-state probability of system failure
f _f	Frequency of system failure: the mean number of system failures per unit time
E	ith event
Pr(E _i)	Probability of the ith event
\overline{E}_{i}	Complement of the ith event
p	Probability of success for identical components
^p i	Probability of success of the ith element
q _i	Probability of failure of the ith element
T _i	ith minimal path or tie-set
Ī.	Denotes failure of at least one element of T_i
cj	jth minimal cut-set
\overline{c}_{j}	Denotes failure of all components in C_j
R U	Upper bound of system reliability
R _{Ui}	ith upper bound of system reliability
$^{ m R}{}_{ m L}$	Lower bound of system reliability
$^{ m R}_{ m Li}$	ith lower bound of system reliability

Symbols (Cont'd..)

λj	Failure rate of jth component
μj	Repair rate of the jth component
μ _i	Sum of μ_j over all j ϵ C _i
 μ _{i+k+l}	Sum of μ_j over all j ϵ C _i U C _k U C _l
λ_{line}	Failure rate of a transmission line
λ_{node}	Failure rate of a node or busbar
MCi	ith minimal cut-set
$\mathtt{MC}_{\mathtt{ij}}$	ith minimal cut-set represented by a j-tuple, $j = 1, 2,, NE$
CNV _{ij}	ith cut-node vector represented by a j-tuple, j = 2,3,,NN
CF _i	Cut-flag of the ith cut-set
CI _k (i)	Cut index of the ith cut-set when scanning node k
NK	Node-counter
FN(k)	Flag of the kth node
[k]	Set of edges incident into a node
[¤]	Set of edges incident out of a node
NN	Number of nodes
NE	Number of elements
NC	Number of components
^{BF} ij	Busbar-flag of the ith cut-set represented by a j-tuple, $j = 2, 3, \ldots, NN$.
NMC	Number of minimal cut-sets
$\sum_{i=1}^{n} a_{i}$	$(a_1 + a_2 + a_3 \dots + a_n)$

CHAPTER I

INTRODUCTION .

Protection systems use real time system measurements for their fault detection and tripping decisions. Due to the inherent uncertainty of raw measurements, system data may not be reliable enough for immed- . iate use because if bad data are processed incorrect or unreliable results are obtained. A significant advantage of the use of digital computers for protection functions over conventional schemes is their ability to handle discrete data and to perform some checks on the incoming crude data. A systematic treatment of the input measurements using data validation techniques enhances the reliability of the decisions taken to protect and control a power system. The results of such data-processing algorithms in a computer-operated protection system have been presented in the form of a data base having common access for all the protection strategies. Since any failure of the computer could affect one of the protections associated with every item of the substation some form of redundancy is normally envisaged for such a system. Reliability theory needs to be used for appraising each redundant configuration and a proper selection can only be made on the basis of quantitative reliability analysis. This thesis deals with the use of data validation techniques and develops reliability calculations required for computer-based protection systems.

1.1 Data validation

For power system protection there is a need to take decisions in the presence of gross errors. Data validation techniques may be used as a

framework to convert observations into reliable data in a computeroperated protection system so that decisions can be made to produce more secure operation of the protection system.

Data validation has been extensively used in communication systems by means of coding techniques for error detection and correction. Unfortunately these techniques are mainly related with transmission errors but do not provide checks for measurement errors. There is a growing literature about the bad data problem in electric power systems but this only applies to the steady state estimation problem. Similarly, considerable work has been done in the application of digital computers to protection function but relatively little has been done on the validation of relaying quantities.

The following tasks are part of the overall relaying data validation problem. First the raw measurement sets are processed to detect bad data. Once gross errors have been identified, the algorithm should replace any missing data and provide the best estimates for the highly corrupted measurements. Since the data acquired is to be useful for system protection, the speed of acquisition and validation must be compatible with the desired fault detection times.

1.2 System reliability

The reliability of the data-acquisition and data-processing system of a computer-based protection system is a very important factor since an objective of protection is to increase the reliability of the power system.

Before considering actual problems let us define the concept of reliability. In its most simple and general form, reliability is a probability of success, but an accepted definition⁽¹⁷⁾ can be stated as:

"Reliability is the probability of a device performing its purpose adequately for the period of time intended under the operating conditions encountered"

Criteria of what is considered an "adequate performance" and how to measure it, have to be exactly specified in each case. For example, a computer-operated protection system must operate in the presence of a fault but it must not produce false operations. As long as the system performs these functions, the scheme is judged as operating satisfactorily. Measurement of adequate performance requires calculation of the probability of failure to operate and the probability of unnecessary operations. K. Conelly et al⁽⁷³⁾ refer to the former as the "system unreliability" and to the latter as the "system security" against false operations. To judge adequate performance also involves the determination of the frequency of failures and malfunctions.

Failures of digital systems are the consequences of faults which arise as a result of a step-type change in the performance of one or several of the components of the system. These failures can be properly modelled by random variables governed in many cases by an exponential $law^{(48),(67)}$.

The reliability calculations of even simple devices are frequently quite complicated because normally a number of components must operate properly to perform the system function. In the literature there are methods that could be used to calculate the exact reliability (18),(26-27),(34) but unfortunately they involve excessively voluminous computations for complex systems. This situation has prompted considerable research in approximate methods (16),(18),(30-33) for the reliability evaluation of complex systems from known reliability indices of constituent elements. Moreover a deterministic appraisal of the performance of

the system fails in many cases to provide a true picture of the system reliability because of the uncertainties associated with component parameters. Statistical approaches (56-60) are required to construct confidence from which the goodness of the performance can be assessed.

1.3 Contents of this thesis

This thesis deals with the development and testing of computer methods suitable for the on-line relaying data validation and for the reliability assessment of computer-based protection systems.

Chapter II covers the area of relaying data validation. Protection statistics are analysed and the incidence of bad data points in faultclearance equipment is discussed. The relaying data validation problem is described and a new algorithm is proposed and tested for the detection and identification of loss of data and large unexpected errors. Special consideration is given to the identification of zero data. An estimation procedure is presented which provides replacements for missing data or a best value for doubtful measurements. The feasibility of using both algorithms in one corner of a typical 400 kV mesh substation on the CEGB's network is reported. On-line results are given of the data validation algorithm tests on a laboratory substationcomputer model system.

Chapter III describes and analyses possible methods for the evaluation of the terminal-pair reliability. Two of the four main lines of attack (tie-set and cut-set) are followed because of the computational advantages of these approaches for complex systems. Consideration is given to the effect of term simplification and ordering of the partial reliability table constructed from the system success paths. A set of rules are formulated for the selection of the evaluation method in complex systems with unreliable branches and nodes. Three different approximations and bounds from the tie-set and cut-set approaches are numerically compared in both lower and upper reliability regions.

Chapter IV deals with the minimal cut-set enumeration problem. Since much of the work reported is based on graph theory, important concepts and definitions used are summarised. A description of system modelling by reliability graphs is included and the concept of cutnode incidence matrix is introduced. Three new algorithms are presented for separable, nonseparable and acyclic directed graphs. A simple sparsity technique is proposed for the efficient storage of the minimal cut-set and cut-node incidence matrices. Several graphs were used to test the algorithms and computation times are provided in each case. The node failure problem is analysed and reference is made to the incidence of busbar faults on the reliability indices of each node in a power system. A new branch-node cut-set algorithm is presented and the results of tests on three systems reported.

Chapter V presents two different approaches for the variances and confidence limits of the reliability indices derived from the minimal cut-set approach. The first technique involves the orthodox statistical procedure of characterising distributions by their low order moments and uses Chebyshev's inequality to provide probability bounds for the indices. The second approach applies Monte Carlo simulation to generate numerical distributions from where confidence limits are computed. An extensive analysis is made of the effect of component availability, parameter variances and sample size of component tests on the performance of both methods.

Chapter VI treats the problem of sensitivity analysis in reliability evaluations. A new approach is presented which calculates sensitivity coefficients of each reliability index given by the minimal cut-set approach with respect to each component reliability parameter of the system. General considerations used for the reliability evaluation of computer-based protection systems are described. An application of the algorithms presented in this thesis on two particular protection system configurations is reported.

Chapter VII discusses the conclusions reached and suggests possible future developments.

After Chapter VII there are two appendices. Appendix 1 includes the edge-listing of all graphs used in the tests of the algorithms proposed in this thesis. Appendix 2 presents a complete discussion and the proofs developed by the author of his minimal cut-set algorithms described in chapter IV.

The contributions offered by this thesis are the following:

- i. the development and testing of a data validation algorithm to carry out validation routines during the fault period (chapter II).
- ii. an estimation procedure is proposed to form a reliable data base in high voltage substations (chapter II).
- iii. The development and testing of three new algorithms based on graph theory for the minimal cut-set enumeration of separable, nonseparable and acyclic directed graphs (chapter IV, appendix 2).

- iv. the design and testing of a branch-node cut-set algorithm particularly suitable for the evaluation of the incidence of busbar faults on the reliability of distribution and transmission systems (chapter IV).
 - v. the development and testing of a new formulation for the variance and confidence limits of the probability and frequency of system failure (chapter V).
- vi. the development of sensitivity coefficients for the location of redundant components to improve system reliability based on the minimal cut-set approach (chapter VI).

CHAPTER II

RELAYING DATA VALIDATION AND ESTIMATION

2.1 Introduction

A computer-based relaying system uses measurements taken in real time to perform the relaying calculations to produce decisions for the different protection strategies. On-line measurements are usually corrupted by some degree of error or can be missed by malfunctioning of the data acquisition system. If some measurement has a large unexpected error or simply becomes not available, the relaying calculations might produce an unwanted operation or the system might become inoperative in the presence of a fault. The risks derived from these possibilities justify the need to use a data validation technique for the detection and identification of bad data points and a process of estimation to replace missing or erroneous data.

This chapter deals with the software problem of validation and estimation of the relaying quantities which form the relaying data base from which the protection strategies can draw the appropriate data. The possibility of using such techniques in one corner of a typical 400 kV mesh substation on the CEGB's network is also analysed.

2.2 <u>Review of the literature</u>

In spite of all the recent interest in this field, very little attention has been paid to the validation and estimation of relaying quantities for real time application in a computer-based relaying system.

The Swedish municipal power distribution system⁽¹⁾ and the Hamburg-West Germany system⁽²⁾ have reported the use of coding techniques for data validation. The Swedish safeguarding of data is in the form of a combination of parity bit checks, bit length error detection and code restriction checks. The latter is used for validation of data such as switch status and command operations which are generated as defined combinations of bits, other combinations are detected as errors. In the Hamburg system the majority of measured values, indications and commands, are transmitted digitally by an electronic supervisory control system using pulse duration modulation. Commands and indications are transmitted independently of each other and protected by a special code. The decoding of the pulse trains is performed by hardware as well as by computer program. The limitation of these techniques is that it only endeavours to maintain the original data unchanged but it is powerless if the data is wrong in the first place.

Couch⁽³⁾ developed a configuration analysis technique which he applied, among other, for data validation. He uses the substation internal configuration and Kirchhoff's current and voltage laws to establish zero differential current zones and equal voltage zones. If, under non-fault conditions, for any such zero differential current zone, the differential current is non-zero, then there is an inconsistency. Similarly, if in any such equal voltage zone, the measured voltages are not equal, then there is an inconsistency in the information. In both cases the inconsistency can be due to a measurement error and/or error in switch position indication. The analysis of inconsistencies once detected is less straightforward and no

systematic procedure is proposed for a real time application. Also the main type of measurement errors assumed are major ones such as those due to loss of secondary potential or current.

Another data checking procedure is included in the programs at the CEGB new national control centre (4-6). Raw data of circuit power flow and switch positions of the complete network are taken at 1 minute intervals and organised into tables to associate them with appropriate nodes and circuits. The checking procedure begins by taking two consecutive samples which are compared to see if they are consistent within a margin of 5% of actual flow or 20 MW if flow is small. If they are, the program continues with the latest of the 1 minute samples of complete data, otherwise one sample is retrieved from the computer backing store and a decision is made on a 2 out of 3 basis. The switch indication check uses two independent pieces of data telemetered to indicate switch position. A (1,0) combination indicates the open state and a (0,1) the closed state. Any other combinations are doubtful and require checking. With a doubtful switch position, the method is to assume it to be closed if current flows in the adjacent line and assume it to be open if no current flows. Alternatively a decision table constructed with the three pieces of data available can be used to make a decision on a 2 out of 3 basis. Such techniques are used to provide information for secure operation of the system.

Horne and $Cory^{(7)}$ propose a data validation method aimed at fulfilling its task in an interval of 625 µs i.e. sampling rate of 32 samples/power frequency cycle. The switch position check uses also two pieces of data which are logic complement of each other. If there is any inconsistency between them the switch data is given

a logic 1 to indicate an error. No other check is made during the validation to correct a doubtful switch indication. For non-zero analogue data, the status of the neighbouring switch(es) is examined. If closed, the data is assumed to be consistent, otherwise the data is tagged as doubtful. After these checks in the data acquisition processor, all data is transferred to the control and protection processor (CPP). It is the job of the CPP to choose either to disregard tagged data or to replace it by correct data calculated from other measurements. The method proposed does not have the capability of detecting and identifying a missing or erroneous data and does rot propose any corrective action to replace a doubtful data, missing or erroneous.

The research work reported in this chapter deals with the software problem of data validation during the time interval available between successive samples and specifically, the error detection and identification problem.

2.3 Protection statistics

The probability that a protection system might fail to operate or might produce unwanted operation in the presence of bad data can be found from protection statistics available from real systems. The results of a 1974 survey of failures in fault-clearance equipments⁽⁸⁾ are summarised in table I. Only the figures representing failure to operate and unwanted operation are presented. The results from a previous 1958 CIGRE survey⁽⁹⁾ are also included in this table.

The largest single cause of failure in both surveys is attributed to relay malfunction. Both reports indicate that in many instances relay failures result from wrong application or incorrect

TABLE I

PROTECTION STATISTICS

	197	1050	
Cause of incident	Failure to operate (%)	Unwanted operation (%)	(%)
Relay failure	43.2	48.5	43.0
Circuit-breaker mechanism	9.0	3.6	7.0
Circuit-breaker failed to interrupt	7.0	0.2	13.5
a.c. wiring	5.2	4.5	12.0
Current transformer	5.2	3.0	7.0
d.c. supply	5.2	0.2	1.0
Trip wiring	4.2	2.4	5.0
Voltage transformer	4. 0	0,9	3.0
Auxiliary switch	2.8	-	3.0
Circuit-breaker trip coil	2.5	0.4	2.5
Multicore cable	1.4	0.4	-
Miscellaneous or unknown	10.3	35•9	3.0

setting. The use of a computer-based relaying system would effectively eliminate problems related to these effects. Unfortunately it is not possible, from this report, to obtain a figure for the availability of the relays because they have not been analysed for all relay trip operations i.e. the figures for correct and desired operations are not reported. However, from previous utility reports⁽¹⁰⁾ it has been found that conventional relays have a typical availability of 97.5% when evaluated on situations where relay action should have taken place or which took place inadvertently. By comparison, digital computer availability in excess of 99% can be reasonably obtained. This availability is based on overall operation and not just on reliability when certain events occur; thus computer implemented relaying should result in more dependable relay functions and should reduce the number of failures attributed to the protection system.

The next highest number of failures is for those resulting from some form of circuit-breaker malfunction. However, if we assume that failures in a.c. wiring, current transformers and voltage transformers produced that the measured quantities were not available or were wrong, then such conditions can be considered as bad data. The figures attributed to such causes are comparable or higher than the overall percentage of circuit-breaker failures for both types of incidents. If one considers valid the previous assumptions, the actual number of failures that can be attributed to the presence of bad data demand the use of data validation and estimation routines in a computer-based relaying system. Protection statistics of the CEGB network⁽¹¹⁾ also indicate that the second highest maloperation rate of distance protection was due to loss of voltage transformer supply for both types of relays, electromagnetic and static.

2.4 Relaying data validation

The relaying data validation problem can be described as the detection in real time of any bad data, missing or erroneous, which may lead to possible failure, misoperation or deterioration of the protective relaying system.

Its solution requires the use of a data processing algorithm which fits the requirements of the fault clearance time desired and form a relaying data base from which all the protection strategies can draw appropriate data.

2.4.1 Information from the system

The information from the system can be divided for the purpose of validation into:

- (i) analogue data corresponding to system current and voltage information which are derived from the output of the current and voltage transformer.
- (ii) digital data representing the information of the switch status which is usually obtained by monitoring auxiliary contacts of each switch.

In a computer-based relaying system within the substation environment, current and voltage waveforms are sampled at all selected points in the substation and are then input, together with the switch status information, into the processor which is responsible for data acquisition and validation. The processed data is assembled into the relaying data base from which all the protection strategies can draw the appropriate data. Fig. 1 illustrates the complete process.



Figure 1. Block diagram of a computer-based relaying scheme

2.4.2 Types of error

Fig. 1 shows that there are two main types of error in the input information to the data validation process namely:

- (i) measurement error
- (ii) switch indication error

Type (ii) errors arise normally from failure in the auxiliary contacts of the switch. Type (i) errors are due to the fact that no physical measurement is perfect being usually corrupted by some degree of error. However, the measurement errors of concern in the relaying data validation problem are not the normal randcm errors but gross errors due to failures in the data acquisition link or in the voltage and current transducers.

To solve the bad data problem the relaying data validation process must be able to identify two possible measurement errors that might produce the worst effects from a protection point of view namely:

- (i) loss of data that might result in failure to operate in the presence of a fault
- (ii) large unexpected errors that might degrade the relaying computations and eventually produce unwanted operations

2.4.3 Problem formulation

Given the set of analogue and digital data from the substation the critical functions during the validation process are

- (i) detection and identification of bad data
- (ii) estimation of the most probable value for each single quantity in the data base

The main constraint in the design of a relaying data validation algorithm is the short time available between samples for validation assuming that the data base must be updated after each sample cycle. For instance, if the data is sampled at a rate of 32 samples per cycle corresponding to a sampling interval of 625 μ s for a system of 50 Hz then during this time the processor must read in system data, perform data validation and assemble the data base.

Considering this time restriction, the design of a relaying data validation algorithm to fulfil the previous functions must be based on the following principles:

- (i) the algorithm must be simple so that the speed of acquisition and validation will be compatible with the desired fault clearance times.
- (ii) the method must be flexible enough to cope with any changes with the minimum of modifications
- (iii) the data base must be reliable i.e. the presence of gross errors must be detected, identified and each single quantity presented with the most probable value.

2.5 Data validation and estimation procedure

There is normally considerable redundancy in the information available in a substation which aids malfunction detection. In a computer-based relaying system there must be special concern for the reliability of the information because it is used to automatically control elements of the substation. The data required must be previously validated. The checks proposed in this section are to supplement the techniques proposed by Horne and Cory⁽⁷⁾, but are more general in that they supply means to detect and identify gross measurement errors and an estimation procedure is proposed to provide replacements for missing or erroneous data.

2.5.1 Digital data validation

Normally in a computer-based relaying system the information on the open and close status of each switch will be read as two bits. These two bits reflect the status of a pair of contacts operated by the opening and closure of the switch.

These two bits are logic complement to each other and the validation consists of checking if both pieces of data are complements of each other. If there is an inconsistency the switchdata is given a logic 1 as an error flag to indicate a doubtful switch indication⁽⁷⁾. The corrective action for error in the switch position is taken during the validation of the analogue data using a sampleswitch routine. Fig. 2 shows a flow diagram for check of switchstatus information.

Whatever the structure of the switch-data word, the previous check requires one bit to be used as error flag.

2.5.2 Analogue data validation (12)

A zero data can be associated with any of the following conditions:

- (i) the waveform can be passing through zero
- (ii) the circuit is de-energized
- (iii) a fault in the data acquisition link



Figure 2. Flow diagram for check of switch-status.

The third condition affects the reliability of the data base, therefore the validation process must provide the steps to discriminate between these three conditions.

Zero data recognition by software will generally take more time than that available for all the validation process due to the conversion tolerance of the A/D converter. For instance in a NOVA minicomputer it takes 14 μ s/phase. To avoid this time-consuming check the zero data identification, within the tolerance of the A/D converter, must be done by hardware and a zero data can be given a logic 1 as a zero flag. This check by hardware takes just a few nanoseconds for each three-phase point.

With zero data recognition by hardware the analogue data validation checks the zero flag in each phase and this check over all three phases enables the identification that zero data in just one phase is due to the sample being taken when the waveform passes through zero.

If all three phases have been tagged as zero by hardware, a positive indication that the adjacent switch(es) is open is used to indicate that the circuit is de-energised; otherwise the data is given another logic 1 as error flag to indicate doubtful data. The missing data will be replaced or estimated during the estimation process.

A non-zero value in any phase of a current transformer is an indication that the circuit is energized and therefore used for validation that the adjacent switch is closed. This sample-switch routine together with the previous validation of the switch by the two complementary bits provides a 2 out of 3 decision.

The presence of a large unexpected error or the loss of data in one or two phases is detected by comparison of the residual value against a tolerance, only if the tolerance is not exceeded the data will be accepted as valid, otherwise it is given a logic 1 to indicate doubtful data which needs to be replaced or estimated. This tolerance will also be exceeded during any system abnormality and therefore a residual counter is used to discriminate between transducer error and a system abnormality. The latter will be reflected in all the active monitoring point.

The checks mentioned for analogue data require that the dataword uses two bits as zero and error flags. A detailed flow diagram of the algorithmic procedure for digital and analogue data validation is shown in Fig. 3.

2.5.3 Relaying data base estimation

The estimation concept here is used to describe a process which can provide replacements for missing or erroneous data. Such a process can only be effective if there is some redundancy in the information. In the substation environment this redundancy must be provided either by additional measurements (type I) or using the relationship implicit by the Kirchhoff's current law (type II). The latter is an inherent redundancy in electrical networks and the former is very often present when two different types of protection are applied to one item of plant, e.g. first and second main line protection. The estimation process should use any type of redundancy available to provide the best replacement for missing or erroneous data or the best estimate for clean data.



Figure 3. Flow diagram for data validation

The relaying data base (RDB) must be constructed considering the specific tasks to be performed by the protection strategies. There is no need to include any redundancy for a single quantity if this has been previously validated e.g. the line protection routines must have available three-phase line current and voltage.

Once the presence of a missing or doubtful data has been detected and identified the estimation procedure for each single quantity of the RDB is based on the following principles⁽¹²⁾:

- (i) a zero data with error flag clear is transferred directly to the RDB
- (ii) a zero data with error flag set is estimated from any type of redundancy available
- (iii) a non-zero value with error flag clear is transferred directly to the RDB if at least one of the correlated measurements has an error tag, otherwise the redundancy is used to estimate the best value
- (iv) for the case of a non-zero data with an error tag, the information from the residual counter is used to discriminate between a transducer error and a system abnormality. In both cases an estimation is done for that value but in the first case the value being checked is not used in the estimation unless the redundancy available has the error flag set.

The estimation routine for each quantity to be entered in the RDB can be done considering two types of measurement-redundancy relationship:
- I) Measurement with single redundancy : The estimate will be given by
 - any one of the measurements which has a clear error flag if the other is doubtful (missing or erroneous) and the residual counter limit has not been exceeded.
 - 2. the average value, if both measurements have clear or set error flags and the residual counter limit has not been exceeded.
 - the average value, if the residual counter limit has been exceeded except in case 4.
 - 4. the non-zero measurement if the other is zero and the residual counter is exceeded.

II) Measurement with double redundancy : The estimate will be given by

- any of the measurements which has a clear error flag if the other is doubtful (missing or erroneous) and the residual counter limit has not been exceeded.
- the average value of the measurement and the series redundancy (type I) if the residual counter limit has been exceeded or both data have clear error flags.
- 3. the redundancy type II if the other two data are doubtful (missing or erroneous) and the residual counter limit has not been exceeded.
- 4. the non-zero measurement between the measurement and the series redundancy if the other is zero and the residual counter limit is exceeded.

Table II summarises the possible cases considering all combinations of the zero and error flags between the measurement (first measurement)

TABLE II

1

Estimate values using zero and error flags information

	Fire	st	Second		Estimate Values			
	measurement		measurement		Single redundancy		Double redundancy	
Case	ZF	EF	ZF	EF	RC 🛠 lim	RC > lim	RC ≷ lim	RC > lim
			0	0	2	3	2	2
7	0	ο	0	1	1	3	11	2
ц.			1	0	1	4	11	4
			1	1	1	4	1	4
			0	0	1	3	1	2
	o	1	0	1	2	33	· 3	2
2			1	0	1	4	3	4
			1	1	2	4	3	4
			0	0	1	4	11	4
3	1	l	0	1	3	4	3	4
		1	0	<u> </u>	3	1	2	
			1	1	2	3	3	2

38

.*

;

and its redundancy (second measurement). For the case of measurements with only redundancy type II, the flags from all contributions must be equal to allow estimation, otherwise the first measurement is preferred except in case 3 of table II where the estimation is based on the flags of a single contributor. The case when the measurement has been identified as a real zero data (ZF = 1 and EF = 0) is not considered because a zero data without error flag is transferred directly to the RDB. The solutions presented in table II provide estimates more conservative than otherwise and are intended to use the flag information provided by the validation algorithm of sections (2.5.1) and (2.5.2) to provide replacements for a doubtful data (missing or erroneous).

2.6 Substation-computer model system

For research and development $purposes^{(7),(13)}$, the requirements for data acquisition and control of a typical 400 kV mesh substation on the CEGB's network has been chosen. As indicated in Fig. 4(a), one corner of such a station has been modelled by setting up 220 V, threephase circuits in the laboratory in which the circuit breakers and isolators have been simulated by three-phase contactors. These are slugged electronically to provide realistic operating times and the control circuitry enables them to be operated from a digital processor. In the mesh corner, full instrumentation is applied by means of voltage and current transducers, but in the remainder of the substation only the status of breakers and isolators is modelled by single phase relays so that the station configuration can be determined.

The digital system connected to the model is shown in Fig. 4(b) and consists of two processors having a common data base (DB). The



Figure 4(a). Mesh substation

data acquisition processor (DAP) is used to scan both analogue (transducer) signals and digital (c.b. and isolator position) signals from the model, to make validation checks, estimate each single quantity of the relaying quantities and to send the data to the DB. The control and protection processor (CPP) is responsible for running all the algorithms which determine the state of the substation and surrounding system, to switch it to eliminate faults or from a manual command from the local control room.

In this section only the hardware and software which are relevant to the data validation and estimation process in the one corner substation are described. Computer requirements in terms of memory and execution time are presented.

2.6.1 Data acquisition processor

In the substation-computer model system (Fig. 4(b)) the DAP is a NOVA 1210 mini-computer having a 12 k word core store, a 1.20 μ s memory cycle time, 16-bit length word and 4 programmable accumulators. The arithmetic instructions, with an execution time of 1.35 μ s, operate on fixed point binary numbers either unsigned or the equivalent signed numbers using two's complement conventions.

2.6.2 Data system hardware (7),(13),(14)

The interface provides 4-16 bit digital word inputs for switchstatus information and 48 multiplexed analogue inputs for current and voltage measurement.

Each switch position is read as two bits which reflect the status of a pair of auxiliary contacts operated by the opening and closing of the switch. This data is packed into 16 bit words each



Figure 4(b). Proposed hardware configuration for mesh corner protection and control [13].

containing 6 switch positions (12 bits) and 4 spare bits which can be used as error flags. Fig. 5(a) presents the switch-data word format.

Analogue signals which are monitored by the computer are currents and voltages. The current signals are measured by air cored current transformers (linear couplers). The sensitivity of the linear couplers is about 6 mv pp/amp r.m.s. The voltage, measured between phase and neutral, is obtained via a small step down transformer of 240/6 V r.m.s.

Each analogue output from the model is connected to a sample/ ζ hold circuit controlled by a single phase locked clock synchronised to the 50 Hz supply. A clock pulse causes all sample/hold circuits to sample simultaneously with a window of 1 µs. The sampling rate can be varied by computer control at 32, 24, 16, 12, 8, 6, 4, and 2 samples per cycle and an integrator is included in the sample/hold circuit to give a measure of filtering.

The analogue/digital conversion is performed by 3-10 bit, 25 μ s converters within the interface, thus enabling all 48 inputs to be converted and input in less than 525 μ s. For program control purposes the converters may be considered as a single peripheral device but for data input purposes each one is treated separately and 3 input instructions are required. For validation purposes a data "zero within tolerance" flag is available from each converter and is input with the converted data to bit 0 of the chosen accumulator. The flag is a logic 1 for the zero data condition. The tolerance limit is hardwired on each converter card and may be altered as required. Initial tolerance is preset to $\frac{+}{2}$ least significant bits of the converted word. Fig. 5(b) presents the analogue-data format.

(a) SWITCH - DATA WORD



(b) ANALOGUE - DATA WORD



Figure 5. Data-word formats

Each of the analogue/digital converters is served by a 16 way analogue input multiplexer which may be operated either by program control or in an automatic mode slaved to the A/D converters.

2.6.3 Intermediate storage

The DAP is linked to the CPP (a PDP 15 with 16 k store in this case) via a first-in, first-out (FIFO) buffer. Data output to this link is under NOVA program control and on completion of a data block the link generates a data channel request to the CPP and the block is transferred to the DB which is a portion of the FDP 15 core store. The DB stores two complete 50 Hz cycles of data before being overwritten.

2.7 Software system

During the time interval available between samples e.g. 625 μ s for a sampling rate of 32 s/c, the DAP is responsible for reading in system data (i.e. analogue to digital conversion, multiplexing and read in), performing the validation checks to identify doubtful data, labelling and storing it, performing a simple estimation and sending the data to the CPP.

With the NOVA, input and output of a single quantity takes about 2.55 μ s and 3.15 μ s respectively. In the model of Fig. 4(a) the total data movement time for the 4 switch-data words and 39 data words for current and voltage takes 245.10 μ s. Also the analogue/ digital conversion of each three-phase quantity takes 25 μ s i.e. a total conversion time of 325 μ s for all 13 three-phase monitored points.

So, besides the interface, specifically designed to minimise data acquisition overheads, it is also necessary that the validation be structured to optimise the overlap between input, conversion and validation procedures.

2.7.1 Programming considerations

The data acquisition cycle starts upon receiving a pulse from the synchronised clock which initiates a sample/hold operation and informs the computer to start an acquisition cycle.

After considering various alternatives it was decided that the validation process should read in the data and then, while the data is still in the accumulator, perform validation before storage. This process is quite attractive to the NOVA architecture which has four programmable accumulators. Also the validation checks (sections 2.5.1 and 2.5.2) were designed to treat each monitoring point individually without bothering too much about correlated measurements through the substation switch configuration. Zero data recognition by hardware was found essential to maximise the time available for validation between samples. Priority interruption is deliberately not used because once an interrupt is requested on completion of conversion, data needs to be read in and stored. Moreover, interrupt handling, restoration of the interrupted program, reading and storing of the data are very time-consuming. If the processor can read data in any time at its "convenience", it can wait until validation on the current data has finished and stored before reading in another set of data. While the processor is validating a particular word, the ADC is doing its conversion for the next data and is timed such that when the processor has finished with the present data, the ADC has also

completed its conversion. This arrangement allows conversion and validation to be carried out simultaneously and the processor is not idling during conversion periods.

The sequence of events at the start of data acquisition is shown in Fig. 6 and the corresponding flow diagram in Fig. 7. However, testing for completion the conversion takes too much processing time so the program is constructed to repeat the section of the analogue data validation for all the monitoring points. This requires increase core storage but the program is simple and storage is not a critical problem. So, once started the validation procedure continues until all monitoring points have been validated and stored in labelled locations for data retrieval by the estimation routine. Once each single quantity of the RDB has been estimated and sent to the CPP, the DAP waits to receive the next pulse from the synchronized clock.

The validation checks of sections (2.5.1) and (2.5.2) demand data manipulation at a "bit-level". Moreover, since very little arithmetic calculations are required, full advantage of assembly language programming can be taken.

2.7.2 Computer requirements

Memory size and execution times required for the validation checks of digital and analogue data are presented in table III (excluding data output). The memory requirements shown are in words (groups of 16 bits). All routines have been written in assembly language. The same requirements for the laboratory substationcomputer model described in section (2.6) are presented in table IV. The model considered in table IV includes 4 data-switch words (24 switches) and 13 three-phase monitoring points i.e. 9 three-phase



Figure 6. Sequence of events in the data acquisition and validation cycle



۰.

Figure 7. Flow diagram for data acquisition and validation

TABLE III

Memory Requirements and Execution Time for Data Validation

Routine	Memory Requirement (Words)	Execution Time (µs)
Switch Status	8	15.60
Three-phase Current	43	59.25
Three-phase Voltage	35	43.65

TABLE IV

Computer Requirements for Laboratory Substation-Computer Model

Routine	Memory Requirement (Words)	Execution Time (µs)
Data and Symbol Table	54	
Start and Switches	40	91.45
Analogue Data Validation	613	707.85
Line Estimation	132	101.00
Busbars Estimation	136	194.40
Transformer Estimation	71	70.95
Total	1046	1165.65

currents and 4 three-phase voltages. Whilst this provides a basis for comparison, it should not be inferred that the routines have been developed to a high degree of programming efficiency.

2.8 Off-line tests

The aim of the tests was to check the performance of the algorithms of section (2.5). A test program written in NOVA assembly language was controlled from a teletype and provided special codes to select the validation algorithms for switch status, analogue data or the estimation procedure. Once one of the algorithms was selected, the appropriate routine was run using the data previously stored in the NOVA and the results were written out on the teletype.

2.8.1 Trials 1

The objective of these trials was to evaluate the performance of the algorithm of section (2.5.1) to check the switch status information. Different simulated combinations of the substation switches (Fig.4(a)) were used for the tests.

Figure 8 presents, as an example, the results of 4 tests and each case shows the data-switch word for each corner of the meshsubstation model. The results of the first test shows that words 1 and 4 which have no inconsistency in any of the switches is given a logic "O" as error flag but words 2 and 3 have set their error flags to indicate the inconsistency of the pair of bits of the first switch in both cases. Similar results can be seen from test 3 where the error flags in words 1, 3 and 4 correspond to a doubtful indication of switches 1, 3 and 1 respectively. The results of test 2, where the error flag of words 1 and 3 arise from a doubtful information for all the switches indicate that each data-word gets the same error flag whatever the number of wrong switch information is in the word. The result confirms the need to provide within the relaying data validation algorithm the corrective action to clear any doubtful switch-data word i.e. all possible doubtful switches of each word. Test 4 presents a case where all 4 error flags are clear because there is no inconsistency in any one of the switches of each data-word.

2.8.2 Trials 2

This group of trials were performed to study the ability of the algorithm of section (2.5.2) to detect and identify gross measurement errors i.e. missing or erroneous data and to check the performance of the sample-switch routine to provide corrective action for a doubtful switch indication.

All tests start by checking the data-word for the switches of one corner. Two types of tests were performed to evaluate the algorithm for both types of data, current and voltage. Both routines are similar but for current values the presence of a non-zero value is used to correct a doubtful indication of the switch adjacent to the value being checked.

Figures 9 to 17 present the results from selected tests. The first test of Fig. 9 shows the identification of a real zero value using the zero flag of the 3 phases and the positive indication that the adjacent switch is open. All other tests detect a doubtful zero data either by the adjacent switch being closed (test 2) or having a doubtful indication (tests 3 and 4). Fig. 10 presents similar results when 2-phases are zero and the adjacent switch is closed. Similar tests were done for the current routine and selected results presented in Figs. 11 and 12.

Figure 13 illustrates the use of the sample-switch routine to correct a doubtful switch indication. Figures 14 and 15 show results from tests made to detect measurement errors when the residual value exceeds or not a pre-set tolerance (3 least significant bits for all tests). Tests were made to check the validity of such checks for positive and negative tolerance and some results are presented in Figures 16 and 17 respectively.

2.8.3 Trials 3

These trials were performed to check the ability of the estimation procedure proposed in section (2.5.3) to provide replacements for missing or erroneous data. The routine provided for the transmission line of the substation model (Fig. 4(a)) was used because it enabled a check on the procedure with measurement (voltage A or current S) having single redundancy (voltage C) or double redundancy (currents Q and T + U).

Figure 18 presents the results of a case where the current and voltage estimates are provided by the duplicate measurement (redundancy type I). Fig. 19 shows similar results but the line current estimate is taken from the line measurement (current S) because the series redundancy (current Q) has been tagged as doubtful data. The results of Figs. 20 and 21 correspond to the case when the estimates are given by the average value, if both measurements have either an error flag (voltages A and C) or have been accepted as valid data (currents S and Q). Fig. 22 illustrates the case when the averages provide the estimates as soon as the residual counter limit is exceeded. Fig. 23 shows the use of the redundancy type II (currents T and U) for the estimate when the measurement and the duplicate measurement are doubtful data.

SUITCH STATUS VALILATICL

CLCSE	OFEL	FLAG
111111	012213	· Ø
111111	102002	1 .
C11111	686408	1
668616	111101	£

SWITCH STATUS VALIEATION

CLCSE	OT EL	FLAG
100112	162110	1
211001	128116	£
501101	CC11C1	1
1@1@1@	$\mathbb{S} 1 \in \mathbb{I} \subset \mathbb{I}$	<u>e</u>

SUITCH STATUS VALILATICE.

CLCSE .	CFIL.	FLAG
201101	010210	- 1
110011	661166	Ģ
101011	011100	1
11111	102220	1 .

SUITCH STATUS VALIEATION

CLCSE	CFER	FLI.G
111889	000111	C
$e_{1}e_{1}1_{1}$	101000	e
186181	Ø1101Ø	\mathcal{L}
\$6111C	1.1%	Z

Figure 8. Data-switch word validation

CPILL	FL/.G
160666	C
606666	1
666666	1
600000	1
	CPIIL 100700 000707 717020 200000

T. VALIEATICE

PEASE	7F	Ľ.F	LEASUTENENT :
R	1	£	22272272CCCC
•	1	C	FECTERERE
Б	1	Ø	68666666666

RESIDUAL COULTER LESS THAD THREE

CLOSE	CPEN	FLAG	
111111	668666	Ē	
166666	003039	1	
166886	888288	1	
100000	250555	1	

T. VALIEATICE

FEASE	25	EF	TEASURENLAT
R	I	1	6466666666
Y	1	F	6220260062
Б	1	ø	6205556565

RESILVAL COUNTER LESS THAN THREE

CLCSE	CTLL.	FLAC
111111	100222	1
156205	606465	1
160662	222262	1
166066	0002280	1

T. VALILATICE

PEASE	2F	LF	MEASUPERENT
2	1	I	BBCC EEEEEE
Y	1	С	60060000000
Б	1	e	000000000000

RESILVAL COUNTER LESS THAN THREE

CLUSE	CTEL.	FLAG
811111	81 2 8 V C	1
126028	666668	1
162228	0000000	1
158562	688885	1

E	1	1	000000000000
Y	1	٤	C. 3 & F & C & C & C & C & C & C & C & C & C
E	1	e	PASCACLESS

PESILEAL COUNTER LESS TEAL THREE

Figure 9. Identification of three-phase zero voltage

CLCSE	CPEN	FLAG
111111	666666	C
111411	666666	Ø
111111	0200200	e
11111	666666	e

• •

T. VALIDATICE

PFASE	7 F	EF	NEAGUREMENT
P	e	1	0001010200
Y	1	ſ.	2233333333333
Ē	1	e	2605666666

PESILUAL COUNTER LESS THAN THREE

T. VALILATION

.

FEASI	7F	EF	NEASUPERELT
R	1	1	66968866666
Y	£.	ç	0001010000
E	1	£	0202828668

RESIDUAL COULTER LESS THAL THREE

T. VALIEATION

FEASE	7 F	EF	MEASURLMERT
P	1	I	66666666666
Y	I	£	<u>0</u> 220262000
В	Ø	Е	6221212522

FESTEUAL COUNTER LESS THAN THREE

. .

Figure 10. Identification of two-phase zero voltage

۰.

C	L	С	Ś	F		CPEN	FLAG
£	1	1	1	1	1	12222	ε
1	1	1	1	1	1	2222 2 00	Ľ
1	1	1	1	1	1	8 E E E E E E	ĺ.
1	1	1	1	1	1	626666	Σ

OT. VALIDATION

PEASE P Y E	77 1 1 1	EF Q Q Q	NEASUFI ECEC260 EEEEEEE EECEEE	ELLELT CCG2 CGC2 CCC2	
CLCSE 211111 FESIDUA	1 .L C	CPEN 22000 CULTE	FLAG C .r Less	TEAU	THREE
CLCSE 111111 111111 111111 111111 111111	0 9 2 9 2	FLL CCCCC CCCCCC CCCCCC CCCCCC CCCCCC CCCCC	FLAG 2 2 2 2 2 2 2	·	

CT. VALILATICE

PEASE	75	ΞF	MEASURENEDT
F	1	1	6600220000
Y	1	E	0002020020
Ŀ	1	2	000000000000

CLUSE	CFIN	- FLAG
11111	666666	Ø

PESIDUAL COUNTER LESS THAD THREE

Figure 11. Identification of three-phase zero current with clear switch indication

CLCSE	CPLU	FLAG
111111	188820	1
11111	666666	Ø
1,11111	868888	Ø
111111	886688	õ

CT. VALIEATION

ZF	EF	HEASULENEDT
1	1	000000000000
l	R.	6635666666
1	£	66666666666
	2F 1 1	RF EF 1 1 1 0 1 0

CLCSE OFEN FLAG 111111 100000 1

RESILVAL COUNTER LESS THAN THREE

CLCSE	CPEN	FLAG
C11111	200000	1
111111	656566	Ø
111111	6.22626	ø
11 1 1 1 1	2223222	e

CT. VALIDATION

FEASE	7 F	EF	MEASUFENELT
Р	1	1	66666666666
Y	1	£	222222222
E	1	2	02000000000
CLCSE		CPEN	FLAG

PESILUAL COUNTER LESS THAN THREE

Figure 12. Identification of three-phase zero current with doubtful switch indication

CL(SE	CPEL.	FLAG
@11111	265666	1
111111	& C C C C C C	E C
111111	698268	£
111111	000000	C

GT. VALIEATICL

FEASE	7F	EF	MEASUR ENENT
p.	e.	C	0000101000
Y	2	£	2221212222
E	C	2	1110110000

CLCSL	CFIL	FL1.G
11111	eccecc	1

PESIEVAL COUNTER LESS TRAIN THTEE

CLOSE	C F EL	FLAC
11111	200000	e .
111111	223923	C
111111	282780	C
111111	000000	C

CT. VALIEATICE

FLASE	÷.	ĿF	HEASUREMENT
E	C	6	8881888881
Ý	1	C	68666666666
В	C	P	1110111111

CLCSE	CPEN -	FLAG	
11111	CEEZEE	Ø.	

PESILUAL COUNTER LESS THAN THREE

.

Figure 13. Illustration of the sample-switch routine

CLCSE	CFLL	FL <i>I</i> .G
111111	666666	e
111111	666665	С
111111	666666	2
111111	626666	8

CT. VALILATICE FHASE PF EF MEASUFLIEHT P Y 1 1 PCSCERFECC C 0901910001 E Ð 1 Ç. 01033977333 CLOSE CFEI. FLAG 2 111111 2923393 RESIDUAL COULTER LLSS THAN THREE OT. VALILATICE TE 32.17 IF HEASUPELELT 1 1 200000000 1 0 000000000 -5 e 1110111111 £ (-EI) CLCE FLAG 111111 223932 C PESIEVAL COULTER LESS THAL THREE CT. VELICETICE PEASE 7F EF MEASURELENT 7 1 1110111111 1 2 202000000 F. Y Э 1 e 0020020222 CLOSE CPLI: FLAG 111111 200200 0 PESIEVAL COUNTER LESS THAT THELE CT. VALIEATICE. PFASE 77 EF REASUTEMENT F 1 1 50565556655 Ŷ C 8 66616656651 6 1111110110 E 8 CLCSE CPEI. FL.C.G 111111 226729 @ RESIDUAL COULTER LESS THAL THREE

Figure 14. Detection of doubtful data using the residual value and residual counter with zero data in one or two phases

CL(SE	CFEL	FLAG
11111	6886688	C
111111	666666	e
111111	292393	V.
111111	290093	Ø

CT. VALIEATION

FEASE	7F	EF	DEASURENENT
÷.	£.	1	LETTIKERK
Y	C	C	1111121100
E	F	ſ	1111110110

CLCSE	CFLI.	FLAG
111111	002000	C

FESIEVAL COUNTED LESS THAN THEEE

CT. VALIDATION

FFASE	7.F	EF	LEASUPENENT
P	1	£	10010000000
Y	C	C	5261626281
E	C	C	111011111

CLOCE CPEN FLAG 111111 COCCCC C

FESIEUAL COUNTER LESS THAL THREE

Figure 15. Detection of doubtful data using the residual value and residual counter

OT. VALIEATION

PHASE	7 F	ĽF	HEASURENENT
Б	e	£	1111011101
Y	C	2	6666616166
Б	e	£	0000010100

RESILVAL COUNTER LESS THAN THREE

CT. VALIEATICE

PEASE	2F	ĹF	NEASUTIENENT
R	P	£	1111011110
Y	C	c	2222210102
E	C	£	6666616165

PESIEVAL COUNTER LESS THAN THREE.

CT. VALIEATICE

77	EF	DEASUFERENT
6	C	1111011111
e	e	8660016166
C	£	2222010102
	77 0 0 0 0	77 EF 6 C 6 C C C

RESILVAL COUNTER LESS THAN THREE

CT. VALIEATIC.

PEASE	2F	EF	HEASUFENENT
F	e	1	1111120020
Ý	e	£	0000010100
E	C	ε	0000010100

-

RESILVAL COULTER LESS THAL THREE

CT. VALILATICE

PLASE	7F	ĿF	REASUPERENT
F.	e	1	1111166221
Ŷ	2	e	20101139202
Ε	e	ø	0028617192

RESILVAL COULTER LESS THAN THREE.

Figure 16. Detection of doubtful data by positive residual value exceeding tolerance CT. VALIEATION

PEASE	2F	EF	MEASUTEMENT
7	ç	£	1111201021
Ŷ	e	Ø	6655611651
E	ę	C	666C6116C1

RESIDUAL COULTER LESS THAN THREE.

CT. VALIEATICE

FEASE	ZF	EF	DEASURELEDT
E	2	C	1111021022
Y	C	£	0200011001
E	c	Ø	0020011201

RESIDUAL COUNTER LESS THAN THREE

CT. VALIEATICE

PEASE	27	ΞF	MEASUREMENT
E	ē.	2	1111220111
Ý	2	£	0200011201
È	ε	C	6665511561

PESILUAL COUNTER LESS THAN THREE

CT. VALIEATION

FEASE	27	EF	HEASUELNEL T
F.	c	2	1111090110
Y	2	£	1221133233
B	£	٤.	6666611661

RESIDUAL COUNTER LESS THAN THEEL

CT. VALILATICI.

PEASE	75	EF	NEASUNELIEUT
ŗ	C	1	1111666161
N.	ſ	Ç	6666611661
E.	£	£	0000011001

RESHUAL COULTER LESS THAN THREE.

Figure 17. Detection of doubtful data by negative residual exceeding tolerance

CLCSE	CEED	FLAG	
111111	656666	e	
111111	222622	£	
111111	666666	ę	
111111	608866	2	

VELTAGE "A"

PFASE	7 <i>F</i>	EF	MEASUT ENELT
E	1	1	00000002200
Ý	1	e	0000002002
E	1	2	6666666666

VOLTAGE "C"

FEASE	7F	ΞF	NEASUPENELT
P	1	C	000000000000
Y	e	c	6661666661
E	ε	r	1110111111

CURRENT "S"

FFASE	2F	ī.F	NEASUREMENT
F	1	1	00000000000
Y	C	£	1110110000
E	٤	e	0000101020

CUPPENT "S"

FEASE	7F	EF	MEASUREMENT
12	ſ.	e	00001010000
Ŷ	£	£	1116116022
E	e	E	0000101000

LINE	VCLTAGE	ESTIMATE
E	6 6	22672222222
Y	6 6	6661666661
Ξ	C C	1110111111

LINE	CUREL	ЦТ	EST INATE
Γ.	£	٤	2000101002
Ŷ	e	٤	1110110060
5	e	C	0220101000

Figure 18. Transmission line estimates by duplicate measurement

Ċ	L	¢	5	Ε		CFEL	FL1.G
1	1	1	1	1	1	222667	2
1	ł	1	1	1	1	Ceccre	£
1	1	1	l	1	l	\$22202	Ø
1	1	1	1	1	1	212222	e

VELTAGE "A"

FELSE	7 F	EF	HEASURENEL T
Π.	1	1	CRECCCCCCC
Y	c	ſ	00010000101
E	1	£	00000000020

VOLTAGE "C"

FEASE	7F	ĒF	LEASUF ELENT
F	1	c	00000000000
Y	2	٤	0001000001
£	F	С	1110111111

CUPRENT "S"

FEASE	7F	EF	LLASULENEUT
Ξ	ç	٤	0220121220
Ŷ	2	e	1110110000
E	e	£	6666151566

CURPENT "G"

FEASE	7 F	EF	REASUNELIENT
5	e	1	0000101000
Y	1	ę	0201020200
Б	c	e	0020101020

LINE	VCLTA	GE	ESTIMATE
Б	2	e	00000000000
Ŷ	e	2	2001220021
E	Ø.	ĺ.	1110111111

LINE.	CURRE	ъτ	ESTIMATE
E	C	ę	2000101200
N.	8	f.	1112110000
Ľ	e	e	8686161666

Figure 19. Line voltage estimate by duplicate measurement and line current estimate taken directly from the line measurement

SWITCH STATUS VALIDATION

С	L	С	ξ	Ξ		CFLI:	FL£.G
1	1	1	1	I	1	886688	c
1	1	t	1	1	t	666566	c
1	1	t	1	1	t	000000	C
1	1	l	1	1	l	CEEEEC	e

VELTAGE "A"

FHASE	77	EF	HEASURELLEUT
Γ.	e	I	6601011012
Ÿ	2	5	1111100101
Ε	C	£	1111115000

VOLTADE "C" CTPL S TO STAFT

FEASE	75	ΞF	LE#SITENELT
Γ.	£	1	0201212920
Y	ç.	c	1111112260
3	C	٢	1111116220

CUPPELT "S"

FFASE	2 <i>F</i>	LF	LLASUFEDEUT
F	£	e	8526156518
Ŷ	£	C	1110112012
Ε	C	e	0000101100

CURRENT "O"

FRASE	7F	EF	HEASUDELEL7
E	ç	6	2011219333
Y	£	£	1110101100
E	e	\mathcal{D}	6666156118

			•
LINE	VOLTAG	E	ESTILATE
Е	Z	ę	6661616161
Y	ſ	e	1111101010
Е	C	e	1111112022

LINE	OURRE	ТЛ	ESTIMATE
2	Ø	£	0220100111
Y	C	C	1112101111
Е	C	C	0000101201

Figure 20. Transmission line estimates by average values

STITCE STATUS VALILATION

CLCSE	CFE::	FLAG
111111	162505	£
111111	00-2020	С
111111	000000	e
111111	000000	C

VELTASE "A"

TEASE	ZF	EF	UEASUF ENERT
F	C	1	0001000110
Y	2	e	1111100000
E	C	C	1111110020

VELTAGE "C"

F11.SE	7 <i>F</i>	EF	NEASURLIEUT
. P	ç	1	0001010000
Y	ſ	2	1111112600
Ē	٤	2	1111001110

. CUPELLT "S"

FEASE	2 F	ΞF	NEASURENERT
I :	£	e	0100110000
Ŷ	e	ε	1110110010
Σ	ç,	£	6606121162

CUPFLUT "C"

.

FFASE	7F	EF	HEASUPERENT
в	C	C	6666161666
Y	C	2	1110110000
Б	E.	Ø	222121223

LINE VOLTAGE ESTIMATE R B 0 0001001011 Y 7 0 1111101000 B C 0 1111011111

LINE	CUPER	ĿТ	ESTIMATE
5	5	8	0000100101
Y	ç	Ø	1112110051
Б	٤	e	0000121010

Figure 21. Transmission line estimates by average values

SUITCH STATUS VALILATICS.

CLOL	CTLI.	FLAC
111111	011102	(°
111111	009910	2
111111	000003	C
111111	000000	C

VOLTAGE "A"

PEASE	7F	EF	DLASUREDIT
Г	C	1	6661316626
Y	Ċ	ſ	1111112200
Ē	e	ſ.	11111101000

VELTASE "C"

FEASE	7F	EF	LEASUPELELT
P	e	1	CCC1616200 .
Y	C	٤	1111011000
E	C	C	1111110110

CUPPENT "F"

FULSE	51 6	EF 1	LEASURELEUT
Ŷ	P	£	1100111000
Ē	1	2	000000000000

CUEPELT "C"

FLASE	75	ΣF	HLASUF EILELT
E	e	٤	c1c201111C
Y	£	e	1011100010
Е	1	C	00000000000

CUPERENT "T"

FFASE	7 F	ΞF	LEASUPLEELT
Ŀ	e	1	6616611111
Ŷ	e	7	1101110001
Ē	1	ç	660066666666666

CUPPENT "U"

FLASE	77	ZF.	REASURENELT
Ŀ	£	e	6610010020
Y	ę	6	1101101220
В	1	ę	6366626266

PESILUAL COUNTER GREATER THAN THREE

LINE	VELT	GE	LETILATE
E	ø	ø	2001210292
Ŷ	e	ę	1111102122
Ē	g	e	1111121111

LILL	CURPENT	ESTI	LATE
R	8 6	£ 1	119412333
¥ .	6 6	11	66861161
Ľ	66	62	202122000

Figure 22. Transmission line estimates by average values when residual counter has been exceeded

SVITCE STATUS VALIDATICS

CLCSE	CTEN.	FLAG
111111	0000000	2
111111	2223232	E.
111111	279779	r
111111	222293	2

VCLTAGE "A"

PFASE	7F	EF	LET SUPELEL T
P	r	C	0001011010
Y	C	ſ	1111001110
E	?	٤	1111211222
Ŀ	Ÿ	£	1111211266

VELTAGE "C"

FHASE	7F	LF	LEASUPELLI,T
E	e	1	011020130
Ŷ	C	£	1111162010
E	e	e	1111110000

CUFFELT "S"

FLASE	77	EF	REASURELELT
P.	C	1	1112116220
Ŷ	1	ε	6666666666
Ξ	ę	C	6626161666
· . ·	•		

CURPELT "C"

FLISE	7F	E.F	MEASUTERED
F	C.	1	1115111612
Y	R	. C	6600161106
E	R	e	SECCISCEIC

CUEPENT "U"

FFASE	7F	EF	MEASUREMENT
I.	ε	e	1111011020
Ŷ	C	ę:	2202211221
B	e	C	6666616168

LINE VOLTAGE ESTIMATE 7 C C CCCICIICIE Y C C IIIICCIIIC E C C IIIICII2CC

LINE	CURFI	ПT	ESTIMATE
F	Q.	ç	1110111012
Ŷ	ε	£	00001010111
Б	e	6	222012C122

Figure 23. Line current estimate by network redundancy

·

2.9 On-line tests

On-line data validation tests were made using the substation computer model system described in section (2.6). In this section only the hardware set-up and the software which were used for the switch and analogue data validation are described. Some typical results of the on-line tests are presented.

2.9.1 Hardware arrangement

The system hardware configuration set up for data validation for on-line tests is illustrated schematically in Fig. 24. As indicated in section (2.6) the mesh substation in Fig. 4(a) is simulated in two parts, called for identification purposes, the primary and secondary circuits. The primary circuit refers to the mesh corner simulated in 3-phase form with transducers for current and voltage measurements and contactors acting as circuit breakers and isolators. Push button switches are available for manual closing and opening. The term secondary circuit refers to the rest of the substation and simulates switch positions in the adjacent mesh corners by single phase relays which are manually operated. Substation switching configuration for on-line tests are simulated using this manual operation facility available for each switch. An interface has been specially constructed⁽¹⁴⁾ to provide 4-16 bit digital word inputs for switch status from the four corners of a mesh substation.

One primary circuit of the substation $model^{(13)}$ consisting of a contactor unit and a current and voltage transducers is connected to a 3-phase resistive load for analogue data validation on-line tests. The load current is limited to a maximum of 10 amp r.m.s. The current signals monitored by the computer are measured by air cored current



Figure 24. Hardware configuration for on-line data validation tests.

transformers (linear couplers). The linear coupler output is proportional to the rate of change of input current, such differentiation leads to the amplification of noise and high frequency components producing distortion on the output. Active low-pass filters with a 130 Hz cut-off frequency and a gain of about 50 are used for filtering and scaling the distorted waveforms of the linear coupler outputs during normal operation. The maximum voltage that can be applied to the A/D converter is $\stackrel{+}{=}$ 10 volt. No other precautions were taken to obtain more accurate measurements. The rest of the hardware configuration has been described in section (2.6.2).

2.9.2 Test program

Figure 25 shows the flowchart of the program which was used for on-line tests. The program starts by setting the synchronised clock to the sample frequency desired, the multiplexer addresses and mode of operation. Then it waits for the user to select either the switchdata validation test or the analogue data validation test. In both cases, once the proper code test is selected, the validation procedure starts reading the digital data-word for the switch status of the first mesh corner (primary circuit) and the validation is carried out using the procedure of section (2.5.1). Once validated the switch-word is stored and the process is repeated for all corners of the mesh substation. If the switch validation results and returns to the start point.

In the case of analogue test, the program, after validating the switch-data words, continues reading, validating and storing current and voltage values up to the number of cycles previously pre-set. The hardware configuration set up for the on-line tests assigns the first


Figure 25. Flow diagram for on-line data validation

channel of the multiplexers to the monitored current values and the second one to the voltage transformer, so the analogue validation starts checking the first three-phase current sample, continues with the first three-phase voltage sample and repeats the process for the number of cycles desired. The validation is done using the procedure of section (2.5.2) so during the current sample validation the presence of a non-zero value is used to correct a possible doubtful indication of the associated switch of the mesh corner. Then after the storage of each current sample the program also stored the switch-data word used during that validation. Once the program has completed its number of cycles, the results are printed out on the teletype and returns to the clock setting.

2.9.3 On-line test results

The software described in section (2.9.2) was used to detect and identify doubtful indication of the switch status and to validate current and voltage samples taken through the hardware configuration described in section (2.9.1). Many tests for different substation switching configurations and tolerance limits of the residual value were carried out.

Figures 26 to 29 show some typical results for the digital and analogue data validation algorithms.

The large unbalance present in all three-phase current and voltage samples was mainly due to:

- (i) the linear coupler outputs were different for equal inputs
- (ii) inequality in the turns-ratios of the voltage transformers
- (iii) different phase shifts produced by asymmetric assembling of the linear couplers in the model.

	CLC2E 110101 111111 101110 111111	CTEL 110702 620207 100610 620022	F L A B 1 7 1 0		CLCSE 111111 111111 111111 111111 111111	072N 022200 020200 020200 000000 000000	FL7.2 C Q Q
	• • ••••••				•		
	CLCSE 111111 110101 111111 101115	CPEI C20020 120001 C10005 000110	FLAG C 1 C 1		CLCSE 111111 111111 111111 111111 111111	CFEN: 510601 500007 700011 500000	FLA3 1 2 1 2
	•				· · · · ·		
	CLCSE 111111 111111 11C11E E111C1	CPEN C19000 E01002 E01010 F00110	FLAS C C 1 1	•	CLCST 111111 111111 111111 111111 111111	CPEN 200221 100214 002000 712210	FLAJ 2 1 5 1
,	• •	•		•	· · · · ·		
	CLCSE 115161 011110 111621 101110	CREN 102001 102001 001010 001010	FLAC 1 6 . 1 1		CLCSE 111111 111111 111111 111111 111111	CPEL C20101 100210 010010 010010	FLA-2 1 1 1 1
			•		•	•	
	CLCSE 11C171 121110 111021 11C171	CPEI 100001 100001 001010 00110 200110	FLAG 1 1 1 1		CLCSE 111111 111111 111111 111111 111111	OPEN 101000 100010 000000 800000	FLAG 1 1 C 2
	•						•
	CLCSE 011111 011101 011101 011111 011110	CPEN 200083 210101 202781 202781 201771	FLAG I I I I	ł	CLCSE 111111 111111 111111 111111	CPEI: 5255200 - 555522 515511 254615	FLAG © C 1 1

Figure 26. On-line substation switch status validation tests.

CLCSE 111111 111111 111111 111111 111111	1350 13522 13522 13522 13522 13522	N FLAC AA C CC C AC C AZ Q					
PLASE P Y B	EF EF C 1 1 Q 1 C	neasurli elt escissiiii escesaese creeseese creescess		PFASE T Y U	27 0 0 0	EF 1 6 5	NEASURENENT THICLERCT CCCCCCTCCC CCCCCTCCC
CLCSE 111111	OPEI CCSI	; Flag 220 [.] 8	•	CLC52 11111	L ·	CPEL Cocc	FLAG Ef Ø
PHASE P Y E	2F EF C 1 C E 1 E	LLASUTELEUT 1111163661 1111161666 1111111111		FRASE R Y E	2F 6 C 2	EF 1 C C	HEASUREHEIT 1111101100 GEGELEICO1 CCCCCLCCL1
PLASE R Y E	77 27 2 1 6 (2 2	NEASUFELTUT SECISIISII 1111115111 1111115115		FPASE R Y E	2F C C C	ef 1 C C	LEASUTERINT CCECEICICO CCECEILIIC CCECEILIIC CCECEILIIC
CLCSE 111111	1770 1770	5 FLAG 127 7		CLCEE 111111	l	CFER CZCS	7145 80 c
PHASE R Y E	EF EF C C C C C C C C	112ASUR11.201 5262512100 1111101111 5262621621		FRASE R Y Z	12 C C G	1F 1 0 2	UEASUTEDELT 1111150551 1111155551 1111155551 111111515
PHASE R Y E	27 EF C 1 1 0 0 C	132250725527 6666111162 111111116 11161616		PFASZ R Y B	2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	1F 	11225072021.T 2021211211 111112201 111112202
CL CSE 111111	CP21 5227	FLAG		CL CSE 111111		0971: 2222	7İAS Se s
FEASE F Y B	27 27 2 1 2 2 2 2 2 2	NEASURENEN T 1111110101 1111010001 111010001		diaçe Ç E	zf C Z Z	27 0 0 0 0	NEASUFENELT FEFERICIIC IIIIICIIII COCCECTEEI
PHASE E Y E	27 27 2 1 0 0 C 2	LEASUFENENT 1101101101 1111000102 1110110110	·	PYACE E E	77 C 1 C	EF 1 C 2	LEASUFENENT CC62115615 111111116 1115155111
CLCSE 111111	CFEN SSCC	FLAG Ce c		CLCSE 111111		OPEN Cecci	FLAG 20 2
PIIASE P Y E	7F LF 8 1 6 6 8 0	HEASURENENT 2200701121 0000101101 1111121121		PHASE P Y B	77 6 7 2	E7 1 7 5	NEASUTENENT 1111110001 1111010010 1110101000

· Figure 27. On-line analogue data validation with four bits tolerance and four samples/cycle

CL (cE 111111 111111 111111 111111 111111	11570 11570 11570 125700 125700 125700 125700 1257000 125700 125700 125700 125700 125700 125700 125700 125700 125700 125700 125700 125700 125700 125700 125700 1257000 125700 125700 125700 125700 125700 125700 125700 125700 125700 1257000 125700 125700 125700 125700 1257000 125700 125700 125700 125700 125700 1257000 125700 125700 1257000 1257000 125700 1257000 1257000 1257000 1257000000000000000000000000000000000000	FLAC C C C C C C C C C	•					
FIIASE P Y S	2F EF C 1 C C C C C C	11275UF ENENT 1111160111 0000001101 2001101100		•	FHASE P Y B	77 C C Q	E)* 1 2 2	NEASURENEL T FEETECSELL EFEETECLELL SECELLELEL
CLCSE 111111	CPZI: ZCZCI	FLAG SC C			CLCSE 111111		CPEL CCCC	FLAG S.2 C
FFASE E Y B	ZF EF C O Ø C I C	MEASUPENENT 111111065 2000110205 2000200211			PEASE R Y E	₹F © © ©	ef C C C	4)EASURLHELT 1111111253 6263012623 6666126111
PHASE F T E	77 EF C 1 C 2 S C	1.EASUPENER T 1111101010 6026601110 6001101160			PHASE R Y B	77 © © Q Q	ef 1 6 0	NEASUNEMENT BOSICIICII BSSICCICII SSRCSICICI
CLCSI 111111	C721) C326	FLAC EC C			CLCSE 111111		CPEII 2220	FLAG FC C
FIASE R Y B		NEASUT LULLT 1111113122 CZEC181C19 CZC2821121			PFASE R Y E	27 1 0 0	2F 	NEASUPENENT 111111111 8080050102 8002100110
PHASE T Y E	2F 2F C 1 2 C 0 C	<u> </u>		•	PHASE B Y E	₹7 © C ©	е 1 С	NEASUELNENT SCC1100101 SCC2000111 1111110101
CLCSE 111111	CPLI CCCC	FLAC 2C 2			CLCSE 111111		CPII. SCSS	ELAG CC E
PFASE E Y D	2F ZF 2 2 2 2 2 2 2 C	NEASURENELT 1111112011 2008100080 0807011010			PHASE R Y E	77 6 1 6	ef C C C	NEASUR EMER. T ECCESSIERI 111111162 ECCE1261C1
PM.SZ E Y B	27 EF C 1 C 2 C 7	NEASUREMENT 6007107111 0007001100 6061001111	-	•	FRASE R Y E	2F 2 2 2	ef 1 2 2	NEASUTENENT 6201101010 2000000111 1111010111
CLCSE 111111	6921 6 002	: FLAG 186 C			CLCSE 111111		CPE!! CCCC	FLAG CC ©
FLASE F Y B	27 EF 5. C 6. 2 6. G	MEASURENEL 1111110100 COCFEI12EC ECCEICECEI	r .		FLASE F Y D	2F C C	EF C C C	NEASUFENERT CCCCC1111 11111C1CC CCCC1CCC11

Figure 28. On-line analogue data validation with 6 bits tolerance and 16 samples/cycle

GLCSE CPFL FLAS 111111 100000 1 111111 000000 0 111111 000000 0 111111 000000 0	•
FRASE FF EF REASUREMENT	FRASE PF EF MEASUREMENT
F G Ø ØCCICCOOCC	R 0 0 1110162010
Y I C ECCOCCOCC	Y 0 0 200001000
E C Z III2IECEEE	E 0 0 000001010
CLCSE CFER FLAG	GLCSE CPEN FLAG
111111 CE2222 1	MININ SECTOS I
PHASE RF EF MEASUTEMENT	FRASE 27 EP MEASURELELT
P C G GECCIFCEIS	R C C RECECTORIE
Y R C HINNOICE	Y C C RECECTORIE
E C COCCCEICEC	E C R HITTRICCI
PHASE PF EF LE/SUBLED.T F C CATCHINGC T Y 1 C THINNIN E 2 C THICTNICE	FRASE EF EF MEASUREMENT R Ø 2 litslegge Y C Ø 200001010 E V C RECISCOLO
CLOSE CPEN PLAG	CLCSE CFEN FLAG
111111 COSCES I	111111 C220C2 I
PHASE OF ZF NEASUTEMELT	PLASE 27 EF NEASUTENELT
R C C COCOLOLICO	R 1 2 SOCOCC22C/
Y C C INTINICALO	Y E S SOCOCC21111
E C C INTINICALO	B C 2 1112111010
PHASEZFEFNEASURENENTP-15C002200001Y10C200000000E021110100000	FRASE EF EF MEASURENT E C 1111001011 Y C CCCCCC1101 B C CCC1100011
CLCSE OFEN FLAG	CLOSE CFEN FLAG
111111 SECCCE 1	111111 SECCE 4
PHASE2FEFNEASUREMENTPCCCCYIZCCSQQ1111011001	PHASE EF EF MEASUFINENT R C Z 1111110100 Y C & Ceccicici B Z & CECCCCICII
PHASE 2F EF NEASUREMENT P C C 1111000111 Y 1 0 111111101 E C 0 1111201000	PHASE 2F EF MEASUMEMENT P C C CCCCCCIII Y C C CCCCCCIII . E C C CCCCCCIIC
CLOSE CPEN FLAG	CLOSE OPEN FLAG
111111 CEECEC I	111111 Sector 1
FLASE EF EF ILLASUFLMENT	FRASE RF EF MEASUPEMENT
R 2 6 6265126167	R C C IIIIIISICF
Y 6 6 6267611121	Y G G SCICCIICI
E 2 2 1111166126	E C C COCCIIIII

Figure 29. On-line analogue data validation with 7 bits tolerance, 16 samples/cycle illustrating the use of the sample/swith validation

2.10 Final comments

The use of two complementary bits for each switch position and a sample-switch routine provides a simple and reliable validation for any doubtful switch indication.

Zero data detection must be considered within the hardware design to allow validation routines to be carried out during the sampling interval. The hardware identification (zero flag) of zero data and the information of the switch adjacent to the value being checked, provide simple detection of missing data from all 3 phases. The use of the residual value and a residual counter allows the detection of missing data in one or two phases and the presence of errors bigger than a required tolerance. The self-checking criterion provided for analogue data validation gives great flexibility to the validation scheme.

The use of two data flags (zero and error) with any type of redundancy available for each measurement, enables use of a simple estimation routine to provide replacements for missing or erroneous data. The estimate values are transferred via a first-in, first-out buffer to assemble the relaying data tase in the control and protection processor. The execution time obtained for the one corner meshsubstation model shows the feasibility to apply the data validation algorithm proposed in this chapter, in such model, with a sampling rate up to 16 samples per cycle. The additional memory requirements of the validation program for a computer relaying scheme are modest.

CHAPTER III

SYSTEM RELIABILITY ANALYSIS

3.1 Introduction

In system reliability analysis, the system is normally modelled by a probabilistic graph in which each node and each edge has an assigned weight which is the probability of being good. A basic measure in system reliability for relay networks⁽¹⁵⁾ and process systems⁽²³⁾ is the terminal-pair reliability but for communication networks several other reliability measures based on node and edge connectivity⁽³⁷⁾ are also important.

This chapter will briefly review known procedures for the terminal-pair reliability calculation and the limitations that arise in their application. A comparison is made of the approximation techniques for the minimal cut-set approach. Only the case where the failure probabilities of the elements are statistically independent is considered.

3.2 Terminal-pair reliability problem

The terminal reliability with respect to a pair of nodes s and t, here called the system reliability, is the probability that there exists at least one path from s to t along which all nodes and branches are good. In performing such reliability analysis the logical approach is to decompose the system into functional entities composed of units, subsystems or components. The subdivision generates a reliability graph where the system components are replaced by single lines here called elements, branches, edges or links and their terminals are called nodes. Models are then formulated to fit this logical structure and the calculus of probability is used to compute the system reliability in terms of component reliabilities.

When the reliability graph is series-parallel with respect to the terminal pair (s,t) and all nodes are perfectly reliable, their reliability can be evaluated very simply⁽¹⁷⁾. It is obvious that the structural nature of practical systems will generally be such that neither a pure series nor a pure parallel reliability model is appropriate. The reliability analysis when all components have identical reliability and all nodes are perfectly reliable or vice versa becomes a combinatorial problem for which some methods are available^{(15),(38)}. For the general case of non-series parallel networks with unequal different reliabilities the problem is more complicated and more general techniques are needed. The following methods can be used for the terminal-pair reliability problem:

- (18),(19),(20) 1. Event enumeration
- 2. Factoring theorem^{(21),(22)}
- 3. Tie-set enumeration⁽²³⁾, (24), (25), (26), (27)
- 4. Cut-set enumeration (30), (31), (32), (33), (34)

3.3 Event enumeration

In order to calculate the system reliability it is necessary to find all mutually exclusive events which result in successful system operation. The basic approach to apply the event enumeration method requires a table of all possible logical occurrences in the system, called the reliability table. The table is then separated into successful and unsuccessful events and these tables are called partial reliability

tables. If the tables are properly prepared all the events are mutually exclusive. The probability of success is then merely the sum of the occurrence probabilities of each successful event. The reliability could also be computed by first finding the probability of failure, which is given by the sum of the occurrence probabilities of the unsuccessful events.

The technique can be illustrated using the reliability graph of Fig. 1. The event-enumeration procedure consists of constructing a table⁽¹⁸⁾ with k event-groups, k = 0, n where n is the number of elements in the system.

If elements are represented by their own designation when they work properly and by its complement if they fail, then group 0 of Table I represents no failures, and the number of terms in this group is given by ${}_{0}C_{3}$ Group 1 represents one failure and contains ${}_{1}C_{3}$ elements. The success of an event may be determined by inspecting the reliability graph as follows: the failed units in the event in question are deleted from the graph, and if the remaining structure has at least one continuous path connecting nodes s and t, the event is successful. All the circled events in Table I are unsuccessful ones, and the uncircled ones are all successful. Thus the probability of successful communication between nodes s and t is given by

$$P_{S}(s,t) = Pr \{ E_{1} + E_{2} + E_{3} \}$$
 (3.1)

Since all these events are mutually exclusive, the probability of the union of these events is the sum of the probabilities of each event taken separately. If we assume that all units are identical, and that the units have a probability of success p and a probability of failure 1-p, Eq. (3.1) reduces to



Figure 1. Reliability graph







$$P_{S}(s,t) = p^{3} + 2p^{2} (1-p) = 2p^{2} - p^{3}$$

Dubes⁽¹⁹⁾ presents a procedure to simplify the reliability expression when all the elements have unequal probabilities of success based on successive applications of the theorem

$$p_a = p_a \cdot (1-p_b) + p_a \cdot p_b$$

for which the partial reliability table of successful events is assumed to be available. Wing and Demetriov⁽²⁰⁾ report a method to check automatically if there exists at least one path from every node to every other for each of the events of the complete reliability table. This method also has the advantage that when all the events have been examined, all the terminal reliabilities are obtained.

Event enumeration methods are exhaustive algorithms where the total number of events is given by 2ⁿ, n being the number of elements in the system. This technique has great computational difficulties for large systems. Systems with more than 20 elements are very difficult to solve even with a digital computer because they have over one million events.

3.4 Factoring theorem

Although the series-parallel technique^{(17),(21)} is extremely useful for the reliability analysis of a large class of redundant networks, not all reliability diagrams can be reduced to the seriesparallel model. For example, bridge-type networks cannot be decomposed into series-parallel elements.

For redundant networks the theorem⁽²²⁾ may be stated in the following form:

"A redundant network reliability function is equal to the reliability factor of any one single element of the network times the resulting

network reliability function with the terminals of the element short-circuited, plus the unreliability factor of the element times the resulting network reliability function with the terminals of the element open-circuited".

The probability of successful communication between a pair of node s and t can be expressed by the factoring theorem as follows:

$$P_{S}(s,t) = p_{j} \{ P_{S}(s,t) \}_{j=0} + (1-p_{j}) \{ P_{S}(s,t) \}_{j=1}^{p}$$
(3.2)

where p_j is the probability that the jth element of the network will not fail, $\{P_S(s,t)\}_{\substack{p_{j=1}\\j=1}}$ and $\{P_S(s,t)\}_{\substack{p_{j=0}\\j=0}}$ denote the probability of successful communication between nodes s and t assuming that element j is replaced by an open-circuit and a short-circuit, respectively. In complex reliability calculations it is often necessary to apply the factoring theorem in several steps to break up the system into smaller units of series and parallel networks for a systematic evaluation⁽²¹⁾ of the network reliability function. Figure 2 shows an example of the application of the factoring branches. Misra⁽²¹⁾ uses different combinations of the states of the interconnecting branches to define corresponding series-parallel networks reducing the total number of combinations in networks with non-adjacent interconnecting branches.

The algorithms based on an iterative application of the factoring theorem for all non series-parallel elements are very effective for networks containing few non series-parallel elements between any pair of nodes. However, for highly interconnected graphs (non series-parallel graphs) with b-branches and n-nodes, the required computation time can be shown to grow approximately as 2^{b-n}. The method is not applicable















when nodes as well as branches are unreliable.

3.5 Tie-set enumeration

One of the steps in the event enumeration technique suggests that a simplest way to find the reliability expression is to locate the successful paths at the outset. Then each successful path forms a favourable event and the union of these events gives the reliability expression.

3.5.1 Problem formulation

A path or tie-set is a group of branches which form a connection between input and output when traversed in a stated direction. A minimal path is a path which contains a minimum number of elements. If no node is traversed more than once in tracing a path, the path is minimal.

The probability of successful communication between nodes s and t is defined as the probability of successful operation of all elements in at least one tie-set and is given by

$$P_{S}(s,t) = F_{r} \left\{ at least one path is good \right\}$$
$$= Pr \left\{ \bigcup_{i=1}^{m} T_{i}(s,t) \right\}$$
(3.3)

where m is the number of tie-sets between nodes s and t and U denotes the union.

For the oriented graph in Fig. 3 there are 5 tie-sets between nodes s and t given by the following combinations of elements $T_1 = (e_1e_4e_7), T_2 = (e_1e_3e_5e_7), T_3 = (e_1e_3e_6), T_4 = (e_2e_5e_7)$ and $T_5 = (e_2e_6)$. Thus $P_S(s,t)$ is given by Eqn. (3.4) where the + indicates the logic sum or union.

$$P_{S}(s,t) = Pr \{ T_{1} + T_{2} + T_{3} + T_{4} + T_{5} \}$$

= $Pr \{ e_{1}e_{4}e_{7} + e_{1}e_{3}e_{5}e_{7} + e_{1}e_{3}e_{6} + e_{2}e_{5}e_{7} + e_{2}e_{6} \}$
(3.4)

Alternatively, the probability $P_f(s,t)$ of failure between any pair of nodes (s,t) is given by

$$P_{f}(s,t) = Pr \left\{ all \text{ paths have a failure} \right\}$$
$$= Pr \left\{ \bigcap_{i=1}^{m} \overline{T}_{i}(s,t) \right\}$$
(3.5)

 \cdot where \cap indicates the intersection.

3.5.2 Network representation

As a path is an ordered sequence of branches between the source and sink terminals, each tie-set can be associated with a subnetwork of a series arrangement of the elements of the ith tie-set. Since the communication between s and t is successful if and only if all of the elements in at least one tie-set work the network may be represented as a parallel arrangement of the minimal tie-sets. An example is shown in Fig. 4 for the network in Fig. 3. The probability of a failure between nodes s and t of Fig. 3 is given by the probability that all 5 tie-sets have a failure, that is, for each set at least one element fails.

3.5.3 Methods of solution

Algorithms based on direct expansion of Eqn. (3.3) involve the probability of the union of m-paths which are generally not mutually







Figure 4. Minimal tie-sets between nodes s and t of Fig. 3

exclusive. Such evaluation⁽¹⁸⁾ contains 2^{m} -1 terms. The probability of the intersection of m events as given by Eqn. (3.5) is expressed as the joint product of one independent probability and (m-1) dependent probabilities. Evaluation of dependent probabilities can become quite cumbersome; e.g., in the evaluation of Eqn. (3.3) there are 2^{m} -(m+1) dependent probabilities involved. Okada⁽²⁹⁾ has shown that the number of s-t minimal paths in a connected graph may be as large as 2^{b-n+1} . Systems of moderate complexity are very difficult to solve even with modern digital computers.

Improved methods have been developed which reduce the number of terms of the reliability expression $P_S(s,t)$ either using Boolean algebra^{(25),(26)} to generate mutually exclusive events from the tie-sets or by a special term-reduction technique based on zero-valued terms of conditional probabilities⁽²⁷⁾. A few of the more important results of these methods are now presented.

Brown⁽²⁵⁾ solves the problem by generating the reliability expression from the Boolean algebra transmission function constructed by enumerating all the success between the input and output terminals of the system. Instead of generating the $2^m - 1$ events required by the event-enumeration technique he generates directly only the successful events from the binary representation of the success paths. The basic principle that must not be violated when translating and combining the elementary Boolean events into the probability function is never to include a given elementary event more than once. The algorithm is easily programmed to obtain all mutually exclusive events but it has two drawbacks for large systems:

 It assumes that the task of storage and enumeration of the various paths of system success is nominal compared to that of obtaining the reliability function. However, the 16-element redundant system

of Fig. 5 has 55 success paths which are increased to 418 if an element is added between each pair of connected nodes. The resulting system has just 26 elements.



Figure 5. A 16 - element system

To solve the system of Fig. 5 with this algorithm requires the generation of 163840 binary numbers stored in a matrix (163840 x 16) which is extremely large. To overcome this problem it is advantageous before proceeding to find the success paths to combine all parallel elements between common nodes i and j using Boolean algebra rules and replace them by an equivalent link⁽²¹⁾. With this process the system of Fig.5 is reduced to the system in Fig. 6 which has only 10 elements and 6 success paths instead of 16 and 55 respectively.



Figure 6. Equivalent to system in Fig. 5

To solve the system of Fig. 6 only 320 binary numbers are required stored in a matrix of dimension (320 x 10) compared with a matrix (163840 x 16) for Fig. 5. Having reduced the redundant elements the success paths can be found by any of the several path-finding algorithms now available in the literature (28).

2) A second drawback for large systems is that each transformed term is compared with each succeeding binary number up to 2^n-1 where n is the number of elements. The number of binary number comparison doubles with each additional component. For instance, a 10-element system would require 1024 binary number to be generated and evaluated by the computer, whereas an ll-element system would require 2048, and a 15-element system 32768. The problem becomes very serious for systems of 20 to 30 elements. The inefficient comparison digit by digit of each number with the succeeding numbers up to 2ⁿ-1 is overcome by generating a restricted set of numbers directly from the binary representation of each success path taking advantage of the nature of binary numbers. This procedure reduces the 1024 binary numbers required for the 10-element system in Fig. 6 to 320 binary numbers. However an upper bound for the binary number comparisons is given by $(n_b-1)!$ where n_b is the total binary numbers required by the algorithm.

Reductions greater than 60% of the number of terms are obtained using a simplification procedure based on the theorem $p_A \cdot q_B + p_A \cdot p_B = p_A \cdot$ In general higher reductions (up to 80%) are obtained when an ascendent or descendent ordering is applied to the terms of the partial reliability table before applying the simplification procedure. The use of an ascendent or descendent ordering does not add significantly to the execution time. Table II shows the results for several networks (see Appendix 1) with different number of elements and success paths. The times do not include success path generation.

Fratta and Montanari⁽²⁶⁾ use the same approach of reducing the Boolean algebra transmission function $F = \bigcup_{i=1}^{m} T_i$ to a sum of disjoint products by performing simple Boolean algebraic operations on the initial form obtained from the set of all tie-sets. For each term of F they perform the operation $F \leftarrow \overline{T}_i$. $(F - T_i)$ and reduce $P_S(s,t)$ to a sum of mutually exclusive products. The approach uses the well-known identity $\overline{e_1e_2} = \overline{e_1} + \overline{e_2}$ for computing F and then uses distributivity. The method has no space limitations for its implementation but is time limited even for systems having less than 20 paths due to the large number of fundamental products of F and the number of symbolic multiplications required. They recognised the deficiency mentioned in Brown's method by incorporating a series-parallel reduction procedure and a path-finding algorithm as the first and second steps in the implementation of their algorithm.

Lin et al⁽²⁷⁾ present a modification for the direct expansion of Eqn. (3.3) expressing the joint probabilities of success paths as conditional probabilities. For example, $P_S(s,t)$ in a probabilistic graph having only 2 paths (P_1, P_2) from s to t is given by

 $P_{S}(s,t) = p_{1} + p_{2} - p_{1,2}$

$$p_1 + q_{1(2)} p_2$$

where $q_{1(2)} = 1 - p_{1(2)}$, $p_{1}(2)$ being the probability that p_{1} is good under the condition that P_{2} is good and p_{1} and p_{2} are the success probabilities of paths P_{1} and P_{2} respectively. For the general case of m paths, the explicit expression for $P_{S}(s,t)$ by the present method has 2^{m-1} terms

TABLE II

τ

· •

COMPARISONS OF DIFFERENT ORDERINGS OF THE BINARY TERMS IN THE PARTIAL RELIABILITY TABLE

Network	Number of Number of Success elements		PARTIAL RELIABILITY TABLE		ARBITRARY ORDERING		ASCENDENT ORDERING			DESCENDENT ORDERING				
(App 1).	Paths		Terms	Multipl- ications	Time (sec)	Terms	Multipli- cations	Time (sec)	Terms	Multipl- ications	Time (sec)	Terns	Multipl- cations	Time (sec)
1	3	4	8	31	0.450	3	11	0.482	3	11	0.485	3	11	0.484
2	3	5	17	84	0.516	5	24	0.537	5	24	0.543	4	19	0.537
3	4	6	19	113	0.609	5	29	0.634	5	29	0.646	5	29	0.648
· 4	6	7	58	405	1.096	13	90	1.351	8	55	1.383	10	69	1.408
5	8	7	61	426	1.345	10	69	1.388	10	69	1.414	10	69	1.482
6	8	7	82	573	1.854	14	97	1.885	9	62	1.977	11	76	1.996

which is about half of the number from the direct expansion of Eqn. (3.3). Fortunately the q-factors usually produce zero-valued terms which do not need to be generated thus reducing the actual number of terms. The method for systems having fewer than 20 paths has shown⁽²⁷⁾ to reduce the number of terms of the reliability expression up to the order of two or three hundred instead of the 2^{20} -1 terms given by direct expansion of the probability of a union.

3.6 Cut-set enumeration

One way to alleviate the computational difficulties of the event enumeration method is by using a technique which requires the enumeration of a much smaller set of system states. These states are called minimal cut-sets.

3.6.1 Problem formulation

A cut-set of a graph can be defined as a group of elements that, when they fail, the system fails regardless of the condition of the other elements in the system. A cut-set is minimal when there is no proper sub-set of its elements whose failure alone will cause the system to fail. The elimination of any element of a minimal cut no longer makes it a cut. This is because a nonminimal cut corresponds to more elements than are required to cause system failure. The minimal cut-sets are the group of distinct cut-sets of the system containing a minimal number of elements. All system failures can be represented by the removal of at least one minimal cut-set from the graph.

Consequently, a cut-set with respect to a specific pair of nodes s and t in a connected network, called an s-t cut is such that its removal destroys all paths between nodes s and t. Therefore, the probability of failure between any pair of nodes s and t is given by the probability that at least one minimal cut-set fails. If C_j represents the jth minimal cut-set and \overline{C}_j the failure of the jth cut-set the preceding statement can be expressed as follows:

$$P_{f}(s,t) = Pr \{ at least one cut-set occurs \}$$

= Pr $\bigcup_{j=1}^{M} \overline{C}_{j}(s,t)$ (3.6)

where m is the number of minimal cut-sets. Similar to the path-enumeration method, $P_f(s,t)$ is not readily calculated from Eqn. (3.6) because the $\overline{C}_j(s,t)$ are not mutually exclusive events. On the other hand, the terminal-pair reliability is given by

$$P_{s}(s,t) = 1 - P_{t}(s,t)$$
 (3.7)

or alternatively

$$P_{S}(s,t) = Pr \left\{ \begin{array}{l} all cut-sets are good, i.e. at least \\ one element of the set is working \end{array} \right\}$$
$$= Pr \left\{ \begin{array}{l} \bigcap_{j=1}^{m} C_{j} \end{array} \right\}$$
(3.8)

3.6.2 Network representation

The system may have a large number of cuts and a particular element may be in more than one of them. An example is shown in Fig. 7. As long as any success path exists between terminals 1 and 4 of the network, the network is said to be successful. An element

failure opens the path between the two terminals of the component. The cuts of this network are listed in Table III and the numbers of the elements describe the cuts. The failure of any one of these cuts will cause the network of Fig. 7 to fail. The minimal cut-sets between nodes 1 and 4 of Fig. 7 are listed in Table IV along with their probability of occurrence. q_i is the probability that the ith element fails. Note



Figure 7. Bridge network



Figure 8. Parallel arrangement of elements in the minimal cut-sets of Fig. 7



Figure 9. Series arrangement of the minimal cut-sets of Fig. 7

that cut 2 of Table III is not a minimal cut since it is contained in cut 1 which is a minimal cut.

Each minimal cut-set C_j can be associated with a subnetwork of a parallel arrangement of the elements of the jth cut-set. These subnetworks for the network of Fig. 7 are shown in Fig. 8. Since the network fails if, and only if, at least one of the minimal cut-set fails, the network of Fig. 7 may be represented as a series arrangement of the minimal cut subnetworks of Fig. 8 as is shown in Fig. 9.

3.6.3 Methods of solution

Direct expansion of Eqn. (3.6) involves, as in the path-enumeration method, the probability of a union which contains 2^{m} -1 terms. Bellmore and Jensen⁽³³⁾ have shown that the number of s-t minimal cut-sets in a n-node network is bounded by 2^{n-2} . The computation for systems of moderate complexity is clearly infeasible even with large computers. This situation has led to the development of techniques either to generate mutually exclusive sets of cutting states⁽³⁴⁾ or to provide lower and upper bounds approximation to system reliability. Shooman⁽¹⁸⁾ and Barlow and Proschan^{(35),(36)} provide good mathematical background material and give many references to previous work in the area of reliability approximation.

Nelson et al^{(30),(31)} provides an algorithm for system reliability with unidirectional elements based on the concepts of success paths and cut-sets. Jensen and Bellmore^{(32),(33)} provide an algorithm for determining the reliability of a complex network with bidirectional elements. A single application of the algorithms of both papers gives the minimal cut-sets between two nodes of a network. A few of the more important results of these papers will now be presented.

TABLE III

Cut	Elements in Cut
1 2 3 4 5 6 7 8 9 10 11 12 13 14 15 16	1,2 1,2,3 1,2,4 1,2,5 1,2,3,4 1,2,3,5 1,2,4,5 1,2,3,4,5 1,2,3,4,5 4,5 3,4,5 2,4,5 1,4,5 2,3,4,5 1,3,4,5 1,3,5 2,3,4

Cuts between nodes 1 and 4 of Fig. 7

TABLE IV

Minimal cuts between nodes 1 and 4 of Fig. 7

Minimal Cut	Elements in Cut	Probability of failure of minimal cut
Çl	1,2	^q 1 ^q 2
C2	4,5	^q 4 ^q 5
C3	1,3,5	q1q3q5
C4	2,3,4	^q 2 ^q 3 ^q 4

The Nelson et al⁽³⁰⁾ approach provides bounds for system reliability based on the concepts of success paths and cut-sets. A listing of the elements of the system, their predecessors, and the probability of successful operation of each element are the inputs of the computer program⁽³¹⁾. The outputs are the success paths, the cut-sets, and a series of upper and lower reliability bounds; these bounds are based on the inclusion-exclusion method of section 3.9 The algorithm for determining the cuts from the success paths is based on Boolean logic which is simple to understand. The algorithm is easily implemented for computer analysis of system reliability but has two distinct drawbacks in practice. First, it needs to determine and store the success paths. As mentioned before if we add one element between each pair of connected nodes in Fig. 5, we have a system with 418 success paths instead of the original 55 by just increasing the network from 16 to 26 elements. The use of the success paths as they are determined by Nelson et al requires operation with a matrix of dimension (418 x 26). Second, to generate the cuts of order k (k > 1) in a network of n elements it is necessary to cover all the k logic sums or unions between the n column vectors of the path matrix i.e., ${}_{k}C_{n}$, and this is repeated until all "possible" cuts of order 1,2,...,n have been exhausted. Thus, it is necessary to calculate

$$S = \sum_{k=2}^{n} C_{kn} = 2^{n} - (n+1)$$

logic sums in this procedure. (e.g. n = 10 S = 1013; n = 20 S = 1,048,555). Besides, after a possible cut of order k is identified, it needs to be checked against all cuts of lower order using Boolean algebra for intersection (AND operation) i.e. for a network of m cuts, m (m+1)/2 checks must be done at the end of the minimal cut-set generation (e.g. m = 100, 5000 checks). This is a serious problem to handle if we consider that,

depending on the configuration of the network, the number of minimal cuts between two nodes (IN and OUT nodes) varies from n-1 to 2^{n-2} . There are n-1 cuts if the network is a simple chain and the IN and OUT nodes are the ends of the chain and 2^{n-2} cuts if there is a component between each pair of nodes. For instance a network of 12 nodes will have, in the second case, 66 elements and 1024 minimal cuts.

Jensen and Bellmore (32) provide an algorithm for determining the reliability of a complex system in which all components have two terminals and are bidirectional i.e. a path may traverse it in either direction. For a n-node network, source(s) and sink(t) nodes are numbered 1 and n respectively, the numbering of all other nodes is arbitrary. They solve the problem of generating the set of minimal cuts as the problem of generating the set of two-part partitions of a set N into two mutually exclusive subsets X and \overline{X} . The subset X must define a connected subnetwork that includes the source node s. The subset X must define a connected subnetwork that includes the sink node t. The minimal cut corresponding to a two-part partition is the set of components with one terminal in X and one terminal in \overline{X} . The absence of an unscanned node signals termination of the algorithm. They use a lower bound approximation⁽³³⁾ based on the disjoint property and their algorithm is highly efficient for networks of single input-output nodes (source and sink nodes) i.e. finding the minimal cut-sets between source and sink nodes in networks of two terminals.

Hansler et al⁽³⁴⁾ have developed a procedure to iteratively calculate a minimal set of mutually exclusive events containg all minimal cut-sets and summing the probabilities of these events produces the probability of service interruption $P_f(s,t)$ between the specified pair of nodes. The procedure starts with the minimal cut-set consisting of the branches incident to node t. Subsequently, these branches are reconnected in all combinations and we then cut the minimal set of branches adjacent to those that lie on a path to node s. The branch replacements are repeated until the set of branches connected to node s are reached. The procedure has been found to be particularly effective for graphs of small diameter. Furthermore, a very small amount of storage is required since each event is generated from the previous one but due to the inherent computational complexity of the problem, the computation time grows exponentially with the system size.

3.7 Selection of the method

The event enumeration method can be used for all systems but considering that the number of events is equal to the total number of combinations of n units i.e., 2ⁿ the method is impractical except for extremely simple systems.

Breaking down the network structure into a series of elementary ones by an iterative use of the factoring theorem is efficient and simple when the system has few bridging elements. For highly non seriesparallel networks the labour involved increases rapidly. Further the technique is not applicable when nodes as well as branches are unreliable.

For the general case of complex networks (non series-parallel) with unreliable branches and nodes the path and cut-set methods are the most practical for system reliability evaluation. A simple set of comparative rules which can help to determine whether path or cut-set enumeration should be used can be formulated as follows:

1. In any network of b branches and n nodes, the order of the number of s-t cut-sets is 2^{n-2} whereas the order of the number of s-t paths is 2^{b-n+1} . For networks having nodes of

average degree (number of incident branches) greater than four, b > 2n and 2^{b-n+1} > 2^{n-2} . Consequently such networks have a larger number of paths than cut-sets and the computation is simpler using cut-set enumeration.

2. In both cases the terminal reliability measure is the probability of the union of the events corresponding to the existence either of the paths or of the cut-sets. These events are not disjoint and the complete expansion of the union becomes a hopeless task. Several approximations have been suggested (see section 3.8) to provide upper and lower bounds on system reliability using the path and cut-sets methods. The bounds based on the cut-sets are best in the high reliability region, and those based on the tie-sets are best in the low reliability region.

If the path method is used, computations can be simplified if parallel elements are reduced to a single equivalent element. Series branches need not be combined. If the cut-set method is used, series branch reduction simplifies the computation and parallel links need not be combined.

3.8 Approximations for reliability evaluation

Exact reliability analysis using the path or cut-sets approaches involve the complete expansion of the probability of a union which contains 2^m-1 terms. One way to alleviate this difficulty is to obtain approximate equations which provide bounds on the system reliability, making feasible the reliability evaluation of large networks. Two approximations which are useful in simplifying the computations of

Eqns. (3.3) and (3.6) are discussed in this section and a sample calculation is made for $P_{S}(1,4)$ in Fig. 7 assuming all components have equal reliability.

3.8.1 Mutually exclusive approximation

If the events are mutually exclusive, i.e., possess the disjoint property, the probability of a union is greatly simplified. Messinger and Shooman⁽³⁹⁾ have shown that for any set of events A, B, C the disjoint approximation gives an upper bound

$$Pr(A \cup B \cup C) \not\subset Pr(A) + Pr(B) + Pr(c)$$

Thus if the tie-sets are assumed to be mutually exclusive an upper bound R_{II} can be written for $P_{S}(s,t)$ as

$$P_{S}(s,t) \ll R_{U} = P_{r} \left\{ \sum_{i=1}^{m} T_{i}(s,t) \right\}$$

$$= P_{r} \left\{ \sum_{i=1}^{m} \left(\prod_{j \in T_{i}} p_{j} \right) \right\} \quad (3.9)$$

Similarly from Eqns. (3.6) and (3.7) of the cut-set approach a lower bound R_L for $P_S(s,t)$ is given by

$$P_{S}(s,t) \geqslant R_{L} = 1 - Pr \left\{ \sum_{j=1}^{m} \overline{C}_{j}(s,t) \right\}$$

$$1 - P_{r} \left\{ \sum_{j=1}^{m} \left[\prod_{i \in C_{j}} (1 - P_{i}) \right] \right\} (3.10)$$

Table V shows these bounds for P_S between nodes 1 and 4 in Fig. 7. The upper bound from the tie-sets becomes a good approximation in the low reliability region and the lower bound from the cut-sets is good in the high reliability region.

3.8.2 Independence approximation

Esary and Proschan⁽⁴⁰⁾ have shown that a computation of reliability representing the system as a set of m independent minimal paths acting in parallel gives an upper bound on actual system reliability. This upper bound is given by

$$P_{S}(s,t) \ll R_{U} = 1 - \prod_{i=1}^{m} (1 - \prod_{j \in T_{i}} p_{j})$$
 (3.11)

Similarly they proved that a reliability calculation considering the system as a set of m independently operating minimal cuts acting in series gives a lower bound on actual system reliability. This lower bound is obtained from

$$P_{S}(s,t) \geqslant R_{L} = \prod_{i=1}^{m} \left\{ 1 - \prod_{j \in C_{i}} (1 - p_{j}) \right\} \quad (3.12)$$

Table V presents these bounds for the probability of successful communication between nodes 1 and 4 in Fig. 7. Similarly to the results given by the disjoint property the bound from the tie-sets is a good approximation in the low reliability region while the bound from the cut-sets becomes good in the high reliability region. Note that the bounds calculated from the independence approximation are sharper than the ones from the mutually exclusive approximation.

3.9 The inclusion-exclusion method

Let \overline{C}_j describe the event that all components in minimal cutset C_j fail. Then system failures correspond to the event $\bigcup_{j=1}^{m} \overline{C}_j$ if the system has m minimal cut-sets and the system reliability is given by Eqn. (3.7).

Let

$$S_{k} = \sum_{\substack{1 \leq i_{1} \leq i_{2} \dots \leq i_{k} \leq m}} \Pr \left\{ \overline{C}_{i_{1}} \cap C_{i_{2}} \cap \cdots \cap \overline{C}_{i_{k}} \right\} (3.13)$$

Component	Disjoint a	pproximation	Independence	Exact	
reliability p	Lower bound Cut-sets	Upper bound Tie-sets	Lower bound Cut-sets	Upper bound Tie-sets	reliability P _S (s,t)
0.10	0.00000	0.02200	0.00265	0.02186	0.02152
0.20	0.00000	0.09000	0.03086	0.09309	0.08864
0.30	0.00000	0.23400	0.11227	0.21601	0.19836
0.40	0.00000	0.44800	0.25176	0.38183	0.34048
0.50	0.25000	0.75000	0.43066	0.56934	0.50000
0.60	0.55200	1.00000	0.61817	0.74824	0.65952
0.70	0.76600	1.00000	0.78399	0.88773	0.80164
0.80	0.90400	1.00000	0.90691	0.96914	0.91136
0.90	0.97800	1.00000	0.97814	0.99735	0.97848
0,92	0.98618	1.00000	0.98623	0.99884	0.98637
0.94	0.99237	1.00000	0.99238	0.99961	0.99243
0.96	0.99667	1.00000	0.99667	0.99992	0.99668
0.98	0.99918	1.00000	0.99918	0.99999	0.99919

 $\frac{\text{TABLE V}}{\text{COMPARISON OF BOUNDS FROM THE DISJOINT AND INDEPENDENCE APPROXIMATIONS}}$ $P_{S} (1,4) \text{ IN FIG. 7}$

÷

By the inclusion-exclusion principle (41) Eqn. (3.7) can be written as

$$P_{S}(s,t) = 1 - \sum_{k=1}^{m} (-1)^{k-1} S_{k}$$
 (3.14)

and bounds on the system reliability are expressed as

$$P_{S}(s,t) \ge R_{L1} = 1 - S_{1}$$
 (3.15a)

$$P_{S}(s,t) \ll R_{U1} = 1 - S_{1} + S_{2}$$
 (3.15b)

$$P_{S}(s,t) \gg R_{L2} = 1 - S_{1} + S_{2} - S_{3}$$
 (3.15c)

$$P_{S}(s,t) \ll R_{U2} = 1 - S_1 + S_2 - S_3 + S_4$$
 (3.15d)

and so on. Thus using the minimal cut-set approach the inclusionexclusion method provides successive lower and upper bounds on system reliability which converge to the exact system reliability. Table VI shows these bounds using four terms approximation for the probability of successful communication between nodes 1 and 4 of Fig. 7. In this simple example the bounds save no computation as a four term expansion gives the exact probability of terminal-pair success. In practice it may be necessary to calculate only a few S_k 's to obtain a close approximation and the computations can be stopped when the margin between two successive bounds becomes negligible. The bounds from the minimal cut-sets as shown in Table VI are good in the high reliability region.

Let T_i describe the event that all components in minimal path T_i work. Then system success corresponds to the event $\bigcup_{i=1}^{m} T_i$ if the system has m minimal tie-sets and the system reliability is given by Eqn. (3.3).

Defining

$$\mathbf{s}_{\mathbf{k}}^{\prime} = \sum_{\substack{\mathbf{1} \leq \mathbf{i}_{1} \leq \mathbf{i}_{2} \leq \cdots \leq \mathbf{i}_{k} \leq \mathbf{m}}} \Pr\left\{\mathbf{T}_{\mathbf{i}_{1}} \cap \mathbf{T}_{\mathbf{i}_{2}} \cap \cdots \cap \mathbf{T}_{\mathbf{i}_{k}}\right\}$$
(3.16)

the bounds on system reliability given by the inclusion-exclusion principle are

$$P_{S}(s,t) \ll R_{U1} = S_{1}'$$
 (3.17a)

$$P_{S}(s,t) \gg R_{L1} = S_{1} - S_{2}$$
 (3.17b)

$$P_{S}(s,t) \notin R_{U2} = S_{1} - S_{2} + S_{3}$$
 (3.17c)

$$P_{S}(s,t) \gg R'_{L2} = S'_{1} - S'_{2} + S'_{3} - S'_{4}$$
 (3.17d)

and so on. Thus using the minimal tie-set approach the inclusionexclusion method provides successive upper and lower bounds on system reliability which also converge to the exact system reliability. Table VI presents these bounds for $P_S(1,4)$ in Fig. 7 using a four term expansion. These bounds are a good approximation in the low reliability region. The bounds on system reliability given by Eqns. (3.15) and (3.17) are calculated simply only if statistical independence is assumed between components. This assumption has been used for the sample calculation in Table VI.

3.10 System reliability indices

During the operating period of a system, components may be deemed non-repairable or repairable. The non-repairable case is much the easier, but for digital protection systems which are intended to operate for 20 years or so, it is evident that all detectable failures will be
Component	Minimal c	ut-sets	Minimal t	Minimal tie-sets		
p p	Lower bound	Upper bound	Lower bound	Upper bound	P _S (s,t)	
0.10	0.00000	0.02152	0.02152	0.02153	0.02152	
0.20 ·	0.00000	0.08864	0.08864	0.08896	0.08864	
0.30	0.03029	0.19836	0.19836	0.20079	0.19836	
0.40	0.26272	0.34048	0.34048	0.35072	0.34048	
0.50	0.46875	0.50000	0.50000	0.53125	0.50000	
0.60	0.64928	0.65952	0.65952	0.73728	0.65952	
. 0.70	0.79921	0.80164	0.80164	0.96971	0.80164	
0.80	0.91104	0.91136	0.91136	1.00000	0.91136	
0.90	0.97847	0.97848	0.97848	1.00000	0.97848	
0.92	0.98637	0.98637	0.98637	1.00000	0.98637	
0.94	0.99243	0.99243	0.99243	1.00000	0.99243	
0.96	0 .99 668	0.99668	0.99668	1.00000	0.99668	
0.98	0.99919	0.99919	0.99919	1,00000	0.99919	

 $\frac{\text{TABLE VI}}{\text{COMPARISON OF BOUNDS USING THE INCLUSION-EXCLUSION APPROXIMATION} P_{S} (1,4) IN FIG. 7$

÷

repaired. Thus, for the reliability evaluation of digital protection systems we will assume that the system has reached a steady-state condition of failure and repair. The inherent high reliability achieved in digital hardware suggests that the use of the minimal cut-set approach will provide the best bounds on system reliability.

Two key indices in system reliability evaluation are the probability that the system has failed and the frequency of system failure. The former is obtained from Eqn. (3.6) and the frequency of failure⁽¹⁶⁾ for an m-cuts system is given by

$$f_{f} = \Pr\{\overline{C}_{1}\} \mu_{1} + \Pr\{\overline{C}_{2}\} \mu_{2} + \dots + \Pr\{\overline{C}_{m}\} \overline{\mu}_{m}$$

$$- \Pr\{\overline{C}_{1} \land \overline{C}_{2}\} \overline{\mu}_{1+2} - \Pr\{\overline{C}_{1} \land \overline{C}_{3}\} \overline{\mu}_{1+3} - \dots$$

$$+ (-1)^{m-1} \Pr\{\overline{C}_{1} \land \overline{C}_{2} \land \dots \land \overline{C}_{m}\} \overline{\mu}_{1+2+} \dots + m \qquad (3.18)$$
represents the sum of μ over all is $C \rightarrow C \rightarrow C$

where $\overline{\mu}_{j+k+1+\ldots}$ represents the sum of μ_i over all i $\varepsilon \ C_j \cup C_k \cup C_1 \ldots$, i.e., the sum of the repair rates of the components which belong to any or all the cut-sets C_j, C_k, C_1, \ldots

If λ_j and μ_j represent the failure and repair rates respectively of the jth component then the steady-state probability of the j-component being failed is

$$P_{jd} = \frac{\lambda_j}{\lambda_j + \mu_j}$$
(3.19)

and the probability of the system being in the state corresponding to the failure of the elements of the minimal cut-set C_i is

$$\Pr\left\{\overline{C}_{i}\right\} = \prod_{j \in C_{i}} P_{jd} \qquad (3.20)$$

Once the probability P_f that the system is failed and the frequency of system failure f_f has been calculated other quantities of interest can be obtained as follows:

Availability	A	=	1 - P _f	(3.21)
Mean cycle time	T	=	l/f _f	(3.22)
Mean up time	U	=	$(1 - P_{f})/f_{f}$	(3.23)
Mean down time	D	a	P _f /f _f	(3.24)

3.11 Final comments

Since each component is assumed to be in either of two states, working or failed, the task of determining the exact system reliability is computationally unfeasible for large systems because the number of possible states in 2^n for n-element networks.

Two approximations analysed suggest lower and upper bounds to system reliability that requires the enumeration of a much smaller set of system states. The analysis is based on the concepts of minimal cut-sets and minimal paths. These terms refer to the effects of element failure or success on operation of the network.

From the analysis and the sample calculations performed it can be concluded that the minimal cut-set approach is more attractive and efficient than the tie-set method for the reliability analysis of complex networks with highly reliable components. This is the case of digital protection systems.

CHAPTER IV

MINIMAL CUT-SET ALGORITHMS

4.1 Introduction

Digital hardware configurations are integrated by highly reliable elements. In chapter III it was shown that the minimal cut-set approximation gives bounds which are good in the high reliability region. The importance of any algorithm based on the minimal cut-set approximation is the procedure used to enumerate the set of minimal cuts. This chapter presents new algorithms for the minimal cut-set generation in networks with directed (1-way), undirected (2-way) or both types of elements. The case of both unreliable branches and nodes is considered.

4.2 Basic assumptions and definitions

To determine the reliability of a system, the system may be represented either by a reliability graph or by a reliability block diagram as illustrated in Figs. 1 and 2 respectively. A block in the reliability diagram will be called an element and the whole assembly will be called a network.

The reliability analysis treated in this chapter is based on the concepts of minimal cut-sets and coherent system. These terms refer to the effects of element failures on operation of the network.

A coherent system (40) is defined by the following four conditions:

 When a group of elements in the system has failed, causing the system to be failed, the occurrence of any additional failure or failures will not return the system to a successful condition.

- 2) When a group of elements in the system is successful and the system is successful, the system will not fail if some of the failed components are returned to the successful condition.
- 3) When all the elements in the system are successful the network is successful.
- 4) When all the elements in the system have failed the system has failed.

Since much of the study of system reliability is based on graph theory (42) basic concepts and definitions which are relevant to the present study are summarized in the following paragraphs.



Figure 1. Reliability graph No. 1

A graph G(N,E) consists of a set N of n nodes and a set E of b branches or edges. The number of edges incident on a node is called the degree $d(n_i)$ of node n_i . A node having no incident edge is called an isolated node and a node of degree one is called a pendant node or an end node. In the definition of a graph G = (N,E), it is possible for the edge set E to be empty. Such a graph, without

any edges, is called a null graph. Although the set E may be empty, the node set N must not be empty i.e., by definition a graph must have at least one node. A graph in which there exists an edge between every pair of nodes is called a complete graph.

A graph g is called a subgraph of G if all nodes and edges of g are in G and all the edges in g have the same node terminals as in G. Two (or more) subgraphs g_1 and g_2 of a graph G are edge disjoint if g_1 and g_2 do not have any edges in common. The union of two graphs $G_1 = (N_1, E_1)$ and $G_2 = (N_2, E_2)$ is another graph $G_3 = (N_3, E_3)$ whose node set $N_3 = N_1 \cup N_2$ and the edge set $E_3 = E_1 \cup E_2$. The intersection $G_1 \cap G_2$ of graphs G_1 and G_2 is a graph consisting only of those nodes and edges that are common to G_1 and G_2 . The ring sum of two graphs G_1 and G_2 , $G_1 \oplus G_2$, is a graph consisting of the node set $N_1 \cup N_2$ and of edges that are either in G_1 or G_2 , but not in both.

It is necessary to distinguish between two types of graphs, directed and undirected. In a directed graph some of the elements may not be able to allow flow in both directions. On the other hand, all elements in undirected graphs are two-way branches. An undirected graph is said to be connected if there is at least one path between every pair of nodes. Otherwise it is disconnected. A null graph of a single node is defined to be a connected graph. A digraph (di-rected graph) is said to be connected if its corresponding undirected graph is connected.

A cut-set in a connected graph G, is a set of edges whose removal from G leaves G disconnected. A cut-set is minimal provided no subset of the removed edges has the same property. For instance, in Fig. 1 the set of edges (1,2) is a minimal cut-set. The set of

edges (1,2,3), on the other hand, is not a minimal cut-set because one of its subset (1,2) is a cut-set. Removal of the edges of any cut-set in a graph G always separates the graph into two connected subgraphs producing a partition of all nodes into two mutually exclusive subsets Y and \overline{Y} . A proof of this last statement is given by Jensen and Bellmore⁽³²⁾. This property of the cut-sets will be used later to introduce the concept of the cut-node incidence vector. Then a cut-set can be seen also as the minimal number of edges with one terminal in Y and the other in \overline{Y} whose removal from G destroys all paths between these two sets of nodes.

Each cut-set of a connected graph G consists of a number of edges (order of the cut-set). The smallest order of the cut-sets is defined as the edge connectivity of G. Similarly, the node connectivity (or simply connectivity) of a graph G is defined as the minimum number of nodes whose removal from G leaves the remaining graph disconnected. A connected graph is said to be separable if its node connectivity is one. All other connected graphs are called nonseparable. In a separable graph a node whose removal disconnects the graph is called an articulation point.

With each minimal cut-set MC_i , i = 1, 2, ..., NMC say, may be associated a binary representation MC_{ij} , j = 1, 2, ..., NE whose components take the value of 1 if the ith cut contains the jth element of the graph and is 0 otherwise.

The minimal cut-set matrix MCM_{ij} is defined as the array of all the binary numbers MC_{ij} where each row i shows the incidence of all jth elements of the network in the ith minimal cut-set of the network. Similarly each column j shows the incidence of the jth element in all cut-sets of the network. The minimal cut-set matrix

for the reliability graph of Fig. 1 between nodes 1 and 4 is

	1	2	3	4	5
MOM	[1	1	0	0	٥
	0	1	1	1	0
	1	0	1	0	l
	0	0	0	l	1

The four cut-sets between nodes 1 and 4 are thus (1,2), (2,3,4), (1,3,5) and (4,5) as indicated by the 1's in the corresponding positions in rows 1 to 4.

This chapter makes the following assumptions:

- 1) the reliability networks are coherent networks.
- each element has two terminals and can be in either of two states, working or failed. There is no standby or switched redundancy.
- 3) the states of all elements are independent i.e., the failure of one element does not affect the probability of failure of any other.

Other assumptions concerning the type of systems, repairable or nonrepairable, are introduced in the next section.

4.3 Modelling of complex systems by reliability graphs

A complex system will be modelled by a weighted linear graph where the weights of the elements are either the probability of success p_j for the non-repairable component or the steady-state probability of being failed p_{jd} for the repairable case defined through the failure and repair rates of the component. For the case of repairable systems it is also assumed that the system has reached a steady state condition of failure and repair and that repaired components are as good as new.

Elements are interconnected at nodes of the network and all the inputs in the system are combined into a single node called the reference node. All other nodes are considered as output nodes. For example, in the representation of a power system, inputs correspond to electrical generators, elements to transmission lines, and nodes to busbars. Electric power flows from the generators via the network of transmission lines to the busbars. In the representation of a digital protection system components of the hardware structure are considered the elements of the graph, the electrical junctions represent the nodes and the signal transducers are the inputs. If data validation and estimation routines similar to those described in chapter II are available, missing data can be replaced or estimated from the redundant measurements. In that case it is assumed that only the simultaneous failure of all inputs could cause a system failure. Otherwise the system has to be analysed as an r-out-of-n configuration.

Unidirectional elements are given a direction from IN to OUT terminals. In graphs with only unidirectional elements the signal from one node will pass to other nodes just through those elements which have their IN terminals connected to that node. For graphs with both types of elements, unidirectional and bidirectional, the signal from one node can pass to the other nodes of the network either through any of the bidirectional elements connected to the node or through the unidirectional elements which have their IN terminals connected to that node. A signal from any of the inputs of the system will reach another node through a path which includes at least one of the undirected elements incident to that node or any directed element

with their OUT terminal incident to that node.

All the elements connected to the reference node are directed elements with their IN terminals joined by the reference node. In separable networks with an articulation point n_j all edges incident to n_j which are disconnected from the inputs when n_j is removed are considered directed elements with their IN terminals connected to n_j .

4.4 Cut-node incidence matrix

The minimal cut-sets of the system are defined here as the set of distinct minimal cuts such that when they occur one or more output nodes of the network are disconnected from the reference node. The minimal cut-sets of node j are a subset of the minimal cut-sets of the network such that when at least one cut occurs the node j is disconnected from the reference node. Each one of these cut-sets is called a minimal cut-node CN_j and the group is the minimal cut-nodeset MCN_j. To illustrate these concepts tables I and II show the minimal cut-sets and the minimal cut-node-sets of Fig. 1.

Note in table II that the minimal cut-set (1,2) causes the disconnection of all the output nodes of the network. Such a minimal cut-set is called the basic minimal cut-set and contains all the elements connected to the reference node. Note also that the minimal cut-set (1,3,4), (2,3,5) and (4,5) cause the disconnection only of the nodes 2,3 and 4 respectively. Each one of these minimal cut-sets is called the proper minimal cut-node set and contains for each node all the elements incident to the node. The order of this cut-set is a cut-set for all the output nodes of the network, all the other minimal cut-sets of the network will cause the disconnection of maximum of n-1 nodes where n is the total number of output nodes.

TABLE I

Minimal cut-sets of system in Fig. 1

Cut	Elements		
1	1,2		
2	1,3,5		
3	1,3,4		
4	2,3,4		
5	2,3,5		
6	4,5		
1			

TABLE II

Minimal cut-node sets of system in Fig. 1

2	3	4
1,2 1,3,5 1,3,4	1,2 2,3,4 2,3,5	1,2 4,5 1,3,5 2,3,4

Based on these concepts, with each minimal cut-set MC_i i = 1,2,..,NMC of the system, may be associated a binary number CNV_{ij} , j = 2,3,...,n called the cut-node incidence vector whose components take the value of 1 if the ith cut-set causes the disconnection of the jth node and is 0 otherwise.

The cut-node incidence matrix CNM_{ij} is defined as the array of all the binary numbers CNV_{ij} where each row i indicates the incidence of the ith minimal cut in all the output nodes of the network, j = 2,3,... because reference node is numbered 1. The cut-node incidence matrix for the reliability graph of Fig. 1 is shown below. Minimal cut-sets

ī	1	1	1,2
1	0	1	1,3,5
1	0	0	1,3,4
0	1	1	2,3,4
0	1	0	2,3,5
0	0	1	4,5
	1 1 0 0 0	1 1 1 0 1 0 0 1 0 1 0 0	1 1 1 1 0 1 1 0 0 0 1 1 0 1 0 0 0 1

The problem of generating the minimal cut-sets for each node in a n output-node network is then the same as the problem of finding the minimal cut-sets between the reference node and each one of the output nodes and defines the incidence of each new minimal cut on all other output nodes. In other words, when we are considering the generation of the cut-sets between the reference node and node k we look also for the incidence or not of each cut-set over all other output nodes in the network. The removal of the components in a minimal cut-set for node k separates the network into exactly two connected subnetworks. One subnetwork includes reference node and j output nodes of the network, the other subnetwork includes node k and n-(j + 1) output nodes. A minimal cut-set for node k is also minimal

cut-set for the n-(j + 1) output nodes of the same connected subnetwork.

The generation of the cut-node incidence relationship simultaneously with the minimal cut-set enumeration provides an efficient and systematic method of generating the minimal cut-node sets for the (n - 1) output nodes simultaneously in a graph of n nodes when the edge reliabilities are known a priori.

4.5 Nonseparable graphs

The contribution of this section is an algorithmic procedure that generates the set of minimal cuts for nonseparable graphs. The advantage of the algorithm compared with others described in the literature (33),(34) is that when the generation process stops, the minimal cut-sets between the reference node and each output node have been generated all at once.

Each element or edge has two terminals identified as SB and EB and the numbering of nodes is arbitrary except that reference node is numbered 1.

4.5.1 Description of the algorithm (43)

The algorithm is a node-scan procedure which generates the minimal cut-sets between all output nodes and the reference node when (NN - 1) nodes are scanned, NN being the total number of nodes in the network. Then the algorithm provides a node-counter NK and a node-flag NF(j) to count and identify the scanned j-nodes.

Each time a minimal cut-set is generated a binary vector MC_{ij} is constructed to indicate the j-elements that constitute the ith cut-set. Similarly a cut-node vector CNV_{ij} is generated to mark whether or not the ith cut-set is a cut-set for each one of the j output nodes. As soon as the procedure scans a new node, the proper cutset of the node is generated with all the edges incident to the new node. The ith proper cut-set is identified by a cut-flag CF(i) equal to 1. All other cut-sets generated when scanning node k are produced by the ring sum of the proper cut-set of node k and each one of the i cut-sets previously generated that have at least one edge in common with the proper cut-set. The number of edges in common is called the index $CI_k(i)$ of the ith cut-set when scanning node k. Each time a ring sum is made the resulting set of edges are checked to see if they are contained in one of the minimal cut-sets generated from the ring sums of the same proper cut-set. If they are contained then this set of edges is not a minimal cut-set are made in an ordered sequence from the highest to the lowest indices.

Each time a cut-set is found the appropriate cut-node vector is generated using the cut-node vector and the cut-flag of the cutset which has been combined with the proper cut-set. If the other cut-set is a proper cut-set or has been generated from a proper cut, the new cut-node vector is equal to the old one but will also include the node being scanned. Otherwise the cut-node vector is equal to the old one but will not include the node being scanned.

The algorithmic procedure can be described in 5 steps: Step 1 : Generate the basic minimal cut MC_{lm} with the set of edges incident to the reference node. Construct a cut-node vector CNV_{lj} , j = 2,3,..NN with all components equal to 1 and mark reference node as scanned, FN(1) = 1. Set a nodecounter NK equal to 1 and a cut-flag CF(1) equal to zero. Calculate lower and upper reliability bounds for each node of the network. Go to step 2.

- Step 2 : Check the node-counter. If it is less than NN 1 choose the unscanned node with the lowest number and go to step 3. If it is equal to NN - 1 the algorithm terminates and all minimal cut-node sets have been generated and bounds for each node reliability are available.
- Step 3 : Let the node chosen to be denoted as node i (i > 1). Find the[k]elements incident to node i. Identify from the minimal cut-sets generated which contain one or more of the k elements and give to each an index $CI_i(p)$ equal to the number of k elements in the p-cut. If at least one of the cut-sets already generated has a non-zero index, go to step 4. If not, advance to next unscanned node in increasing order from node i and repeat step 3.
- Step 4 : Generate the proper minimal cut-set of node i with the k elements found in step 3 and a cut-node vector with all elements equal to 0 except for the element corresponding to node i which is equal to 1. Mark the cut as a proper cut by means of a cut-flag equal to 1. From the new minimal cut-set calculate new lower and upper reliability bounds for node i. Go to step 5.
- Step 5 : Make the ring sum between the proper cut-set of step 4 and each one of the already generated cuts with a non-zero index identified in step 3. The new set of edges is a minimal cut-set if and only if it is not contained in any of the cuts already generated when scanning node i. This procedure is applied for all cuts with non-zero index, beginning with the cuts of highest indices and continuing with cuts of decreasing index. Construct for each minimal

cut a cut-node vector equal to that of the cut-set combined with the proper cut-set except for the term of the node being scanned which must be equal to 0 or 1 if the old cut-flag is 0 or 1 respectively. Set the new cut-flag equal to the old cut-flag. Each time a minimal cut and its cut-node vector are generated, calculate lower and upper reliability bounds of the affected nodes. Mark node i as scanned FN(i) = 1, increment node-counter and go to step 2.

To illustrate this procedure, a very simple example (Fig. 1) is used to determine the minimal cut-sets for output nodes 2, 3 and 4. Reference node is numbered 1. In the example we reserve m as subindex of the 5 elements (m = 1,2,3,4,5) and j as subindex of the three output. nodes (j = 2,3,4).

Step	1	:	MClm	12	11000	(first minimal	cut-set)
			CNV _{1j}	8	111 .		
			FN(1)	=	1		
			CF(1)	a	0		
			NK	8	1		
Step	2	:	NK =	1	go to step 3		
			·				
Step	3	:	i	9	2		
			[k]	۵	10110		
		C	1 ₂ (1)	-	1		
Step	4	: M	^C 2m	. ==	10110	(second minimal	cut-set)
		C.	NV 2j	a	100		
		C.	F(2)	13	1		

Step 5	:	MC 3m	-	01	110		(t	hird	minimal	cut-set)
		CNV 3.j	-	01	1					•
		CF(3)	•	0						
		FN(2)	8	1						
		NK	8	2						
Step 2	:	NK	=	2	go to	step	3			
Step 3	:	i	=	3						
		[k]	-	01	101					
		CI ₃ (1)	-	1						
		CI ₃ (2)	3	1						
		CI ₃ (3)	=	2						
		,								
Step 4	:	MC 4m	=	01	101			(four	th mini	mal cut-set)
		CNV _{4j}	-	01	0					
		CF(4)	a	1	·					
Step 5	:	MC _{5m}	=	0 0	011		1	(fift	h minim	al cut-set)
		CNV ₅₁	=	0 0	1					
		CF(5)	-	0						
		MC 6m	-	10	101		((sixt	h minim	al cut-set)
		CNV ₆₁	=	10	1					
		CF(6)		0						
		MC _{7m}	12	11	011	ε		MC5m	1	
		FN(3)	æ	1				<u> </u>	-	
		NK	-	3						
		•								
Step 2	:	NK	83	3	**	NN -	1	st	op.	

Termination of the procedure

•

A detailed discussion and proof of the algorithm steps is included in section A2.2 of Appendix 2.

4.5.2 Programming considerations

The algorithm is designed to generate the entire minimal cut-sets and cut-node incidence matrices and maintain them in the core memory of the computer. In the computer program implementation core memory space must be set aside to keep all the information concerning both matrices.

A very simple and efficient storage scheme is possible taking advantage of the binary nature of both matrices. The minimal cut-set matrix will be kept as a column vector whose components indicate the number of the columns with value equal to 1 for each cut-set i.e., the number of the edges of the graph that contribute to that cut-set. A second vector will be used to keep the information about the location of the cut-set information within the first vector; its dimension will be (1 x NMC), where NMC is the number of minimal cut-sets. Example: Let A be the minimal cut-set matrix of the cut-sets between nodes 1 and 8 of the graph in Fig. 3. The number of minimal cutsets is 10.

 1
 1
 1
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0









Using the storage scheme as described above the information contained in A is kept in a column vector of dimension (37×1) as follows:

$$B^{T} = (1,2,3/3,4,5/1,2,6/4,5,6/7,8,10,14/9,10,14/$$

11,12,13,14/7,8,10,15,16/9,10,15,16/11,12,13,15,16)

where the cut-sets are shown separated by slashes and the elements in each cut-set separated by commas for clarity. A second column vector C will keep the information of the start point of each cut-set within vector B as follows:

$$C^{T} = (1,4,7,10,13,17,20,24,29,33)$$

This example shows how matrix A of dimension (10 x 16) is stored in two vectors $B(37 \times 1)$ and $C(10 \times 1)$ which represents a 70.6% reduction of space requirement.

A similar scheme is used to keep the cut-node incidence matrix but the equivalent vector will identify the nodes affected by each cut-set. Table III shows the percentage space-saving for storage of the minimal cut-sets and cut-node incidence matrices of 4 graphs described in Appendix 1. Graphs 5 and 6 are complete graphs with an edge between every pair of nodes. There are $(n - 1) \cdot 2^{n-2}$ minimal cutsets, between the reference node (node 1) and all other nodes, for these graphs. Graphs 1 and 4 correspond to Figs. 2 and 3 taken from references [32] and [30] respectively.

TABLE III

Graph	Cut-set matrix dimension	Column vectors	Cut-node matrix dimension	Column vectors	% Space saving
1	28 x 10	1 x 114	28 x 7	l x 112	52.5
4	22 x 16	1 x 164	22 x 7	1 x 89	50.0
5	63 x 21	1 x 735	63 x 6	1 x 255	41.8
6	127 x 28	l x 1874	127 x 7	1 x 575	44•9

CUT-SETS SPACE SAVING

With this simple modification the algorithm is limited practically only by the computer time available, while without this change the procedure is limited by the core size of the computer.

4.5.3 Tests

The algorithm for the minimal cut-set generation of nonseparable graphs described in section (4.5.1) has been programmed for the CDC 6400 computer using Fortran IV.

Table IV shows the cut-node generation time for several randomly generated reliability graphs described in Appendix 1. The graph configuration index presented for each graph is defined by the quotient between number of nodes and number of elements. The times presented do not include the time required to calculate all node reliability bounds each time a new cut is generated. The time required by the program cannot be measured entirely in terms of the number of nodes but a rough indicator is the graph configuration index. Graphs with many edges in series (high index) will be solved more quickly than those with more grid-like configuration (low index).

Both sparse and non-sparse approaches were implemented and a comparison of the computation times is presented in table V. The

TABLE IV

1

COMPUTATION TIME FOR CUT-NODE GENERATION

USING SPACE SAVING MODIFICATION

Graph	Nodes	Elements	Configuration index	Number of cut-node	Total time (secs.)	Average cut-node time (msec.)
1	8	10	0.80	84	0.395	4.70
2	20	33	0.61	1514	1.65	1.09
3	15	25	0.60	875	1.12	1.28
4	8	16	0.50	61	0.409	6.70
5	7	21	0.33	192	0.672	3.50
6	8	28	0.29	448	1.78	3•97
7	9	36	0.25	1024	7.27	7.10
8	28	47	0.60	4514	4.88	1.08
9	14	21	0.67	856	1.06	1.24
					l	

TABLE V

Graph	Sparse Average cut-node time (msec)	Non-sparse Average cut-node time (msec)
1	4.70	5.4
2	1.09	5.2
3	1.28	4.7
4	6.70	7.8
5	3.50	6.5
6	3.97	8.8
7	7.10	20.5
	f i i i i i i i i i i i i i i i i i i i	

COMPARISON OF COMPUTATION TABLE

TABLE VI

COMPUTATION TIME FOR DIFFERENT ELEMENT

REDUNDANCY IN FIGURE 3

Number of minimal cut-node sets = 61

Case	Elements	Configuration index	Average cut-node time (msec.)
1	16	0.50	7.8
2	26	0.31	8.7
3	36	0.22	9•4
4	46	0.17	9•9

.

space saving modification increases the efficiency of the algorithm reducing both the memory and computation-time requirements of the computer program implementation of the algorithm.

Computation time is almost independent of the redundancy between each pair of connected nodes in the graph, it increases the space requirement of the program but the number of cut-node sets is the same. This is an advantage of the cut-set approach compared with methods based on success paths which need to reduce parallel edges to an equivalent edge before defining the success paths in order to decrease running time. Table VI shows the cut-node generation time for the graph of Fig. 3 for different redundancy of the element. Case 1 corresponds to Fig. 3 and cases 2, 3 and 4 represent an increase of 1, 2 or 3 elements between each pair of connected nodes.

The input for the algorithm is the graph represented by its edge listing which identify each edge as a pair of unsigned nodes. Figures 4 to 6 list the minimal cut-sets and the cut-node incidence of Fig. 2 as they are output by the computer program and are examples of the node numbering independence of the algorithm.

4.6 Separable graphs

Deo⁽⁴²⁾ has proved that the node connectivity of any graph G can never exceed the edge connectivity of G which at the same time cannot exceed the degree of the node with the smallest degree in G. Then any graph with one or more pendant nodes (degree one) is always a separable graph but the converse is not also true, i.e., not all separable graphs have pendant nodes. This is illustrated by Figs. 7 and 8 taken from references [42] and [44] respectively. Node 6 in both figures is an articulation point but only Fig. 8 has a pendant node identified as node 7.

Ε	SB	E
1	1	3
2	3	4
3	4	7
4	7	8
5	1	2
6	2	5
7	5	6
8	6	8
9	3	5
10	5	7

Edges in the cut	Cut-node incidence
1,5	2,3,4,5,6,7,8
5,6	2
1,6	3,4,5,6,7,8
1,2,9	3
2,5,9	2,4,5,6,7,8
2,6,9	4,5,6,7,8
2,3	4
1,3,9	4,5
3,5,9	2,5,6,7,8
3,6,9	5,6,7,8
6,7,9,10	5
2,7,10	4,6,7,8
3,7,10	6,7,8
5,7,9,10	2,5
1,7,9,10	3,4,6,7,8
7,8	6
6,8,9,10	5,6
2,8,10	4,7,8
3,8,10	7,8
5,8,9,10	2,5,6
1,8,9,10	3, 4, 7, 8
3,4,10	7
4,7	6,8
4,8	8
2,4,10	4,7
1,4,9,10	3,4,7
4,5,9,10	2,5,6,8
4,6,9,10	5,6,8

Figure 4. Minimal cut-sets and cut-node incidence for Fig. 2 First numbering of nodes

Edge listing

E	SB	E
123456789	1 2 3 4 1 6 7 8 2	2 3 4 56 7 8 5 7
10	7	4

Cut	Edges in the cut	Cut-node incidence
1	1,5	2,3,4,5,6,7,8
2	1,2,9	2
3	2,5,9	3,4,5,6,7,8
4	2,3	3
5	1,3,9	2,3
6	3,5,9	4,5,6,7,8
7	3,4,10	4
8	2,4,10	4,5
9	1,4,9,10	2, 3, 4
10	4,5,9,10	5,6,7,8
11	4,8	5
12	3,8,10	5,6
13	2,8,10	3,4,5
14	1,8,9,10	2,3,4,5
15	5,8,9,10	6,7,8
16	5,6	6
17	1,6	2,3,4,5,7,8
18	2,6,9	3, 4, 5, 7, 8
19	3,6,9	4,5,7,8
20	4,6,9,10	5,7,8
21	6,8,9,10	7,8
22	6,7,9,10	7
23	4,7	5,8
24	7,8	8
25	2,7,10	3,4,5,8
26	3,7,10	4,5,8
27	5,7,9,10	6,7
28	1,7,9,10	2,3,4,5,8

Figure 5. Minimal cut-sets and cut-node incidence for Fig. 2 Second numbering of nodes

.

Edge listing

E	SB	E
1 2 3 4 5 6 7 8	1 7 8 6 1 2 5 3	78642534
9	7	5
10	-5	6

Cut

Edges in the cut	Cut-node incidence
1,5 5,6	2,3,4,5,6,7,8 2
1,6	3,4,5,6,7,8
6,7,9,10	5
1 7 9 10	2,7
7.8	3
6,8,9,10	3,5
5,8,9,10	2,3,5
1,8,9,10	4,6,7,8
4,8	4
4,7	う94 スム 5
4,5,9,10	2.3.4.5
1,4,9,10	6,7,8
3,4,10	6
3,6,9	3, 4, 5, 6
<i>3</i> , <i>5</i> , <i>9</i>	2,3,4,5,6
3.8.10	4,0
3,7,10	3.4.6
1,2,9	7
2,7,10	3, 4, 6, 8
2,8,10	4,6,8
2,4,10	6,8
212 2.5.9	0 2 3 4 5 6 8
2,6,9	3, 4, 5, 6, 8
	~ 1 . 1 ~ 1 ~ 1 ~

Figure 6. Minimal cut-sets and cut-node incidence for Fig. 2 Third numbering of nodes If the algorithm of section (4.5) is applied to the graphs of Figs. 7 and 8 the proper cut-set of node 6 will be defined by the set of edges (1,6) (2,6) (3,6) (6,8) (6,4) (6,5) (6,7) and (4,6) (8,6) (6,7)respectively, where each edge has been represented by its terminal nodes. However this cut-set is not a minimal cut-set between nodes 1 and 6 because the subsets (1,6) (2,6) (3,6) and (4,6) (8,6) break all paths between these pairs of nodes respectively. Then the algorithm of section 4.5 cannot be applied directly for the case of separable graphs.

A slight modification to the algorithm of section 4.5 based on the concept of directed edges allow to present in this section an algorithmic procedure for the generation of the minimal cut-sets between the reference node and all other nodes of separable graphs. It can be applied in the reliability analysis of transmission and distribution systems of electric power and is the most efficient cut-set enumeration method known to the author for such application.

4.6.1 Definitions and basic considerations

An undirected edge is a two-way branch represented by an unordered pair of nodes (n, n_j) and represented by a line segment between n_i and n_j . An undirected edge is said to be incident on nodes n_i and n_j .

A directed edge is a one-way branch represented by an ordered pair of nodes (n_i, n_j) and represented by a line segment between n_i and n_j with an arrow directed from n_i to n_j . This orientation indicates that a signal passing through the link (n_i, n_j) must be in at node n_i and out at node n_j . A directed edge is said to be incident out of the initial node n_i and incident into the terminal node n_i .

To generate the minimal cut-sets between node 1 and all other nodes of systems with configurations similar to the ones shown in Figs. 7 and 8,



Figure 7. Separable graph without pendant nodes [42]





both types of edges are used to represent the components. From Fig. 8 it can be seen that every node adjacent to a pendant node is an articulation point. However the algorithm proposed in this section is not designed to test the separability of a graph therefore all the articulation points are assumed to be known. The algorithm has been designed considering that all subsystems separated from the reference node by an articulation point n_k have tree configuration and can be represented by subgraphs with directed edges whose initial nodes are the nearest to nk. This limitation was imposed because separable graphs with two or more nonseparable subgraphs of mesh structure are represented by undirected edges and can be efficiently analysed by successive applications of the algorithm of section (4.5) for each nonseparable In such an application it should be noted that the set of subgraph. minimal cuts between the reference node and any articulation point are also cut-sets for all nodes of the nonseparable subgraph connected to that articulation point. When enumerating the set of minimal cuts of the subgraph the articulation point represents the reference node and all edges incident out of that node are the inputs of the subsystem.

4.6.2 Description of the algorithm

The 5-step algorithm for nonseparable graphs is modified to solve the minimal cut-set enumeration of separable graphs by considering the orientation of the edges in each subgraph defined by the articulation points.

With this modification the proper cut-set generated when scanning each node is formed with the undirected edges incident on the node and the directed edges incident into that node. For example the proper cut-set of node 6 in Figs. 7 and 8 will be generated as the set of edges

(1,6) (2,6) (3,6) and (4,6) (8,6) respectively. In both figures 7 and 8 the set of edges (6,8) (6,4) (6,5) (6.7) and (6,7) respectively will be oriented incident out of node 6. All other cut-sets are generated in a similar way of section (4.5) by the ring sums of the proper cut-set of the node being scanned and all cut-sets from previous nodes which have at least one edge in common with this proper cut-set.

When scanning an articulation point the cut-node vector of the proper cut-set will indicate the incidence of such cut also in all nodes dependent of the articulation point. A node is defined to be dependent of an articulation point if all paths from the reference node to that node pass through the articulation point. It should be noted that all subgraphs dependent of an articulation point n have been assumed to be separable networks with tree configuration. A nonpendant node of this tree will be called an internal tree node. For all other cut-sets generated when scanning an articulation point the cut-node vector is generated as in the separable case but taking into account that a cut-set of the articulation point is also a cut-set for each node of the tree connected to that point. Similarly a cut-set which does not affect the articulation point it neither will affect any tree node connected to that point. As mentioned before the procedure assumes as known all the articulation points in the graph and also the internal tree nodes connected to a particular articulation point. The total number of nodes of the graph is designed by NN.

The algorithmic procedure can be described by 5 steps :

Step 1 : Generate the basic minimal cut-set MC_{lm} with the set of edges incident to the reference node. Construct a cut-node vector CNV_{lj}, j = 2,3,...,NN with all components equal to 1 and mark reference node as scanned FN(1) = 1. Set a node counter

NK equal to 1 and a cut-flag CF(1) equal to zero. Calculate lower and upper reliability bounds for each node of the network.

- Step 2 : Check the node-counter. If it is less than NN choose the unscanned node with the lowest number and go to step 3. If it is equal to NN the algorithm terminates and all minimal cut-node sets have been generated and bounds for each node reliability are available.
- Step 3 : Let the node chosen to be denoted as node i (i>1). Find the k elements incident on (undirected edges) and/or incident into (directed edges) node i. If node i is a tree-node (internal or pendant) go to step 4. If not, identify from the minimal cut-sets generated which ones contain one or more of the k elements and give to each one an index $CI_i(p)$ equal to the number of k elements in the pth cut-set. If at least one of the cut-sets already generated has a non-zero index, go to step 4. If not, advance to next unscanned node in increasing order from node i and repeat step 3.
- Step 4 : Generate the proper minimal cut-set of node i with the k elements found in step 3. Compare this cut with all cuts already generated, if the node is not a tree node, and discard the other cut if it is identical to the proper cut-set. The discarded cut will not be used any more in the process. Construct a cutnode vector as follows: If node i is an articulation point or a tree-node the cut-node vector is equal to node i and other nodes dependent of node i. For all other types of nodes the cut-node vector is just equal to node i. Mark the cut as a proper cut-set setting a cut-flag equal to 1. If node i is a

tree-node or a pendant node, mark node i as scanned, increment node-counter and go to step 2. If not, go to step 5.

Step 5 : Make the ring sum between the proper cut-set of step 4 and each one of the cut-sets with a non-zero index identified in step 3. The new set of edges is a minimal cut-set if and only if it is not contained in any of the complete set of cuts previously generated. This procedure is applied for all cuts with nonzero index, beginning with the cuts of highest indices and continuing with cuts of decreasing index. Construct for each minimal cut-set the cut-node vector equal to the one of the cut-set combined with the proper cut-set except for the terms of node i and any node dependent of node i which must be equal to 0 or 1 if the flag of the cut combined with the proper cutset is 0 or 1, respectively. Set the new cut-flag equal to the flag of the cut combined with the proper cut-set, mark node i as scanned, increment node-counter and go to step 2.

To illustrate the procedure the block reliability diagram shown in Fig. 9 is used and all cut-sets between node 1 and all other nodes of the diagram are determined. As defined in this section nodes 4 and 7 are articulation points, nodes 5 and 3 are tree-nodes and nodes 6, 8 and 10 are pendant nodes. It can be seen that each tree-node is also an articulation point. Table VII presents the results as they are produced when applying the algorithm step-by-step to Fig. 9. A detailed discussion and proof of the algorithm steps is included in section A2.3 of Appendix 2.

TABLE VII

CUT-SET GENERATION FOR FIGURE 9

		Elements in	Cut-node	Cut	Cut index			x	Cut
Node	Cut	the cut	incidence ilag		2 3		4	7	
1	1	1,2	2,3,4,5,6,7,8,9,10 0 1 1					-	
	2	1,3,4	2	1			1	1	-
2	3	2,3,4	3,4,5,6,7,8,9,10	0		1	1	1	1
	4	2,5	3	1				1	-
3	5	1,5	2,4,5,6,7,8,9,10	0				1	1
	6	3,4,5	4,5,6,7,8,9,10	0			1	2	3
	7	3,6	4	1				1	-
	8	1,4,6	2,4,5,6,10	1				2	2
4	9	2,4,6	3,7,8,9	0	Ī			2	3
	10	4,5,6	7,8,9	0	1			3	6
5	11	9	5,6	1					
6	12	8	6	1					-
	13	4,5,6	7	1	discard cut 10				
	14	3,6			discarded by cut 7				cut 7
	15	1,5			discarded by cut 5				cut 5
	16	2,5			discarded by cut 4				cut 4
7	17	1,3,5,6			disjoint union of				n of
	18	2,3,5,6	· · · · · · · · · · · · · · · · · · ·		disjoint union of				n of
		-12121			cuts 4 and 7				
	19	2,4,6			a1:	scar	aea		cut 9
	20	1,4,6			discarded by cut 8 discarded by cut 6			cut 8	
	21	3,4,5						cut 6	
8	22	11	8 .	1					-
9	23	7	8,9	1					-
10	24	10	10	1					-

4.6.3 Tests

The computer program implementation of the algorithm uses the same storage scheme of section (4.5.2) for the information concerning the minimal cut-set and cut-node incidence matrices. The algorithm has been programmed for the CDC 6400 computer using Fortran IV.

Table VIII shows the cut-node generation time for three graphs described in Appendix 1. Graph 3 corresponds to the 1973 West-Venezuelan 115 kV electric power system. The time presented do not include the time required for calculation of reliability bounds for each node and are comparable to those obtained for nonseparable graphs. Graphs with tree configuration have high index configuration as shown in table VIII.

The inputs of the algorithm are the edge listing and the articulation points and pendant nodes of the graph. Each undirected edge is represented as a pair of unsigned nodes and each directed edge by a pair of signed nodes. The initial vertex of a directed edge is expressed by a negative number and the terminal node as a positive one. Each articulation point is defined with all the nodes of the tree connected to that point.

4.7 Acyclic directed graphs

The nonseparable graphs considered in this chapter so far have been undirected graphs. In this section we consider again the case of nonseparable graphs but now all edges are one-way branches. The concept of directed edges of section (4.6.1) is used to represent the edges of a directed graph. Then a directed graph G consists of a set of elements called nodes and a set of ordered pairs of nodes called directed edges. A directed graph that has no directed circuits is called acyclic. All directed graphs considered in this section are assumed to be acyclic



Figure 9. Reliability block diagram No. 3

TABLE VIII

COMPUTATION TIME FOR CUT-NODE GENERATION IN SEPARABLE GRAPHS

Graph	Nodes	Elements	Configuration index	Number of cut-node	Total time (sec.)	Average cut-node time (msec.)
io	10	12	0.83	57	0.557	9•77
11	11	. 14	0.79	106	0.650	6.13
12	15	18	0.83	449	1.660	3.70
and will be called simply directed graphs or digraphs for short.

Many systems can be modelled by directed graphs. In particular digital systems with one-way components are represented by directed graphs. In digital protection systems, analogue data (current and voltages) are sampled from the transducers in the substation through sample-hold circuits. These values are converted to digital form by A/D converters before they are fed into the processor responsible for the tripping criteria. Directed graphs can be employed for the representation of such systems when analysing the terminal reliability between the input and output of the system. Acyclic digraphs have been also extensively used for the representation of the activity networks⁽⁴⁵⁾ defined in the Critical Path Method (CPM) or in the Program Evaluation and Review Technique (PERT).

This section presents an algorithm for the minimal cut-set enumeration between the source node (s) and the sink node (t) of acyclic digraphs.

4.7.1 Basic definitions

A directed path from node s to t is an alternating sequence of nodes and edges beginning with s and ending with t such that all nodes and edges are distinct. Each edge in the sequence is oriented from the node preceding it to the node following it. We refer to such path as s-t directed path or an s-t path for brevity. Node s have only edges incident out of it and node t have only edges incident into it. Then any s-t directed path will always include one edge incident out of s and one incident into t.

A set of edges in a directed graph G is an s-t directed minimal cut-set if its removal from G breaks all directed paths from s to t, and no proper subset holds the same property. Since all graphs in this section

are directed graphs we shall refer to directed paths and directed minimal cut-sets simply as paths and cut-sets. As mentioned before a digraph G is connected if its corresponding undirected graph is connected. The same concept applies for any directed subgraph of G.

4.7.2 Description of the algorithm (43)

The 4-step algorithm presented in this section is concerned with the enumeration of the minimal cut-sets between the origin node s (or reference node) and the terminal node t (output of the system) of acyclic digraphs.

The algorithm is a node-oriented procedure which, starting from the origin, continues the search with the nearest nodes to the origin and ends when (NN - 1) nodes have been scanned.

Each time the algorithm scans a node the set of edges incident into and out of that node are identified. Since the origin has only edges out of it, the first cut-set produced by the process is that set of edges and the algorithm jumps to one of the nearest unscanned nodes. Each time a new node w is scanned, an index is created for each one of the cut-sets already generated. This index is equal to the number of edges incident into w that are contained in each cut-set. Each cut-set with an index not equal to zero will form a possible cut-set replacing the edges leading into w that are contained in the cut by the edges going out of w. A comparison is made with the previous cut-sets generated when scanning w to see if the new cut is minimal or not. This input-output substitution and the comparisons between cut-sets is used iteratively by the algorithm to enumerate all s-t cut-sets.

The algorithmic procedure can be described in 4 steps :

- Step 1 : Generate the basic minimal cut-set MC_{lm} with the set of edges incident out of the reference node. Mark reference node as scanned FN(1) = 1 and set a node-counter NK equal to 1. Calculate lower and upper bounds on system reliability. Go to step 2.
- Step 2 : Check the node-counter. If it is less than NN 1 (NN number of nodes) choose the unscanned node with the lowest number and go to step 3. If it is equal to NN 1 the algorithm terminates and all minimal cut-sets have been generated and bounds on system reliability have been calculated.
- Step 3 : Let the node chosen to be denoted as node i (i > 1). Find the[k]and [p] elements which are incident into or going out of node i respectively. Identify from the minimal cut-sets generated which contain one or more of the k elements and give to each an index $CI_{i(j)}$ equal to the number of k elements in the jth cut. If at least one of the cuts already generated has index equal to k go to step 4. If not, advance to next unscanned node in increasing order from node i and repeat step 3.
- Step 4 : A minimal cut with index x, x = 1,2...,k generates a new cut-set substituting its x element for the p elements of node i, where x are the k elements of node i contained in that cut. The new cut is a minimal cut if and only if it is not contained in anyone of the cuts already generated for the node i. This procedure is applied for all the cuts

with index, beginning with the cuts of highest indices and continuing with cuts of decreasing index. For each new minimal cut-set, calculate new lower and upper bounds for the system reliability. Mark node i as scanned, increment nodes-counter and go to step 2.

To illustrate the procedure we will use the reliability graph of Fig. 3 used by Nelson et al(30) for illustration of their unidirectional algorithm.

Step 1	:	MClm		1,2,3	(first cut-set	;)
		FN(1)	8	1		
		NK	72	1		

Step 2 : NK = 1 (< 7)

Choose node 2, go to step 3

Step 2 : NK = 2 (< 7)

Choose node 3, go to step 3

Step 3 :

1	8)
[k]	=	4,5,6
[P]	=	7,8,10,14
CI ₃ (1)		0
CI ₃ (2)	s .	2
Discar	d no	de 3 and choose node 4

Step 3:
$$i = 4$$

 $[k] = 7,8$
 $[p] = 9$
 $CI_4(1) = CI_4(2) = 0$
Discard node 4 and choose node

Step 4 : i = 5
[k] = 3
[p] = 6

$$CI_5(1) = CI_5(2) = 1$$

Step 4 : $MC_{3m} = 1,2,6$ (third cut-set)
 $MC_{4m} = 4,5,6$ (fourth cut-set)
FN(5) = 1
NK = 3

5

Step 2 : NK = 3 (< 7)

i		3
[k]	8	4,5,6
[ŋ]		7,8,10,14
ci ₅ (1)	=	0
CI ₅ (2)	s .	2
CI ₅ (3)	13	1
CI ₅ (4)	-	3

Step 4	:	MC _{5m}	=	7,8,10,14	(fifth cut-set)
		MC 6m	8	3,7,8,10,14	non minimal
		MC _{7m}	æ	1,2,7,8,10,14	non minimal
		FN(3)	8	1	
		NK	8	4	

Choose node 4, go to step 3

Step 3: i = 4

$$\begin{bmatrix} k \end{bmatrix} = 7,8$$

$$\begin{bmatrix} p \end{bmatrix} = 9$$

$$CI_{4}(1) = CI_{4}(2) = CI_{4}(3) = CI_{4}(4) = 0$$

$$CI_{4}(5) = 2$$

Step 4 : MC_{6m} = 9,10,14 (sixth cut-set) FN(4) = 1 NK = 5

NK = 5 (< 7) Step 2 : Choose node 6, go to step 3 Step 3 : i = 6 [k] = 9,10 [p] = 11,12,13 $CI_6(1) = CI_6(2) = CI_6(3) = CI_6(4) = CI_6(5) = 0$ $CI_6(6) = 2$ Step 4 : MC_{7m} = 11,12,13,14 (seventh cut-set) FN(6) = 16 NK NK = 6 (< 7)Step 2 : Choose node 7, go to step 3 Step 3 : i = 7 [k] = 14 [P] = 15,16 $CI_7(1) = CI_7(2) = CI_7(3) = CI_7(4) = 0$ $CI_7(5) = CI_7(6) = CI_7(7) =$ 1 7,8,10,15,16 (eighth cut-set) MC_{8m} Step 4 : MC_{9m} **- 9,10,15,1**6 (ninth cut-set) = 11,12,13,15,16 (tenth cut-set) MC_{10m} FN(7) =1 NK 7 Step 2 NK 7 stop.

A detailed discussion and proof of the algorithm is included in section A2.4 of Appendix 2.

4.7.3 Tests

The minimal cut-set generation algorithm for acyclic directed graphs have been programmed for the CDC 6400 computer in Fortran IV. The space saving modification presented in section (4.5.2) is not incorporated in this algorithm in order that a more realistic comparison with the approach presented by Nelson et al(30),(31) can be made.

Figure 10 shows the listing of the minimal cut-sets and system reliability bounds for the system in Fig. 3 as they are output by the computer program. Figure 3 corresponds to the system example used by Nelson et al(30). The algorithm presented here listed the 10 minimal cut-sets between IN and OUT terminals and calculated the reliability bounds using the inclusion-exclusion method (see chapter 3, section 3.9) with all cut-sets in 1.45 seconds. The program uses only 0.455 seconds to list the 10 cut-sets without calculating reliability bounds. A probability of success P is given to each element of Fig. 3 as shown in Fig. 10.

Table IX presents the reliability bounds and computing times using the program reported by $Batts^{(31)}$ for 3 different cases. When all cut-sets are generated and used to calculate the bounds the computing time is 11.22 seconds. Even in the case of generating only the cutsets up to third order the computing time greatly exceeds the computing time of the present algorithm in generating all the cut-sets.

Q
30 0.20 30 0.20 90 0.10 35 0.15 75 0.25 87 0.13 82 0.18 89 0.11
35 0.12 35 0.15 35 0.15 35 0.15 75 0.25 70 0.30 70 0.30

INPUT DATA

MINIMAL CUT MATRIX

CUT	ELEMENTS
1	111000000000000000000000000000000000000
2	001110000000000000000000000000000000000
3	1100010000000000
4	0001110000000000
5	000001101000100
6	000000011000100
7	000000000111100
8	0000001101000011
9	000000011000011
10	0000000000111011

SYSTEM RELIABILITY BOUNDS

CUT	LOWER	UPPER
l	•99600	•99600
2	.99240	.99240
3	•98772	•98772
4	•98 351	•98351
5	•98255	.98255
6	•97941	•97941
7	•97860	•97860
8	•97834	•97834
9	•97749	•97749
10	•97727	•9772 7

Figure 10. Minimal cut-sets and system reliability bounds (6th order approx.)for system in Fig. 3

TABLE IX

COMPUTATION	TIME	USING	NELSON	\mathbf{ET}	AL	APPROACH	 ///////////////////////////////////////

Cut-sets generated	System reliability	Computation time (secs.)	
A11	0.97726	11.22	
Up to 4th order	0.97726	9.40	
Up to 3rd order	0.98006	2.48	

The previous results show that the present algorithm is simpler than the Nelson et al⁽³⁰⁾ approach which needs to find all the 55 success paths between nodes 1 and 8 of Fig. 3 before starting to generate the minimal cut-sets. Afterwards they need to perform all k logic sums between the 16 column vectors of that path matrix to generate a k-order cut. Besides, when they generate a cut it needs to be compared with all cuts of lower order to establish if it is minimal or not.

Table X shows the computation times for the minimal cut-set enumeration of the same network of Fig. 3 but varying the redundancy of elements between each pair of connected nodes. Case 1 corresponds to Fig. 3 and cases 2, 3 and 4 represent an increase of 1, 2 or 3 elements between each pair of connected nodes. This is another advantage of the new algorithm compared with Nelson et al⁽³⁰⁾ because greater redundancy increases the number of success paths. Times in table X do not include the time to calculate the system reliability bounds. 154

(30).(31)

TABLE X

Case	Elements Configuration index		Computing time (sec.)		
1	16	0.50	0.455		
2	26	0.31	0.466		
3	36	0.22	0.539		
4	46	0.17	0.595		

COMPUTATION TIMES FOR DIFFERENT ELEMENT REDUNDANCIES IN FIG. 3 USING THE ALGORITHM OF SECTION 4.7

4.8 Node failure problem

In many practical applications, the nodes of a graph may be also unreliable elements. For example, in an electric power system, the nodes represent busbars and the branches represent transmission lines. In a power system the failure of a branch (transmission line) occurs when the branch suddenly becomes incapable of carrying power and the failure of a node (busbar) occurs when electric power cannot flow through that electrical junction.

In general a node is outaged if the node itself has failed e.g., a busbar fault, or if no path exists from the node to any of the inputs of the system through branches and nodes in the up state. Components (branches or nodes) which have failed are said to be down. After a component has been repaired and is operating normally, it is said to be up. Up and down are referred to as the states of the components. The distinction between a node failure, which would usually represent a busbar fault, and a node outage, which is caused by the failure of one or more branch(es) and/or node(s), should be noticed. Whether or not a node is outaged is, in general, a function of the states of all branches and nodes in the system.

A sample network of an electric power system is shown in Fig. 11. In this system there are three sources, one at each of busbars 1, 7 and 8. Figure 12 presents the reliability block diagram of the same system where the inputs have been combined into a single node called the reference node. The reference can never fail and is not included when considering node failures.

The cut-set approach by Billinton and Grover⁽⁷⁹⁾ assumes that generation busbars are 100% reliable. The influence of busbar faults on node outage rates is also neglected in [79] by considering them as 100% reliable when they are in the paths for other busbars. The equations provided in [79] can be used to determine the outage rates and durations associated with first, second and third order cut-set but they are considered only for single, double or triple line outages. It was considered that cut-sets higher than third order contribute very little to the load point reliability indices and were not included in the reliability evaluation.

Considering the third-order cut-set as the highest contingency to be evaluated, there are several combinations which can produce a nodal outage :



Figure 11. Distribution system example [79]



Figure 12. Reliability block diagram of Fig. 11

	Event	Cut-set order
l	One line outage	First
2	One node outage	First
3	Two line outages	Second
4	Node and line outages	Second
5	Two node outages	Second
6	Three line outages	Third
7	Node and two line outages	Third
8	Two nodes and one line outage	Third
9	Three node outages	Third

In order to evaluate the influence of busbar faults on the reliability indices of each node, an evaluation was done of the values of the previous nine events using the equations of [79] and considering that :

- (i) all the lines have the same failure rates (λ_{line}) and outage duration
- (ii) all the nodes have the same failure rates (λ_{node}) and outage duration with values between 1% to 100% of the line outage values.

(iii) line parameters are as follows :

normal weather - permanent outage rate:	0.6	f/y
- temporary outage rate:	1.0	f/y
adverse weather- permanent outage rate:	30.0	f/y
- temporary outage rate:	10.0	f/y
outage duration:	3.0	h
maintenance - outage rate:	2.0	m/y
- duration:	6.0	h

Event	Lines	Nodes
1	100	000
2	000	100
3	1,10	000
4 ·	100	100
5	000	110
6	111	000
7	110	100
8	100	110
9	000	111

The minimal cut-sets used for such evaluation is

Table XI presents the permanent outage rates for these 9 events. It is clear from this table that a first order cut set produced by a node failure has a greater contribution to the node reliability index than a second order cut-set due to a double line outage even at the minimum node failure rate considered, i.e., 0.06 f/y. A second order cut-set produced by simultaneous failures of one line and one node also makes a greater contribution than the third order cut-set of a triple line contingency which is also lower than a double node failure for a node failure rate equal to 15% of the line failure rate i.e., one failure every 10 years. The third order cut-set caused by a double line contingency and one node failure is also comparable to the cutset of the same order constituted by a triple line contingency. The third order cut-sets produced by three node failures or two nodes and one line failures can be neglected due to the small values even for high node failure rate. Table XII presents the temporary outage rates for the same 9 events. Equivalent comparisons can be done in this case as for table XI.

Tables XI and XII show that if meaningful reliability indices are to be obtained in a cut-set approach to failure modes and effects analysis in electric power systems, then node failures must be considered in such analyses.

TABLE XI

PERMANENT OUTAGE RATES FOR FIRST, SECOND AND THIRD ORDER CUT-SETS

λ	FIRST-C	RDERCUT	SECO	ID-ORDER	CUT	THI	THIRD-ORDER CUT				
$\frac{\text{hode}}{\lambda}$ line	l Line Outage	l Node Outage	2 Lines Outage	l Line and l Node Out.	2 nodes Outage	3 Lines Outage	2 Lines and 1 Node Out.	1 Line and 2 Node Out.	3 Nodes Outage		
.01	819	8.2	2.7	.027	.0003	.038	.0004	.0000	•0000		
.05	819	40.9	2.7	.136	.0058	.038	.0019	.0001	.0000		
.10	819	81.9	2.7	•272	.0272	.038	.0038	.0004	.0000		
.15	819	122.8	2.7	.403	.0613	.038	.0056	.0008	.0001		
.20	819	163.8	2.7	•545	.1089	.038	.0075	.0015	.0003		
•25	819	204.7	2.7	.681	.1702	.038	.0094	.0024	.0006		
.30	819	245.7	2.7	.817	.2450	.038	.0113	.0034	.0010		
•35	819	286.6	2.7	•953	•3335	.038	.0132	.0046	.0016		
.40	819	327.5	2.7	1.09	.4356	.038	.0151	.0060	.0024		
•45	819	368.5	2.7	1.23	•5514	.038	.0169	.0076	.0034		
.50	819	409.4	2.7	1.36	.6807	. 038	.0188	.0094	.0047		
•55	819	450.4	2.7	1.50	.8236	.038	.0207	.0114	.0063		
.60	819	491.3	2.7	1.63	.9802	.038	.0226	.0136	.0081		
•75	819	532.3	2.7	1.77	1.15	.038	.0245	.0159	.0103		
.80	819	655.1	2.7	2.18	1.74	.038	.0301	.0241	.0193		
.85	819	696.0	2.7	2.31	1.97	.038	.0320	.0272	.0231		
.90	819	737.0	2.7	2.45	2.21	.038	.0339	.0305	.0274		
•95	819	777.9	2.7	2.59	2.46	.038	.0358	.0340	.0323		
1.00	819	819	2.7	2.7	2.7	.038	.038	.038	.038		

¹ All values are multiplied by 10⁻³

TABLE XII

TEMPORARY OUTAGE RATES FOR FIRST, SECOND AND THIRD-ORDER CUT-SETS¹

λ	FIRST-0	ORDER CUT	SECO	OND-ORDI	ER CUT	THIRD-ORDER CUT					
$\frac{\lambda_{\text{line}}}{\lambda_{\text{line}}}$	l Line Outage	l Node Outage	2 Lines Outage	1 Line and 1 Node Out.	2 Nodes Outage	3 Lines Outage	2 Lines and 1 Node Cut.	l Line and 2 Node Out.	3 Nodes Outage		
.01	1067	10.7	1.36	.0136	.0001	.0137	.0001	.0000	.0000.		
.05	1067	53.4	1.36	.0678	.0034	.0137	.0007	.0000	•0000		
.10	1067	106.7	1.36	.1356	.0136	.0137	.0014	.0001	•0000		
.15	1067	160.0	1.36	.2034	.0305	.0137	.0021	.0003	.0000		
.20	1067	213.4	1.36	.2713	•0543	.0137	.0027	.0005	.0001		
.25	1067	266.8	1.36	.3391	.0848	.0137	•0034	.0009	.0002		
•30	1067	320.1	1.36	.4069	.1221	.0137	.0041	.0012	•0004		
•35	1067	373.4	1.36	•4747	.1661	.0137	•C048	.0017	.0006		
.40	1067	426.8	1.36	•5425	.2170	.0137	.0055	.0022	•0009		
•45	1067	490.1	1.36	.6103	.2746	.0137	.0062	.0028	.0012		
•50	1067	533.5	1.36	.6781	.3391	.0137	.C068	.0034	.0017		
•55	1067	586.8	1.36	•7459	.4103	.0137	.0075	.0041	.0023		
.60	1067	640.2	1.36	.8138	.4883	.0137	.0082	.0049	.0030		
.65	1067	693.5	· 1.36	.8816	•5730	.0137	.0089	.0058	.0038		
.70	1067	746.9	1.36	•9494	.6046	.0137	.0096	.0067	.0047		
•75	1067	800.2	1.36	1.02	•7629	.0137	.0103	.0077	.0058		
.80	1067	853.6	1.36	1.09	.8680	.0137	.0109	.0087	.0070		
.85	1067	906.9	1.36	1.15	•9799	.0137	.0116	.0099	.0084		
.9 0	1067	960.3	1.36	1.22	1.10	.0137	.0123	.0111	.0100		
•95	1067	1013.6	1.36	1.29	1.22	.0137	.0130	.0123	.0117		
1.00	1067	1067	1.36	1.36	1.36	.0137	.0137	.0137	.0137		

¹ All values are multiplied by 10⁻³

4.9 <u>A branch-node cut-set algorithm</u>(46)

In section (4.6) is presented an algorithm for separable graphs which was an extension of the algorithm for nonseparable graphs but introducing the concept of articulation points. The algorithm of section (4.6) can be used for both cases except that for the case of nonseparable graphs it is necessary to specify that the graph has no articulation point. The algorithm of section (4.6) is employed to develop a new procedure to list the feasible branch-node cut-sets as well as branch(es) and node (s) cut-sets.

4.9.1 Basic definitions and assumptions

Previous algorithms give the minimal cut-set and the cut-node incidence matrices between the reference node and all other nodes of the system considering only edge failures. This information is used to develop a procedure which lists the minimal cut-set for a system with unreliable nodes. The concepts introduced in previous sections are still valid but new ones will be introduced for the branch-node cut-set generation algorithm.

A busbar-flag vector is defined for each minimal cut-set determined using the algorithm of section (4.6). This vector contains the information about the terminal nodes of the edges in each cut-set. Each nonproper minimal cut-set defined by that algorithm has constructed for it a busbar-flag vector BF_j , j = 2,...,NN. Table XIII shows the minimal cut-set matrix, the cut-node incidence matrix and the busbar-flag vectors for the reliability block diagram shown in Fig. 1.

By definition a proper minimal cut-set is a set of edges incident on the same node that, when they fail, only the common node fails. The algorithm in section (4.6) identifies such cuts by a cut-flag equal to 1. Each proper cut-set listed when considering only edge failures will also correspond to a cut-set including only the common node when nodes are assumed to be unreliable. This cut-set will be a first order cut-set representing a node failure. When considering separable graphs the failure of an articulation point will also disconnect all the tree nodes connected to that point.

TABLE XIII

Minimal cut-sets	Cut-node incidence	Busbar-flag vectors
1,2	2,3,4	110
1,3,4	2	311
2,3,4	3,4	121
2,3,5	3	131
4,5	4	112
1,3,5	2,4	221

MINIMAL CUT-SETS, CUT-NODE INCIDENCE AND BUSBAR-FLAG VECTORS OF FIG. 1

Considering the results of section (4.8) for the contribution of nodes, branches or branch-node failures on the temporary and permanent outage rates of each busbar in an electric power system, the algorithm for the minimal cut-set enumeration in such systems with unreliable nodes considers cut-sets up to the third-order and includes the following events:

First order cut-sets	:	one line failure
·		one node failure
Second order cut-sets	:	two line failures
		one line - one node failures

Third order cut-sets : three line failures two line - one node failures

From the possible 9 events mentioned in section (4.8) only those including single or multiple line contingency and/or single node failure will be considered. The others due to their small influence on the values of the reliability indices, will be excluded. The branch-node cut-set algorithm as presented in this section has been specially designed for the reliability evaluation of electric power systems.

4.9.2 Description of the algorithm

The information provided by the busbar-flag vector and the minimal cut-set and cut-node incidence matrices listed by the algorithm of section (4.6) allows us to extend such algorithm to include cut-sets produced by single node failure or branch-node failures.

Let us consider the results shown in table XIII to explain how this information is used to list node cut-sets and branch-node cut-sets. Once the algorithm of section (4.6) has been applied for the graph of Fig. 1 the minimal cut-set and cut-node incidence matrices listed in table XIII are available. Then the new algorithm uses each cut-set already listed to produce new combinations as follows: First the busbarflag vector is constructed for the cut-set under scan. Then all combinations are made replacing element(s) by the equivalent node using the busbar-flag vector and the edges contained in the cut-set. For example the cut-set (1,2) in table XIII has a busbar-flag vector $BF_{ij} = 1,1,0, j = 2,3,4$ that indicates that the cut-set has one edge incident on node 2 (edge 1) and the other incident on node 3 (edge 2). The reference node (numbered 1) assumed to be 100% reliable is therefore not included in this approach. In the new out-sets, nodes will be

numbered in increasing order starting from NE + 1 where NE is the number of edges in the graph. The number of components NC (edges and nodes) of the graph will be NC = NE + NN. With such conventions the cut-set (1,2) will generate two new cut-sets given by the sets (1,7)and (2,6). As indicated in table XIII cut-set (1,2) affects nodes 2,3 and 4. Then when an edge is replaced for its node to form a branch-node cut-set, the cut-node vector is also modified excluding the node that is now in the cut-set. This procedure gives the cut-node incidence vectors (2,4) and (3,4) for the branch-node cut-sets (1,7)and (2,6) respectively. The same process is repeated for each minimal cut-sets listed by the algorithm of section (4.6).

- The algorithmic procedure can be described in 5 steps: Step 1 : Generate the minimal cut-set and cut-node incidence matrices using the 5-setps algorithm of section (4,6). Let i be the variable identifying the number of a minimal cut-set, i = 1, NMC where NMC is the total number of minimal cutsets already generated. Initialize a cut-counter.
- Step 2 : Check the cut-counter. If it is less than or equal to NMC, take the next cut from the cut-set matrix of step 1, and go to step 3. If it is greater than NMC the algorithm terminates and all minimal node and branch-node cut-sets have been generated.
- Step 3 : Check the cut-flag produced in step 1 for the ith cut-set. If it is 0 the cut-set is not proper, then go to step 4. If it is 1, the i-cut is proper and must generate only one cut-set corresponding to a single failure of the common node

of all the edges in the i-cut. Construct a cut-node incidence vector equal to the same vector of the ith cut-set. Increment the cut-counter and go to step 2.

- Step 4 : Construct the busbar-flag vector for the ith cut. Generate all possible combinations of events having single node failure and/or branch(es)-node failures. Go to step 5.
- Step 5 : Compare each possible new cut-set with all cut-sets already generated that have the same node failure. Only if the possible cut-set has not been already generated and is a minimal one, consider it is a branch-node cut-set. Construct its cut-node incidence vector equal to that of the ith cut-set except for the term of the node having failed and any other tree node fed from this node. Repeat step 5 for all the combinations generated in step 4. Increment cut-counter and go to step 2.

To illustrate the procedure, table XIV presents the results step-by-step when the node and branch-node cut-set of Fig. 1 are generated by the algorithm. The busbar-flag vector and steps 4 and 5 are made only for non-proper minimal cut-sets. If the same procedure is applied for the proper cut-sets of Fig. 1 the results obtained are presented in table XV. This table shows that each proper minimal cut-set only generates one minimal cut-set, different from those generated by non-proper minimal cut-sets, which corresponds to the failure of the common nodes of the edges in the cut-set. The incidence of the nodes and branch-node cut-sets on the nodes of the graph is contained in a cut-node incidence matrix

TABLE XIV

GENERATION OF MINIMAL CUT-SET INCLUDING

4

SINGLE NODE OUTAGE OF FIG. 1

Old	cut	Busb	ar-fla	ag	Possible 1	new cut	N	ew cut	Cut-node	Observations		
No.	Elements	2	3	4	Elements	Nodes	No.	Components	incidence	·		
					2	2	7	2,6	3,4			
	1,2			0	1	3	8	1,7	2,4			
2	1,3,4				-	2	9	6	2	Proper cut		
					2	2				See cut 7		
3	2,3,4	2	2	1	4	3	10	4,7	4			
					2,3	4	11	2,3,8	3			
4	2,3,5				-	3	12	7	3	Proper cut		
Í					5	2	13	5,6	4			
5	1,3,5	2	2	1	1	3				See cut 8		
					1,3	4	14	1,3,8	2			
6	4,5				-	4	15	8	4	Proper cut		

TABLE XV

Olć	l cut	Busb	ar-fl	ag	Possible	new cut	Ne	ew cut	Cut-node			
No.	Elements	2	3	4	Elements	Nodes	No.	Components	incidence	OUSEL VALIOUS		
					-	2	16	6	2	See cut 9		
2	1,3,4	3	1	1	1,4	3	17	1,4,7	2,4	Non-minimum, see cut 8		
					1,3	4	18	1,3,8	2	See cut 14		
	0.7.5	_			2,5	2	19	2,5,6	3,4	Non-minimum, see cut 7		
- 4	2,2,2		2		-	3	20	7	3	See cut 12		
					2,3	4	21	2,3,8	3	See cut 11		
					, 5	2	22	5,6	4	See cut 13		
6	4,5	1	1	2	4	3	23	4,7	4	See cut 10		
			!		-	4	24	8'	4	See cut 15		

GENERATION OF CUT-SET FROM THE PROPER CUT-SET OF FIG. 1

1 All cuts mentioned are the new cuts in table XIV

÷

similar to that used for the algorithm of section (4.6). It can be seen in table XIV that different cut-sets can produce similar combinations. For example the first combination produced with the old cut No. 3 is similar to the one produced by the first combination of the old cut No. 1. Similarly a new combination can be minimal with respect to others previously generated. Therefore the previous combination must be discarded from the minimal cut-set list. Then each new combination produced by the algorithm must be compared with the previous ones to see that a cut-set will not be generated more than once or if it is minimal. However the number of comparisons is kept to a minimum because only cut-sets with the same node failure are used.

The approach presented for the node and branch-node cut-set enumeration is a combinational procedure that uses the results from the algorithm of section (4.6) to list all possible combinations of branches and nodes that can result in minimal cut-sets. It is an exhaustive procedure that generates all node and branch-node minimal cut-sets.

4.9.3 Tests

The computer program implementation of the algorithm in section (4.6) has been extended to include steps 2 to 5 of the new algorithm. The extension also uses the space-saving modification of section (4.5.2) for efficient storage of the minimal cut-set and cut-node incidence matrices.

Table XVI shows the computing times for two randomly generated graphs and for the West Venezuelan 115 kV system (graph 12). The same graphs were used when testing the algorithm of section (4.6) as shown in table VIII. Table XVI indicates that the inclusion of node and branch-node cut-sets does not increase the average cut-node generation time. Figures 13 to 15 list the minimal cut-set and cut-node incidence INPUT DATA

Е	SB	EB
- 1 2 3 4 5 6	1 1 2 3 7	2 3 4 7 4
8	-7	9
9 10	-4 -4	5 10
L1 L2	-9 -5	8 6

MINIMAL CUT MATRIX

CUT	ELEMEN	ITS	CUT	NO	DES							
1	1 2	2 5 Å	1	2	3	4	5	6	7	8	9	10
3	2	5 4	3	3	4	5	6	7	8	9	10	
5		5	5	2	4	5	6	7	8	9	10	
7	36	+ 2	7	4	5	6	10	U	2	10		٠
8	í	, + 6	8	2	á	5	-6	10				
9	2 1	+ 6	9	.3	7	8	9					
10	9		10	5	6							
11	12		11	6	_	_						
12	4 5	56	12	7	8	9						
13	11	>	13	8	•						•	
14	7 4	0	14	9	8							
15	2 13	τ	15	10 3	Λ	5	6	7	8	9	10	
17		4	10	ź	4	5	6	7	8	ģ	10	
18	13	•	18	2	-1	-	-	•		-		
19	3	4 14	19	4	5	6	7	8	9	10		
20	2	4 15	20	3	7	8	9					
21	. 14		21	3			_		_			
22	5 1	3	22	4	5	6	7	8	9	10		
23			23	2	4	2	ь	10				
24	4 2 7 9		24	1	8	9	10					
25	15		25	4	5	6	10					
20	6 1	3	20	4	5	6	10					
28	1	4 15	28	2		•						
29	4 (5 14	29	7	8	9						
30	2 18	3	30	3								
31	16		31	5	6							
32	17		32	6	_	-						
<u>33</u>	18		33	7	8	9						
34	19		34	8	0							
うう 76	20)) 74	20	σ							
20	~ ~ ~		70	117								

Figure 13. Output of the branch-node algorithm First numbering of nodes of graph 10.

CUT-NODE INCIDENCE MATRIX

.

INPUT	DATA

SB	EB
1	10
1	5
10	7
10	2
5	2
2	7
-2	8
-2	8
-7.	9
-7	- 3
-8	6
-9	4
	SB 1 10 10 5 2 -2 -7 -7 -7 -8 -9

MINIMAL	CUT	MATR	IX	CUT-NOD	E INC	CIDEN	ICE N	IATR	IX				
CUT	ELI	EMENT	S	CUT	NODI	ES							
1 2 3 4	1 10 12 2	2		1 2 3 4	2 3 4 5	3	4	5	6	7	8	9	10
5 6 7	1 4 2	5 5 4	6 6	5 6 7	2 2 5	3 6 2	4 8 6	6 8	7	8	9	10	
8 9	1 11 7	4	6	8 9	367	4	7	9	10			•	
10 11 12 13 14	2 2 7 9	8 3 8	5 4	10 11 12 13 14	7 2 5 8 9	2 6 4	4 8 6	9 7 8	3 7	4 3	9 4	9	
15 16 17 18 19	1 1 2 14	3 16 21	4	15 16 17 18	10 2 2 3 4	3 3	4 4	6 5	7 6	8 7	9 8	10 9	
20 21 22 23	16 1 5 13	13 21		20 21 22 23	+ 5 3 2 2	4 3 6	7 4 8	9 6	10 7	8	9		
24 25 26 27	2 4 2 1	13 6 4 4	16 18 18	24 25 26 27	5 2 5 10	6 2	8 6	8					
28 29 30 31	6 17 18 3	21 13		28 29 30 31	3 6 7 7	4 3 3	7 4 4	9 9 9					
32 33 34 35 36	3 4 19 20 21	4 5	16 18	32 33 34 35 36	2 2 8 9 10	6 6 6 4	8 8	7	3	4	9		

Figure 14. Output of the branch-node algorithm Second numbering of nodes of graph 10.

.

•

E	SB	EB
1	l	6
2	l	8
3	6	10
4	6	4
5	8	4
6	4	10
7	-4	3
8	-4	3
9	-10	2
10	-10	5
11	-3	9
12	-2	- 7

MINIMAL CUT MATRIX

CUT-NODE INCIDENCE MATRIX

CUT	ELI	EMENT	S	CUT	NODES								
1 2 3 4	1 9 7 10	2 8		1 2 3 4	2 2 3 5	3 7 9	4	5	6	7	8	9	10
5 6 7 8	1 2 4 12	3 3 5	4 4 6	5 6 7 8	6 2 4 7	3 3	4 9	5	7	8	9	10	
9 10 11 12	2 1 3 2	5 5 4 4	5 6	9 10 11 12 13	8 2 2 4 9	3 3 3	4 4 9	5 5 8	6 7	7 9	9 10	10	
14 15 16 17 18 19 20	3 1 2 1 13 14 16	6 4 17 19	6	14 15 16 17 18 19 20	10 6 2 2 3 5	2 10 3 7 9	7 2 4 4	5 7 5 5	5 7 6	8 7	9 9	10 10	
21 22 23 24 25 26	17 3 2 15 18	4 4	19 21	21 22 23 24 25 26	6 2 3 4 7 8	3 4 3	4 8 9	5 9	7	9	10		
20 27 28 29 30 31 32 33	1 5 3 4 2 4 20	15 17 15 5 15 6	21 19	20 27 28 29 30 31 32 33	2 2 2 3 8 4 9	5 3 5 4 3	6 4 7 9 9	7 5 10	10 7	9	10		
34 35 36	21 6 1	17 4	21	34 35 36	10 10 6	2 2	7 7	5 5					

Figure 15. Output of the branch-node algorithm Third numbering of nodes of graph 10.

matrices of graph 10 as they are output by the computer program. These figures show the node-numbering independence of the algorithm.

TABLE XVI

COMPUTATION TIMES FOR THE BRANCH-NODE CUT-SET ALGORITHM

Graph	Nodes	Elements	Configura- Number of tion index cut-nod		Total time (sec)	Average cut-node time (msec.)	
10	10	12	0.83	128	0.618	4.82	
11	11	14	0.79	252	0.831	3.29	
12	15	18	0.83	836	1.85	2.21	

4.9.4 <u>A practical application</u>

The branch-node algorithm was used to list all the minimal cut-sets and the cut-node incidence of the electric distribution system of Fig. 11 taken from [79]. The results are listed in Fig. 16 which include cutsets of all orders. Using the minimal cut-set approach to failure mode and effects analysis reported by Billinton and Grover⁽⁷⁹⁾, a complete analysis of the node failure effect on the reliability indices was made for the system configuration in Fig. 11.

To evaluate the reliability of a given electric distribution system, two variables are calculated for each node in the given system: the expected frequency of service interruptions and the expected duration of an interruption. It has been assume that loss of continuity at a node is the only mode of failure for the nodes. This condition can result from component permanent or temporary outages, overlapping permanent outages, temporary outages overlapping permanent outages and temporary and permanent outages overlapping preventive-maintenance outages.

MINIMAL CUTS						CUT-NODE	ICE			
CUT	ELEN	ENTS				NOD	ES			
1	1	7	8 3			2	3	4	5	6
34	2	_3 _3	7 4	8		3	4	5	6	
5	1 4	4 7	8			2 4	3. 5	6,		
7 8 9	4 2 1	5 3 5	5			4 3 2	4	4		
10 11	- 5 5	7 6	8 8			- 5 5	6	-		
12 13	6 4	7 6	8			6 4	5			
14 15 16	2 1 7	3 6 8	6 8 9	8		323	4 3 4	5 4 5	5	
17	i	8	13			2	3	4	5	
19 20 21	7 2	8 3	10 8	13		433	5 4	6 5		
22 23	10 4 1	9 10				2 2				
24 25 26	1 7 4	11 8 8	11 13			2 5 4	3 6 5			
27 28	11 5	9				43	4			
. 29 30 31	2 2 2	10 3 3	11 12			4 3 3	4			
32 33	1 7	12 12				26	3	4		
24 35 36	ン 12 13	8	ţ			5 5 6				
37 38	6	8 8	· 10 11			4	5			
59 40	4 6	12 8	9			4 3	4	5		

Figure 16. Minimal cut-set and cut-node for Fig. 11

. .

The equations⁽⁷⁹⁾ used incorporate a 2-state weather model and consider the following situations:

- (i) no repair during adverse weather

The analysis assumes that all lines have the same reliability parameters indicated in section (4.8). The permanent and temporary failure rates of the nodes were varied from 0.01 to 1.0 times the line failure rates in steps of 0.05. The nodes outage durations were given the same values as line durations. All nodes were assumed to have the same reliability parameters. Figure 17 plots the resultant values of permanent and temporary node outage rates and permanent node outage duration expressed in per-unit of the respective values for perfectly reliable nodes against the node failure rate expressed as percentage of the line failure rate. These curves indicate that the error in the node reliability indices can be appreciable even for highly reliable nodes and is most serious if the node failure rates are a sizeable percentage of the line failure rates. The errors in node 5 are so high due to the fact that when assuming perfectly reliable nodes, node 5 is outaged only from occurrence of cut-sets of third order, therefore the results considering node failure have a high degree of nodal dominance.

4.10 Conclusion

The introduction of the concept of cut-node incidence allows the development of a systematic and efficient procedure for the enumeration of the minimal cut-sets between an origin and all other nodes in separable and nonseparable graphs. Each time a cut-set is generated we compute new reliability bounds for each node affected for that cut-set. When all cut-sets have been generated all node reliability bounds are available.



Fig. 17. Analysis of the influence of node outages on the reliability indices of the system in Fig. 11.

The set of proper cuts defined for the set of edges incident on each node proves to be a fundamental set from which all other cutsets can be obtained by ring sum operations. These operations are reduced to a minimum if only those cuts with at least one edge in common are combined. In this sense the introduction of the index cut is an efficient criterion.

The cascade nature of acyclic directed graphs and the use of a breadth-first search method to fan out all edges at each node permits the listing of cut-sets in these graphs by a combinational approach that proves to be very efficient, compared with a similar technique reported in the literature.

Any physical system where the nodes are also subject to failures requires the enumeration of the feasible branch-node cut-sets for its reliability evaluation. Otherwise the results of such evaluation are meaningless. Unreliable nodes can be easily handled considering that an s-t branch-node cut-set is a minimal set of components (branches and nodes) that break all possible s-t paths. Once all branch cut-sets have been defined the use of a busbar-flag vector to indicate the terminal nodes of the branches in each cut-set expedites the generation of all possible branch-node combinations.

The efficiency of the minimal cut-set algorithms presented in this chapter cannot be measured entirely in terms of the number of nodes in the graph but rather in terms of the graph configuration index (number of nodes/number of branches). Graphs with high indices will be solved faster than those with low indices. The space requirement of the computer program implementation of the algorithms is determined by the number of cuts they may generate. A simple storage scheme based on the binary nature of the minimal cut-set and cut-node incidence matrices overcomes such restrictions.

CHAPTER V

VARIANCE AND APPROXIMATE CONFIDENCE LIMITS FOR THE PROBABILITY AND FREQUENCY OF SYSTEM FAILURE

5.1 Introduction

Reliability component data is generally obtained from two sources, the failure times of various items in a population placed on a life test⁽⁴⁶⁾ or field reports⁽⁴⁷⁾ listing operating hours of replaced components in equipment already in use. A very good way to present these data is to compute and plot either the failure density function or the failure rate as a function of time. Since both functions are continuous variables we first compute a piecewise-continuous failure density function and a failure rate from the data. Study of these piecewise-continuous functions is followed by the choice of a continuous model which fits the data satisfactorily and allows us to draw conclusions from test data on the behaviour of other similar components. Fitting a model to the data can be done either on the basis of experience, intuition and inspection of the appropriate graphs or through statistical techniques which can be used to efficiently process data and obtain "best" repeatable values for model parameters.

A point estimate of a parameter X is a single number which is our "best" estimate of the parameter. A measure of the variability of such a value is called the variance. If X varies relatively little or by a large amount, the variance will be relatively small or large respectively. If we are somewhat more realistic we may wish to quote an interval i.e., an interval estimate, into which the parameter probably falls. To be more precise we measure our sureness by the probability that the parameter falls within the interval. This probability is called the

"confidence"or "confidence coefficient". For a single component it is accepted that a bare statement of its estimated probability of failure during the period of operation is insufficient. A standard used for component procurement, the British Standard BS 4200 or the United States MIL STD 785, makes it very clear than an accuracy measure for the failure probability estimate is also necessary. This is usually in the form of a confidence limit.

Actually in many cases the interval estimate of the component parameters is only an intermediate step in solving the real problem of formulating an interval estimate for the system reliability. This chapter will focus on the development of computational methods for the variance and approximate confidence intervals for the probability and frequency of system failure taking into account the uncertainty of the component parameters. It is assumed that early testing has been performed on components and estimates and variances of component reliabilities are available i.e., this chapter presents methods for constructing approximate confidence limits on the system reliability indices from component test results.

5.2 Distribution of the failure and repair models

There exists a general agreement that the exponential distribution adequately describes the time-to-failure of most complex systems of solid state design but it is less relevant for electromechanical equipment where wear-out effects need to be considered. Hence, for a more satisfactory analysis it is necessary to consider the use of timedependent distributions such as the Weibull distribution, whose density function is

$$f(t) = K t^{m} e^{-K t^{m+1} / m + 1}$$
 (5.1)

The parameter m determines the shape of the distribution and parameter K is a scale-change. When m = 0, the distribution becomes exponential. However, the exponential distribution, if not known to be the case, is usually a safe assumption for highly complex equipment⁽⁴⁸⁾. It is also pertinent to remark that if routine maintenance is carried out sufficiently frequently then it can be assumed that defects due to time-dependent causes do not occur.

There is considerable controversy over the distribution of timeto-repair. The most commonly used are the exponential, the log-normal and the Erlang distributions. The log-normal has the serious limitation that it is very difficult to work with analytically in conjunction with another distribution. Since in many cases it is difficult, if not impossible, to distinguish between a sample from a log-normal distribution and a sample from an exponential distribution it would seem preferable ` to assume the exponential because it lends itself readily to an analytic approach. The Erlang family of service time distribution is defined by

$$f(t;\mu,k) = \frac{\mu(\mu t)^{k-1}}{(k-1)!} e^{-\mu t}$$
(5.2)

where k is the number of stages. The limiting cases for this family occur when k = 1 and $k = \infty$. When k = 1, $f(t;\mu,k)$ reduces to the exponential and when $k = \infty$, $f(t;\mu,k)$ is a straight line intersecting the t-axis at $1/\mu$. The Gamma-distribution has a density function defined by

$$f(x) = \begin{cases} 0 & \text{for } x \leq 0 \\ \frac{\beta^{\alpha}}{\Gamma(\alpha)} x^{\alpha-1} e^{-\beta x} & \text{for } x > 0 \end{cases}$$
(5.3)

therefore, the exponential distribution (with $\alpha = 1$) and the so-called Erlang-distribution (in the case of integer α) are particular cases of a \int -distribution.
Both the nonrepairable and the repairable approaches described in this chapter assume the exponential distribution for the time-tofailure and for the time-to-repair. However the repairable case is also applicable if the Weibull or Gamma distributions are used for any one of both times.

.5.3 The exponential distribution in reliability

The exponential-distribution plays a key role in reliability and calculations because it describes so well the behaviour of components and systems in their useful life period i.e., when they display approximately constant failure rates. Its great advantage over other any statistical distributions is the single parameter λ , or its reciprocal m, which fully and completely describes a given exponential function.

The exponential density function is given by

$$f(t) = \lambda e^{-\lambda t} \quad t \geqslant 0, \quad \lambda > 0 \qquad (5.4)$$

where λ is the constant failure rate. A further advantage of the exponential distribution is that it is independent of the age of a component or system as long as they are operated during the period of constant failure rate.

5.3.1 Estimate and variance of the parameter

The fact that the exponential distribution is a single parameter distribution makes reliability testing of exponential equipment comparatively simple because all that is needed is to determine the value of λ by a test.

Suppose we have a sample $X = (x_1, x_2, \dots, x_n)$ of n independent observations drawn randomly from an exponential distribution with the density function of Eqn. (5.4), where λ is an unknown parameter and we

wish to estimate it using the sample data. Several criteria of estimation⁽⁴⁹⁾ have led to estimates of λ of the form d_n/S , where $S = \sum_{1}^{n} x_i$ is a sufficient statistic for λ and d_n is a function of n. The whole topic of estimation and testing schemes are discussed in many places, e.g. [50] or [51]. The most frequently used estimators are the maximum likelihood estimator (MLE) and the minimum variance unbiased estimator (MVUE).

Lloyd and Lipow⁽⁵²⁾ summarized the methods for estimating the parameters of the gamma distribution from complete samples by the method of maximum likelihood. Epstein and Sobel⁽⁴⁶⁾ derived MLE for the parameter of the exponential distribution based on censored samples. Sathe and Varde⁽⁵³⁾ derived the MVUE for the same distributions, giving the expected value and the variance for the parameter in each case.

5.3.2 Reliability estimate and variance

Let t_1, t_2, \dots, t_n be identically distributed independent random variables having an exponential distribution. These random variables represent the life-lengths of n identical systems or components. Let θ denote the expected life-length, or mean time between failures (MTBF) of the systems or components under consideration; $0 < \theta < \infty$. The reliability function of the system or component is

$$R(t,\theta) = \exp(-t/\theta) \quad 0 \leq t < \infty \quad (5.5)$$

The commonly used estimate of reliability in the exponential case is the maximum likelihood estimate \hat{R} , that for a given mission time is

$$\hat{R} = \exp\left(-t_{n}/\hat{\theta}\right)$$
 (5.6)

where $\hat{\theta}$ is the sample mean given by

$$\hat{\theta} = \sum_{i=1}^{n} t_i/n$$
 (5.7)

Though the estimate \hat{R} has all the properties of MLE it has one serious drawback for application in reliability: it is biased. Pugh⁽⁵⁴⁾ has shown that the best estimate of R is not \hat{R} but

$$\hat{R}^* = (1 - \frac{t_m}{n\hat{\theta}})^{n-1}$$
 (5.8)

where $\hat{\mathbf{R}}^{\star} = 0$ if $\hat{\boldsymbol{\theta}} \boldsymbol{\boldsymbol{\xi}} \mathbf{t}_{m}/n$.

The probability density function of the complete sufficient statistic S = $\sum t_i$ is gamma-distributed g(1/ θ ,n). Accordingly

$$E\{\hat{R}^{*2}(t_{m},S)\} = \frac{1}{(n-1)!} \int_{t_{m}}^{\infty} (1 - \frac{t_{m}}{\theta x})^{2n-2} e^{-x} x^{n-1} dx$$
(5.9)

and the variance of the MVU estimator \hat{R}^* (t_m,S) is obtained by subtracting $e^{-2t_m/\theta}$ from Eqn. (5.9). For numerical computations it is more convenient to apply the Gauss-Laguerre quadrature method (55) for determining the integral of Eqn. (5.9).

5.4 Minimal cut-set basic formulae

In the minimal cut-set approach to system reliability the probability of system failure (30) is expressed as

$$P_{f} = Pr \left\{ \overline{c}_{1} + \overline{c}_{2} + \dots + \overline{c}_{n} \right\}$$
(5.10)

where the + indicates the logic sum or union. The mutually exclusive approximation provides an upper bound to the probability of system failure given by

$$P_{f} \ll \sum_{i=1}^{NMC} Pr \{ \overline{C}_{i} \}$$
 (5.11)

where NMC is the total number of minimal cut-set.

Similarly, the frequency of system failure (16) is expressed as

$$f_{f} = f \{ S_{1} + S_{2} + \dots + S_{n} \}$$
 (5.12)

where each S_i represents a subset of the state space S formed by all \overline{C}_i and

$$f(S_{i}) = \Pr\{\overline{C}_{i}\} \cdot \sum_{j \in C_{i}} \mu_{j}$$

The mutually exclusive approximation yields an upper found for $f_{\mathbf{f}}$ given by

$$f_{f} \notin \sum_{i=1}^{NMC} \left\{ \Pr\left(\overline{C}_{i}\right) \cdot \mu_{i} \right\}$$
(5.13)

where $\overline{\mu}_i$ is the sum of μ_j 's over all j εC_i . The disjoint approximation is preferred here because the simple relationships provided by Eqns. (5.11) and (5.13) with respect to Pr { \overline{C}_i } and f { S_i } respectively, ease the computation of the expected value and variance of P_f and f_f.

If λ_j and μ_j represent the failure and repair rate of the jth element the steady state probabilities⁽⁵⁴⁾ of that element being in the up state P_{u_j} or in the down state P_{d_j} are

$$P_{u_{j}} = \frac{\mu_{j}}{\lambda_{j} + \mu_{j}}$$
(5.14)

and

$$P_{d_j} = \frac{\lambda_j}{\lambda_j + \mu_j}$$
(5.15)

 $P_{u_{j}}$ and $P_{d_{j}}$ can be interpreted as follows: As the operating time of the jth element approaches infinity, the fraction of time in the up state approaches $P_{u_{j}}$ and the fraction of time in the down state approaches $P_{d_{i}}$.

For the nonrepairable condition of an element with the exponential failure model, the probability of failure (17) in the time interval from 0 to t_m is given by

$$Q_j(t_m) = 1 - e^{-\lambda_j t_m}$$
 (5.16)

and the probability of system failure for the same interval using the minimal cut-set approach can be obtained from

$$P_{f} = \sum_{i=1}^{NMC} \left\{ \prod_{j \in i} Q_{j}(t_{m}) \right\}$$
(5.17)

For the repairable case, the system is assumed to have reached an equilibrium (steady state) condition of failure and repair. The steady state probability and frequency of system failure given by the minimal cut-set approach are obtained from

$$P_{f} = \sum_{i=1}^{NMC} \left\{ \prod_{j \in i} P_{d_{j}} \right\}$$
(5.18)

$$\mathbf{f}_{\mathbf{f}} = \sum_{\mathbf{i}=\mathbf{l}}^{NMC} \left\{ \prod_{\mathbf{j}\in\mathbf{i}} \mathbf{P}_{\mathbf{d}_{\mathbf{j}}} \cdot \sum_{\mathbf{j}\in\mathbf{i}} \boldsymbol{\mu}_{\mathbf{j}} \right\}$$
(5.19)

For both repairable and nonrepairable conditions the highest contingency to be evaluated will be the third order cut-sets. Higher orders are assumed to produce negligible effects on the reliability indices.

5.5 Problem statement

We are concerned with the general problem of estimating the probability and frequency of failure that a complex system has during its operational life. The system will consist of a number of components. logically interconnected in some fashion, with particular patterns of component failure causing system failure.

Component failure and repair rates obtained from life-tests or field reports are subject to uncertainty and may most properly be thought of as random variables. It follows from Eqns. (5.17) to (5.19) that P_f and f_f which are functions of these parameters are also subject to uncertainty and must be considered as random variables too. Presently, the expected or average values of the component parameters are used to compute the expected values of the P_f and f_f random variables. The expected values of P_f and f_f provide no information as to the range of values over which P_f and f_f might vary due to variations in component failure and repair rates. Such analysis requires a method for computing not only a point-estimate of both reliability indices but also interval estimates. The interval estimate of P_f and f_f will give upper and lower bounds such that these random variables lie between these limits with a specified probability.

The exact solution of this problem requires complete information about the internal variables λ_j , μ_j . This means that the distribution of each λ_j and μ_j must be known and how changes occur as a function of time. The amount of information required for the exact solution generally will not be available. Also as Eqns. (5.17) to (5.19) involve nonlinear operations, a solution in closed form is generally not possible. These are the reasons why the exact solution is not discussed in this thesis.

Considering that the parallel-series arrangement of components in the minimal cut-set approach adequately represent the system with statistically independent components, we now show how these interval estimates for the probability and frequency of systemfailure can be calculated by approximate methods.

A deterministic (nonprobabilistic) approach to this problem is first discussed. Two different types of approximation will be presented. The first technique involves the orthodox statistical procedure of characterizing distributions by their low order moments and uses the conservative Chebyshevinequality to bound the probabilities that the random variables lie within a certain range. The second technique is to apply Monte Carlo simulation based on common probability models, with which component test data may be translated into approximate system reliability limits at any confidence level.

5.6 Review of the literature

Methods of obtaining confidence bounds on the reliability of nonmaintained systems are surveyed and numerical comparisons are given by $Mann^{(56)}$. It is assumed that failure data have been collected from life tests performed on prototypes of the various components which make up the system, but that the system has not been tested as a whole for economic reasons or simply that it is virtually impossible to test the entire system without destroying it. This paper represents a very good summary of the state of the art and discusses methods applicable to series and/or parallel as well as more logically complex systems.

Rosenblatt⁽⁵⁷⁾ treats the problem of estimation of a probability R interpreted as system reliability when the estimate is to be based on data obtained from component tests. The problem is formulated in a general way leading to a widely applicable method for distribution-free estimation of R and calculation of approximate nonparametric confidence intervals. It discusses comparisons with exact and alternative approximate methods. The method is also well discussed by Mann⁽⁵⁶⁾.

Murchland and Weber⁽⁵⁸⁾ describe a method for the variance of the probability of failure of a complex system represented by a faulttree with independent components using a moment method when the components are not repairable. In the repairable case, a method for approximately determining the variance of the expected number of failures is indicated, for which the authors suggest the use of Chebyshev inequality rather than picking any particular assumptions concerning the distribution of the unbiased simulation estimator of system reliability. Their bounds involve the expression for the exact rather than the asymptotic variance of the estimator of the system reliability. Their procedure presents no particular computational difficulties as long as the system remains logically simple. Calculation of their variance estimate for the reliability estimator becomes very complicated, however, as the system increases in complexity.

Burnett and Wales⁽⁵⁹⁾ present a method for constructing statistical confidence limits for the system reliability from component test results. They suggest Monte Carlo simulation of the fiducial distribution of system reliability given the subsystem failure for the series-system model and also for other models which are logically more complex. The method developed is only for the case of components with exponentially distributed failures and assumes: (i) individual components fail independently, (ii) a mathematical expression relating system reliability and component reliabilities can be written, (iii) estimates of component reliabilities are available from earlier testing and (iv) these estimates of omponent reliabilities have known sampling distributions. The method generates the numerical sampling distributions of component reliability estimates from which is selected a value for each component that is substituted into the mathematical relationship between system-component

reliabilities to give one value for the system reliability. This process is repeated until the numerical sampling distribution of the system reliability estimate is generated. Confidence limits are then computed on this numerical distribution.

Levy and Moore⁽⁶⁰⁾ extended the previous Monte Carlo technique to systems whose component failures follow the normal, log-normal, gamma and Weibull probability distribution. They also assume that individual components fail independently and that failure data is available from component tests. The parameter estimates are assumed to be obtained by the method of maximum likelihood based on life tests from a complete sample or from a censored sample where the distribution of the estimator is known.

Both techniques (59),(60) based in the fiducial method are optimum for one component and approach the optimum confidence bounds as the number of failures increases for all components. They are conservative in general but provide a method for obtaining confidence limits of systems composed of components whose failure patterns are represented by different families of mathematical models. Both are designed specially for series and/or parallel combinations but complex configurations must be broken down, by application of Bayes theorem, into combinations of these types.

Kamat and Riley⁽⁶¹⁾ present a Monte Carlo simulation procedure for non-repairable systems represented by its minimal tie-sets when the components are assumed to be statistically independent. The method is a synthetic one, in that the failure times of individual components are generated and then used to determine the success or failure of the system. Each of these component failure times is converted to the Boolean state representation of success or failure by comparing it with the required operation time. The success or failure of each component

along any given minimal tie-set is checked from its state representation. If all the components along the tie-set are a success, the system is a success. If at least one of these components is a failure, another minimal tie-set is checked. This procedure is continued until either the system success has been identified or all the minimal tie-sets have been checked with each tie-set having at least one failed component. Replications of this procedure will yield n_s success and n_f failures of the system and the reliability estimate for a mission time t_m is given by $R(t_m) = n_s/(n_s + n_f)$. The confidence limits are obtained using the statistical normal approximation to the binomial distribution. The confidence interval obtained is narrowed by increasing the number of replications.

5.7 Deterministic approach

The simple first approach to the problem of section (5.5) is to ignore the probabilistic nature of the problem and attempt a deterministic solution. One technique is to assume that component parameters have distributions entirely concentrated between minimum and maximum values. This technique, called here the worst-case approach, makes an honest attempt to determine how much change there will be in the reliability indices if all the components are at their extremes and combine in the worst possible manner.

We restrict our analysis to the contribution to the reliability indices of minimal cut-sets up to the third order.

Let us consider the case of repairable components where the steadystate probability of the i-th component being in the down state is given by

$$P_{d_{i}} = \frac{\lambda_{i}}{\lambda_{i} + \mu_{i}}$$

If σ_{λ} and σ_{μ} represent the standard deviations of the failure and repair rates respectively, it seems reasonable to use $\lambda_{i} \stackrel{+}{=} \sigma_{\lambda_{i}}$ for λ_{i} and λ_{i} , $\mu_{i} \stackrel{+}{=} \sigma_{\mu_{i}}$ for μ_{i} and μ_{i} , respectively. The minimal cutset approach to the probability and frequency of system failure are given by Eqns. (5.17) to (5.19). Table I summarises the resulting expressions for minimal cut-sets up to third order of the steady-state probability of failure. Capital letters refer to maximum values and minor ones to minimum values. Symbol (^) refers to the expected value of the parameter.

TABLE I

MINIMUM AND MAXIMUM VALUES FOR THE STEADY STATE PROBABILITY OF FAILURE

Cut-set order	Point-estimate	minimum	Maximum
1	$\frac{1}{1 + \hat{\mu}/\hat{\lambda}}$	$\frac{1}{1 + \hat{\mu}_{M} / \hat{\lambda}_{m}}$	$\frac{1}{1 + \hat{\mu}_{m}/\hat{\lambda}_{M}}$
2	$\frac{2}{\prod_{j=1}^{2} \frac{1}{1 + \hat{\mu}_{j}/\hat{\lambda}_{j}}}$	$\frac{2}{\prod_{j=1}^{l} \frac{1}{1 + \hat{\mu}_{j}/\hat{\lambda}_{j_{m}}}}$	$\frac{1}{\prod_{j=1}^{2}} \frac{1}{1 + \hat{\mu}_{j_{m}}/\hat{\lambda}_{j_{M}}}$
3	$\int_{j=1}^{3} \frac{1}{1 + \hat{\mu}_{j}/\hat{\lambda}_{j}}$	$\int_{j=1}^{3} \frac{1}{1 + \hat{\mu}_{j_{M}}/\hat{\lambda}_{j_{m}}}$	$\prod_{j=1}^{3} \frac{1}{1 + \hat{\mu}_{j_m} / \hat{\lambda}_{j_M}}$

Referring to the probability of failure of the system the idea is to combine the minimum and maximum values of the minimal cut-sets in the worst possible manner. A little thought shows that these values are given by

$$P_{f_{M}} = \sum_{i=1}^{NMC} Pr \{ C_{i} \}_{M}$$
 (5.20)

$$P_{f_{m}} = \sum_{i=1}^{NMC} Pr \{ c_{i} \}_{m}$$
 (5.21)

Similarly for the probability of failure of the system in the time interval from 0 to t_m , it seems reasonable to use, for each component, $Q_j(t_m) \stackrel{+}{=} \sigma_{Q_j}$ for $Q_j(t_m)_{Max}$ and $Q_j(t_m)_{min}$ respectively. In such cases general expressions for minimal cut-sets will be NMC

$$P_{f_{M}}(t_{m}) = \sum_{i=1}^{\infty} \prod_{j \in i} Q_{j}(t_{m})_{M}$$
(5.22)

$$P_{f_{m}}(t_{m}) = \sum_{i=1}^{NMC} \prod_{j \in i} Q_{j}(t_{m})_{m}$$
(5.23)

However, all one can say is that P_f probably lies between the values P_f_M and P_f_m . Since the statistics are unknown no confidence coefficient can be stated for this reliability index.

If the reliability index is a rather complicated function involving many parameters, as it is in the case for the frequency of system failure, it is hard to decide by inspection which combination of values gives a minimum and a maximum value of the index. In such a case, approximating the reliability index by the first few terms of a Taylor series expansion simplifies the problem. For the purposes of the deterministic approach it should be sufficient to retain the constant and the first-order terms and for a function of n variables these terms are given by

$$f(\Delta x_1, \Delta x_2, \dots, \Delta x_n) = f_{bv} + \sum_{i=1}^n (\partial f/\partial x_i)_{bv} \Delta x_i$$

For the minimal cut-set approach we need to evaluate the partial derivatives for cuts up to the third-order. Both the function and its partial derivatives are to be evaluated at the best value (bv) of the variables.

First-order cut-set

A point-estimate is given by

$${}^{f}C1 = \frac{\hat{\lambda}_{1}}{\hat{\lambda}_{1} + \hat{\mu}_{1}} \cdot \hat{\mu}_{1}$$
 (5.24)

The coefficients from the first partial derivatives designated by b_{f_1} and b_{f_2} are

$$b_{f_1} = (\partial f_{C1} / \partial \lambda_1)_{bv} = (\hat{\mu}_1 / \hat{\lambda}_1 + \hat{\mu}_1)^2$$
 (5.25)

$$b_{f_2} = (\partial f_{C1} / \partial \mu_1)_{bv} = (\hat{\lambda}_1 / \hat{\lambda}_1 + \hat{\mu}_1)^2$$
 (5.26)

Using worst-case techniques the minimum and maximum values for the frequency of the first-order cut-set will be

$$f_{Cl_m} = \hat{f}_{Cl} - b_{f_1} \Delta \lambda_1 - b_{f_2} \Delta \mu_1$$
 (5.27)

$$f_{Cl_{M}} = \hat{f}_{Cl} + b_{f_{1}} \Delta \lambda_{1} + b_{f_{2}} \Delta \mu_{1}$$
 (5.28)

where $\Delta\lambda_1$ and $\Delta\mu_1$ are substituted as positive quantities, made equal in this case to the standard deviation of the respective parameter.

Second-order cut-set

A single-point estimate is

$$\hat{f}_{C2} = \frac{\hat{\lambda}_1}{\hat{\lambda}_1 + \hat{\mu}_1} \cdot \frac{\hat{\lambda}_2}{\hat{\lambda}_2 + \hat{\mu}_2} \cdot (\hat{\mu}_1 + \hat{\mu}_2)$$
(5.29)

The coefficients from the first partial derivatives are

$$b_{fl} = (\partial f_{C2} / \partial \lambda_1)_{bv} = \frac{\hat{\lambda}_2}{\hat{\lambda}_2 + \hat{\mu}_2} \cdot \frac{\hat{\mu}_1}{(\hat{\lambda}_1 + \hat{\mu}_1)^2} \cdot (\hat{\mu}_1 + \hat{\mu}_2) \quad (5.30)$$

$$b_{f2} = (\partial f_{C2} / \partial \mu_1)_{bv} = \frac{\hat{\lambda}_2}{\hat{\lambda}_2 + \hat{\mu}_2} \cdot \frac{\hat{\lambda}_1}{(\hat{\lambda}_1 + \hat{\mu}_1)^2} \cdot (\hat{\lambda}_1 - \hat{\mu}_2) \quad (5.31)$$

$$b_{f3} = (\partial f_{C2} / \partial \lambda_2)_{bv} = \frac{\hat{\lambda}_1}{\hat{\lambda}_1 + \hat{\mu}_1} \cdot \frac{\hat{\mu}_2}{(\hat{\lambda}_2 + \hat{\mu}_2)^2} \cdot (\hat{\mu}_1 + \hat{\mu}_2)$$
 (5.32)

$$b_{f4} = (\partial f_{C2} / \partial \mu_2)_{bv} = \frac{\hat{\lambda}_1}{\hat{\lambda}_1 + \hat{\mu}_1} \cdot \frac{\hat{\lambda}_2}{(\hat{\lambda}_2 + \hat{\mu}_2)^2} \cdot (\hat{\lambda}_2 - \hat{\mu}_1)$$
 (5.33)

Using worst-case techniques the minimum and maximum values for the frequency of the second-order cut-set are

$$\mathbf{f}_{C2_{m}} = \hat{\mathbf{f}}_{C2} - \Delta \mathbf{f}_{C2}$$
(5.34)

$$\mathbf{f}_{\mathrm{C2}_{\mathrm{M}}} = \hat{\mathbf{f}}_{\mathrm{C2}} + \Delta \mathbf{f}_{\mathrm{C2}}$$
(5.35)

where Δf_{C2} is given by

$$\Delta f_{C2} = b_{f1}\Delta\lambda_1 + b_{f2}\Delta\mu_1 + b_{f3}\Delta\lambda_2 + b_{f4}\Delta\mu_2 \qquad (5.36)$$

The basic assumption that component repair rates are much larger than failure rates would give a negative sign to coefficients b_{f2} and b_{f4} . Therefore, $\Delta\lambda_j$ and $\Delta\mu_j$, j = 1,2 are substituted equal to the respective standard deviation of the parameter with the sign to combine in the worst possible manner.

Third-order cut-set

A single-point estimate is

$$f_{C3} = \frac{\hat{\lambda}_{1}}{\hat{\lambda}_{1} + \hat{\mu}_{1}} \cdot \frac{\hat{\lambda}_{2}}{\hat{\lambda}_{2} + \hat{\mu}_{2}} \cdot \frac{\hat{\lambda}_{3}}{\hat{\lambda}_{3} + \hat{\mu}_{3}} \cdot (\hat{\mu}_{1} + \hat{\mu}_{2} + \hat{\mu}_{3})$$
(5.37)

The coefficients from the first partial derivatives with respect to λ_i and μ_i , i=1,2,3 designated by b_{fj} and b_{fk} , j = 1,3,5; k = 2,4,6 respectively, are

$$b_{f1} = \frac{\hat{\lambda}_2}{\hat{\lambda}_2 + \hat{\mu}_2} \cdot \frac{\hat{\lambda}_3}{\hat{\lambda}_3 + \hat{\mu}_3} \cdot \frac{\hat{\mu}_1}{(\hat{\lambda}_1 + \hat{\mu}_1)^2} \cdot (\hat{\mu}_1 + \hat{\mu}_2 + \hat{\mu}_3) \quad (5.38)$$

$$b_{f2} = \frac{\hat{\lambda}_2}{\hat{\lambda}_2 + \hat{\mu}_2} \cdot \frac{\hat{\lambda}_3}{\hat{\lambda}_3 + \hat{\mu}_3} \cdot \frac{\hat{\lambda}_1}{(\hat{\lambda}_1 + \hat{\mu}_1)^2} \cdot (\hat{\lambda}_1 - \hat{\mu}_2 - \hat{\mu}_3) \quad (5.39)$$

$${}^{b}_{f3} = \frac{\hat{\lambda}_{1}}{\hat{\lambda}_{1} + \hat{\mu}_{1}} \cdot \frac{\hat{\lambda}_{3}}{\hat{\lambda}_{3} + \hat{\mu}_{3}} \cdot \frac{\hat{\mu}_{2}}{(\hat{\lambda}_{2} + \hat{\mu}_{2})^{2}} \cdot (\hat{\mu}_{1} + \hat{\mu}_{2} + \hat{\mu}_{3}) \quad (5.40)$$

$${}^{\mathrm{b}}_{\mathrm{f4}} = \frac{\hat{\lambda}_{1}}{\hat{\lambda}_{1} + \hat{\mu}_{1}} \cdot \frac{\hat{\lambda}_{3}}{\hat{\lambda}_{3} + \hat{\mu}_{3}} \cdot \frac{\hat{\lambda}_{2}}{(\hat{\lambda}_{2} + \hat{\mu}_{2})^{2}} \cdot (\hat{\lambda}_{2} - \hat{\mu}_{1} - \hat{\mu}_{3}) \quad (5.41)$$

$${}^{\mathrm{b}}_{\mathrm{f5}} = \frac{\hat{\lambda}_{1}}{\hat{\lambda}_{1} + \hat{\mu}_{1}} \cdot \frac{\hat{\lambda}_{2}}{\hat{\lambda}_{2} + \hat{\mu}_{2}} \cdot \frac{\hat{\mu}_{3}}{(\hat{\lambda}_{3} + \hat{\mu}_{3})^{2}} \cdot (\hat{\mu}_{1} + \hat{\mu}_{2} + \hat{\mu}_{3}) \qquad (5.42)$$

$$b_{f6} = \frac{\hat{\lambda}_1}{\hat{\lambda}_1 + \hat{\mu}_1} \cdot \frac{\hat{\lambda}_2}{\hat{\lambda}_2 + \hat{\mu}_2} \cdot \frac{\hat{\lambda}_3}{(\hat{\lambda}_3 + \hat{\mu}_3)^2} \cdot (\hat{\lambda}_3 - \hat{\mu}_1 - \hat{\mu}_2) \quad (5.43)$$

The minimum and maximum values for the frequency of the thirdorder minimal cut-set are

$$f_{C3_{m}} = f_{C3} - \Delta f_{C3}$$
 (5.44)

$$\mathbf{f}_{C3_{M}} = \mathbf{f}_{C3} + \Delta \mathbf{f}_{C3}$$
(5.45)

where

$$\Delta f_{C3} = \sum_{i=1}^{3} b_{fj} \Delta \lambda_i + \sum_{i=1}^{3} b_{fk} \Delta \mu_i \qquad (5.46)$$

Again $\Delta\lambda_{i}$ and $\Delta\mu_{i}$ are combined to give the worst possible values.

The minimum and maximum values for the frequency of system failure are

$$\mathbf{f}_{\mathbf{f}_{m}} = \sum_{i=1}^{NMC} \mathbf{f}_{Ci_{m}}$$
(5.47)

$$\mathbf{f}_{\mathbf{f}_{\mathbf{M}}} = \sum_{j=1}^{\mathbf{NMC}} \mathbf{f}_{\mathbf{Ci}_{\mathbf{M}}}$$
(5.48)

5.8 Approximate statistical approach - Moment method

377.00

To determine $(1 - \alpha)$ symmetrical confidence limits on a random variable x, the $\alpha/2$ and $(1 - \alpha/2)$ percentiles of the cumulative distribution F(x) must be found by solving the equations

$$F(x_0) - F(0) = \alpha/2$$

 $F(x_1) - F(0) = 1 - \alpha/2$

where

$$\alpha$$
 is the level of significance
 x_0 is the $\alpha/2$ percentile of the x distribution
 x_1 is the $(1 - \alpha/2)$ percentile of the x distribution

Thus, a knowledge of F(x) contains all the information necessary to compute confidence limits for a random variable. One way to alleviate the difficulties of the exact solution is by using the statistical procedure of characterizing a distribution function by their low order moments.

Since the probability and frequency of system failure given by Eqns. (5.17) to (5.19) are non-linear functions in each component parameter, approximate formulae can be given for the function moments in terms of the component moments by the Taylor series approximation technique.

Considering a general function $f(x_1, x_2, ..., x_n)$ one needs to expand the function in a Taylor series about the mean values $(x_1, x_2, ..., x_n)$, form the appropriate powers of $x_i - x_i$ by the multinomial theorem and to take expectations hoping that as many terms as possible will be zero because they contain a first power of $x_i - x_i$. From such an expansion we compute the first and second moment of the function which gives us the expected value and the variance for the same function. Since the distribution is unknown we bound the probabilities that the function lies within a certain range using Chebyshev's inequality⁽⁶²⁾.

5.8.1 Low order moments formulae

The first- and second-order terms of the Taylor series expansion of a function $f(x_1, x_2, ..., x_n)$ yield

$$f(\Delta x_{1}, \Delta x_{2}, \dots, \Delta x_{n}) = a + \sum_{\substack{i=1 \\ i=1}}^{n} b_{i} \Delta x_{i} + \frac{1/2}{\sum_{\substack{i=1 \\ i=1}}^{n} c_{i} \Delta x_{i}^{2}} + \sum_{\substack{i=1 \\ j=1 \\ i \neq j}}^{n} \sum_{\substack{j=1 \\ i\neq j}}^{n} d_{ij} \Delta x_{i} \Delta x_{j}$$
(5.49)

where $\Delta x_i^n = (x_i - \delta_i)^n$, δ_i is the expected value $E(x_i)$ and the constant term a is the function evaluated at the best value of the variables. The coefficients b_i, c_i and d_{ij} are the partial derivatives $(\partial f/\partial x_i)$, $(\partial^2 f/\partial x_i^2)$ and $(\partial^2 f/\partial x_i \partial x_j)$ respectively evaluated at the best values, $x_i = \delta_i$.

Computing the expected value or first moment E(f) from Eqn. (5.49) yields

$$E(f) = a + \frac{1}{2} \sum_{i=1}^{n} c_i \sigma_i^2 + \sum_{\substack{i=1 \\ i \neq j}}^{n} \sum_{j=1}^{n} d_{ij} cov(x_i, x_j)$$
(5.50)

The computation of the variance is much more involved and is made by computing $E(f^2)-E^2(f)$. If for any reason we agree to neglect higher than second-order moments, Var(f) can be written in general as

$$\operatorname{Var}(f) = \sum_{i=1}^{n} b_{i}^{2} \sigma_{i}^{2} + 2 \sum_{i=1}^{n} \sum_{\substack{j=1\\i\neq j}}^{n} b_{i} b_{j} \operatorname{cov}(x_{i}, x_{j})$$
(5.51)

Proper simplifications for the case under study will allow the use of Eqns. (5.50) and (5.51) efficiently to compute the mean and variance of a function of n random variables.

5.8.2 Expected value and variance

The probability and frequency of system failure given by Eqns. (5.17) to (5.19) are non-linear functions of the failure and repair rates of the components of the system. Therefore, to compute the expected value and the variance of both indices we need to make a linearization of those equations by expanding them in a truncated Taylor series.

The assumptions made in this reliability case are :

(i) component data is available in terms of their expected values and variances, (ii) components are statistically independent. Using the same notation of section 5.8.1 we have

$$cov(x_1, x_j) = 0$$

$$E(\Delta x_i \Delta x_j^n) = E(\Delta x_i) E(\Delta x_j^n) = 0$$

$$E(\Delta x_1^2 \Delta x_j^2) = \sigma_1^2 \sigma_j^2$$

(iii) component repair rates are much larger than their failure rates, then the c and d coefficients used in the Taylor series expansion are smaller than the b coefficients.

Under such assumptions the expected values and variances of the steady-state probability and frequency of system failure are given by

$$E(P_{f}) = (P_{f})_{bv} + \frac{1}{2} \sum_{i=1}^{n} c_{p\lambda_{i}} \sigma_{\lambda_{i}}^{2} + \frac{1}{2} \sum_{i=1}^{n} c_{p\mu_{i}} \sigma_{\mu_{i}}^{2}$$
(5.52)

$$\operatorname{Var}(P_{f}) = \sum_{i=1}^{n} b_{p\lambda_{i}} \sigma_{\lambda_{i}}^{2} + \sum_{i=1}^{n} b_{p\mu_{i}} \sigma_{\mu_{i}}^{2}$$
(5.53)

$$E(f_{f}) = (f_{f})_{bv} + \frac{1}{2} \sum_{i=1}^{n} c_{f_{\lambda i}} \sigma_{\lambda i}^{2} + \frac{1}{2} \sum_{i=1}^{n} c_{f_{\mu i}} \sigma_{\mu i}^{2}$$
(5.54)

$$\operatorname{Var}(f_{f}) = \sum_{i=1}^{n} b_{f_{\lambda i}} \sigma_{\lambda_{i}}^{2} + \sum_{i=1}^{n} b_{f_{\mu i}} \sigma_{\mu_{i}}^{2}$$
(5.55)

The expected value and the variance of the probability of system failure in the interval time from 0 to t_m are given by

$$E\{P_{f}(t_{m})\} = P_{f_{bv}}(t_{m}) + \frac{1}{2} \sum_{i=1}^{n} c_{Q_{i}}(t_{m}) \sigma_{Q_{i}}^{2}(t_{m})$$
(5.56)

$$\operatorname{Var}\{P_{f}(t_{m})\} = \sum_{i=1}^{n} b_{Q_{i}}(t_{m}) \sigma_{Q_{i}}^{2}(t_{m})$$
(5.57)

The linearity of both reliability indices with respect to the minimal cut-set values $Pr\{\overline{C}_i\}$ and $f\{S_i\}$ in Eqns. (5.11) and (5.13) suggest a cut-oriented procedure for the computations of the b and c coefficients of Eqns. (5.52) to (5.57).

<u>Non-repairable components</u> (t_m mission time)

First-order cut-set

Let i be the component contained in the cut set

$$b_{Q_{i}}(t_{m}) = 1$$

$$c_{Q_{i}}(t_{m}) = 0$$

$$E \{P_{fC1}\} = \hat{Q}_{i}(t_{m}) \qquad (5.58)$$

$$Var \{P_{fC1}\} = Var \{\hat{Q}_{i}(t_{m})\} \qquad (5.59)$$

Second-order cut-set

Let i and j be the components of the cut-set

$$b_{Q_{i}}(t_{m}) = \hat{Q}_{j}(t_{m})$$

$$b_{Q_{j}}(t_{m}) = \hat{Q}_{i}(t_{m})$$

$$c_{Q_{i}}(t_{m}) = c_{Q_{j}}(t_{m}) = 0$$

$$E\{P_{fC2}\} = \hat{Q}_{i}(t_{m}) \hat{Q}_{j}(t_{m}) \qquad (5.60)$$

$$\operatorname{Var}\left\{ P_{fC2} \right\} = \sum_{\substack{k=i,j\\ l=j,i}} \hat{Q}_{k}(t_{m}) \operatorname{Var}\left\{ \hat{Q}_{l}(t_{m}) \right\}$$
(5.61)

Let i, j and k be the components in the cut set

$$b_{Q_{i}}(t_{m}) = \hat{Q}_{j}(t_{m}) \hat{Q}_{k}(t_{m})$$

$$b_{Q_{j}}(t_{m}) = \hat{Q}_{i}(t_{m}) \hat{Q}_{k}(t_{m})$$

$$b_{Q_{k}}(t_{m}) = \hat{Q}_{i}(t_{m}) \hat{Q}_{j}(t_{m})$$

$$c_{Q_{i}}(t_{m}) = c_{Q_{j}}(t_{m}) = c_{Q_{k}}(t_{m}) = 0$$

$$E \{P_{fC3}\} = \hat{Q}_{i}(t_{m}) \hat{Q}_{j}(t_{m}) \hat{Q}_{k}(t_{m}) \qquad (5.62)$$

$$Var \{P_{fC3}\} = \sum_{\substack{l=j,k,i\\n=k,i,j\\p=i,j,k}} \hat{Q}_{l}(t_{m}) \hat{Q}_{n}(t_{m}) Var \{\hat{Q}_{p}(t_{m})\} \qquad (5.63)$$

Repairable components

First-order cut-set

Let i be the component in the cut

$$b_{p\lambda_{i}} = \hat{\mu}_{i} / (\hat{\lambda}_{i} + \hat{\mu}_{i})^{2}$$

$$b_{p\mu_{i}} = -\hat{\lambda}_{i} / (\hat{\lambda}_{i} + \hat{\mu}_{i})^{2}$$

$$c_{p\lambda_{i}} = -2\hat{\mu}_{i} / (\hat{\lambda}_{i} + \hat{\mu}_{i})^{3}$$

$$c_{p\mu_{i}} = 2\hat{\lambda}_{i} / (\hat{\lambda}_{i} + \hat{\mu}_{i})^{3}$$

$$b_{f\lambda_{i}} = \hat{\mu}_{i}^{2} / (\hat{\lambda}_{i} + \hat{\mu}_{i})^{2}$$

$$b_{f\mu_{i}} = \hat{\lambda}_{i}^{2} / (\hat{\lambda}_{i} + \hat{\mu}_{i})^{2}$$

$$c_{f\lambda_{i}} = -2\hat{\mu}_{i}^{2} / (\hat{\lambda}_{i} + \hat{\mu}_{i})^{3}$$

$$c_{f\mu_{i}} = -2\hat{\lambda}_{i}^{2} / (\hat{\lambda}_{i} + \hat{\mu}_{i})^{3}$$

Second-order cut-set

Let i and j be the components in the cut-set

$$\begin{split} \mathbf{b}_{p\lambda_{i}} &= (\hat{\lambda}_{j}/\hat{\lambda}_{j} + \hat{\mu}_{j}) \cdot \hat{\mu}_{i}/(\hat{\lambda}_{i} + \hat{\mu}_{i})^{2} \\ \mathbf{b}_{p\mu_{i}} &= -(\hat{\lambda}_{j}/\hat{\lambda}_{j} + \hat{\mu}_{j}) \cdot \hat{\lambda}_{i}/(\hat{\lambda}_{i} + \hat{\mu}_{i})^{2} \\ \mathbf{c}_{p\lambda_{i}} &= -2 \cdot (\hat{\lambda}_{j}/\hat{\lambda}_{j} + \hat{\mu}_{j}) \cdot \hat{\mu}_{i}/(\hat{\lambda}_{i} + \hat{\mu}_{i})^{3} \\ \mathbf{c}_{p\mu_{i}} &= 2 \cdot (\hat{\lambda}_{j}/\hat{\lambda}_{j} + \hat{\mu}_{j}) \cdot \hat{\lambda}_{i}/(\hat{\lambda}_{i} + \hat{\mu}_{i})^{3} \\ \mathbf{b}_{f\lambda_{i}} &= (\hat{\lambda}_{j}/\hat{\lambda}_{j} + \hat{\mu}_{j}) \cdot \hat{\mu}_{i}/(\hat{\lambda}_{i} + \hat{\mu}_{i})^{2} \cdot (\hat{\mu}_{i} + \hat{\mu}_{j}) \quad (5.65) \\ \mathbf{b}_{f\mu_{i}} &= (\hat{\lambda}_{j}/\hat{\lambda}_{j} + \hat{\mu}_{j}) \cdot \hat{\lambda}_{i}/(\hat{\lambda}_{i} + \hat{\mu}_{i})^{2} \cdot (\hat{\lambda}_{i} - \hat{\mu}_{j}) \\ \mathbf{c}_{f\lambda_{i}} &= -2 \cdot (\hat{\lambda}_{j}/\hat{\lambda}_{j} + \hat{\mu}_{j}) \cdot \hat{\mu}_{i}/(\hat{\lambda}_{i} + \hat{\mu}_{i})^{3} \cdot (\hat{\mu}_{i} + \hat{\mu}_{j}) \\ \mathbf{c}_{f\mu_{i}} &= -2 \cdot (\hat{\lambda}_{j}/\hat{\lambda}_{j} + \hat{\mu}_{j}) \cdot \hat{\lambda}_{i}/(\hat{\lambda}_{i} + \hat{\mu}_{i})^{3} \cdot (\hat{\lambda}_{i} - \hat{\mu}_{j}) \end{split}$$

Similar expressions can be used for component j interchanging i by j in the above formulae.

Third-order cut-set

Let i, j and k be the components in the cut-set

$$\begin{split} \mathbf{b}_{p\lambda_{i}} &= (\hat{\lambda}_{j}/\hat{\lambda}_{j} + \hat{\mu}_{j}) \cdot (\hat{\lambda}_{k}/\hat{\lambda}_{k} + \hat{\mu}_{k}) \cdot \hat{\mu}_{i}/(\hat{\lambda}_{i} + \hat{\mu}_{i})^{2} \\ \mathbf{b}_{p\mu_{i}} &= -(\hat{\lambda}_{j}/\hat{\lambda}_{j} + \hat{\mu}_{j}) \cdot (\hat{\lambda}_{k}/\hat{\lambda}_{k} + \hat{\mu}_{k}) \cdot \hat{\lambda}_{i}/(\hat{\lambda}_{i} + \hat{\mu}_{i})^{2} \\ \mathbf{c}_{p\lambda_{i}} &= -2 \cdot (\hat{\lambda}_{j}/\hat{\lambda}_{j} + \hat{\mu}_{j}) \cdot (\hat{\lambda}_{k}/\hat{\lambda}_{k} + \hat{\mu}_{k}) \cdot \mu_{i}/(\hat{\lambda}_{i} + \hat{\mu}_{i})^{3} \\ \mathbf{c}_{p\mu_{i}} &= 2 \cdot (\hat{\lambda}_{j}/\hat{\lambda}_{j} + \hat{\mu}_{j}) \cdot (\hat{\lambda}_{k}/\hat{\lambda}_{k} + \hat{\mu}_{k}) \cdot \hat{\lambda}_{i}/(\hat{\lambda}_{i} + \hat{\mu}_{i})^{3} \\ \mathbf{b}_{f\lambda_{i}} &= (\hat{\lambda}_{j}/\hat{\lambda}_{j} + \hat{\mu}_{j}) \cdot (\hat{\lambda}_{k}/\hat{\lambda}_{k} + \hat{\mu}_{k}) \cdot \hat{\mu}_{i}/(\hat{\lambda}_{i} + \hat{\mu}_{i})^{2} \cdot (\hat{\mu}_{i} + \hat{\mu}_{j} + \hat{\mu}_{k}) \\ \mathbf{b}_{f\lambda_{i}} &= (\hat{\lambda}_{j}/\hat{\lambda}_{j} + \hat{\mu}_{j}) \cdot (\hat{\lambda}_{k}/\hat{\lambda}_{k} + \hat{\mu}_{k}) \cdot \hat{\mu}_{i}/(\hat{\lambda}_{i} + \hat{\mu}_{i})^{2} \cdot (\hat{\lambda}_{i} - \hat{\mu}_{j} + \hat{\mu}_{k}) \\ \mathbf{b}_{f\mu_{i}} &= (\hat{\lambda}_{j}/\hat{\lambda}_{j} + \hat{\mu}_{j}) \cdot (\hat{\lambda}_{k}/\hat{\lambda}_{k} + \hat{\mu}_{k}) \cdot \hat{\lambda}_{i}/(\hat{\lambda}_{i} + \hat{\mu}_{i})^{2} \cdot (\hat{\lambda}_{i} - \hat{\mu}_{j} + \hat{\mu}_{k}) \\ \mathbf{b}_{f\mu_{i}} &= (\hat{\lambda}_{j}/\hat{\lambda}_{j} + \hat{\mu}_{j}) \cdot (\hat{\lambda}_{k}/\hat{\lambda}_{k} + \hat{\mu}_{k}) \cdot \hat{\lambda}_{i}/(\hat{\lambda}_{i} + \hat{\mu}_{i})^{2} \cdot (\hat{\lambda}_{i} - \hat{\mu}_{j} + \hat{\mu}_{k}) \\ \mathbf{b}_{f\mu_{i}} &= (\hat{\lambda}_{j}/\hat{\lambda}_{j} + \hat{\mu}_{j}) \cdot (\hat{\lambda}_{k}/\hat{\lambda}_{k} + \hat{\mu}_{k}) \cdot \hat{\lambda}_{i}/(\hat{\lambda}_{i} + \hat{\mu}_{i})^{2} \cdot (\hat{\lambda}_{i} - \hat{\mu}_{j} + \hat{\mu}_{k}) \\ \mathbf{b}_{f\mu_{i}} &= (\hat{\lambda}_{j}/\hat{\lambda}_{j} + \hat{\mu}_{j}) \cdot (\hat{\lambda}_{k}/\hat{\lambda}_{k} + \hat{\mu}_{k}) \cdot \hat{\mu}_{i}/(\hat{\lambda}_{i} + \hat{\mu}_{i})^{2} \cdot (\hat{\lambda}_{i} - \hat{\mu}_{j} + \hat{\mu}_{k}) \\ \mathbf{b}_{f\mu_{i}} &= (\hat{\lambda}_{j}/\hat{\lambda}_{j} + \hat{\mu}_{j}) \cdot (\hat{\lambda}_{k}/\hat{\lambda}_{k} + \hat{\mu}_{k}) \cdot \hat{\mu}_{i}/(\hat{\lambda}_{i} + \hat{\mu}_{i})^{3} \cdot (\hat{\mu}_{i} + \hat{\mu}_{j} + \hat{\mu}_{k}) \\ \mathbf{b}_{f\mu_{i}} &= -2(\hat{\lambda}_{j}/\hat{\lambda}_{j} + \hat{\mu}_{j}) \cdot (\hat{\lambda}_{k}/\hat{\lambda}_{k} + \hat{\mu}_{k}) \cdot \hat{\lambda}_{i}/(\hat{\lambda}_{i} + \hat{\mu}_{i})^{3} \cdot (\hat{\lambda}_{i} - \hat{\mu}_{j} - \hat{\mu}_{k}) \\ \mathbf{b}_{i} \end{pmatrix} \\ \mathbf$$

Similar expressions can be used for components j and k interchanging i by j or k respectively.

5.8.3 Confidence limits

The expected value E(x) and the variance Var(x) of a random variable x can be computed knowing the probability distribution of the random variable. However, the converse is not true, i.e., from a knowledge of E(x) and Var(x) it is not possible to reconstruct the probability distribution of x and hence quantities such as

 $\Pr\left\{ \left| x - E(x) \right| \leqslant c \right\}$

cannot be calculated. Nonetheless it is possible to give very useful bounds to such probabilities using what is known as Chebyshev's inequality.

If x is a random variable with mean $E(x) = {}^{x}_{x}$ and standard deviation σ_{y} , then Chebyshev's theorem says

 $\Pr\left\{\left|x-\vartheta_{x}\right| \leq k \sigma_{x}\right\} \geqslant 1 - 1/k^{2}$ (5.67)

This form of inequality is particularly indicative of how the variance measures the degree of concentration of probability near $E(x) = \bigvee_{x}$. When deriving such an inequality no distributional assumption about x is made and it is still possible to place a bound on the probability that x is within k standard deviations of its mean.

Additional information about the distribution of the random variable x will enable us to improve on the inequality presented. However probability transformations lead to considerable difficulty and complication, each particular situation depends on the details of the transformation and the distributions involved. Shooman⁽¹⁸⁾ presents a good summary for one or more models of probability distributions which transform in a simple and well-known manner when subject to addition,

reciprocation, multiplication and division. Unfortunately few models work well under certain of these transformations and fall down in one or two operations so that their use is not general. In many problems, like our reliability case, no assumptions concerning the specific distribution of the reliability indices is justified, and in such case Chebyshev's inequality can give us important information about the behaviour of the random indices. A minor nuisance is that this inequality is the wrong way round, one chooses the probability and then find the k value.

5.8.4 Tests

The main purpose of the tests performed was to investigate for the minimal cut-set approach to system reliability and the performance of the deterministic and statistical methods the effect of :

- (i) component availability,
- (ii) variance of component parameters,
- (iii) sample size of earlier testing.

The reliability network of Fig. 1 was used to test both approaches for the repairable and non-repairable cases. The components were assumed to be all equal and estimates and variances of the components parameters were available. If the accuracy of the estimates is given in terms of a definite tolerance, table II from [18] can be used to calculate an estimated value of the respective variance. Only the minimal cutsets between nodes 1 and 4 were considered.

For the steady-state case the variances of the failure and repair rate of the components were computed using the minimum variance unbiased estimator for the exponential case (49). The minimum variance unbiased estimator of the reliability in the exponential case (54) was used to

analyse the performance of the approaches for the probability of failure for a given mission time. The variance of this parameter was calculated using the Gauss-Laguerre quadrature method as employed for the integral of Eqn. (5.9).





TABLE II

RELATIONSHIP BETWEEN TOLERANCE AND STANDARD DEVIATION

No.	Type of component and manufacturer	Relationship
1	High-quality component in mature develop-	E(x) = N
	manufacturer with much experience	$3\sigma_x = TN$
2	Average situation somewhere between 1 and 3	E(x) = N $2\sigma_x = TN$
3	Ordinary comercial component, early development stage, manufacturer little known or has poor reputation and little experience	E(x) = N $\sigma_x = TN$

N = nominal value, σ_x = standard deviation, T = tolerance in per unit

The assumption of the exponential case in the steady-state was needed for the purpose to calculate the variances from the MVU estimator for this probability distribution. As $\operatorname{Singh}^{(63)}$ has shown the form of the probability distribution of the up and down times do not affect the steady-state indices when the components are independent.

Test 1

The objective of this test was to evaluate the performance of the deterministic and statistical approaches for different component availabilities which were varied by selecting failure rates between 2 and 219 f/year for a repair rate of 438 repairs/year. The variances of the failure and repair rate were calculated assuming a coefficient of dispersion for both of 5%.

Tables III and IV present the results for the deterministic approach for the probability and frequency of system failure respectively. It is apparent from these tables that the range defined by the difference between the maximum and minimum values improves as the component availability increases. It can be observed that the exact values for both indices are always inside the range but for highly unreliable components, $\lambda = 219$ f/year.

The results for the approximate statistical approach are presented in tables V to VIII. The performance of the approach compared with the single-point estimate is extremely good and has the advantage of providing information about the variance of the indices. The results of tables VII and VIII giving 95% confidence limits for both reliability indices show that the use of Chebyshev's inequality gives lower and upper limits which always enclose the best value for the high-reliability region. In the low reliability region the inequality will give only an upper limit which in the case of the probability and frequency of system failure is all what is needed.

5.8.5 Test 2

This test was performed to investigate the effect of the dispersion of component reliability data on the minimal cut-set approach to system reliability and the performance of the deterministic and statistical approaches in the steady-state case. Numerical evaluations were made for a failure rate of 4 f/year and repair rate of 438 repairs/ year varying the coefficient of dispersion of both rates from 1 to 10%.

The results obtained from the deterministic approach are presented in tables IX and X for the probability and frequency of failure respectively. These tables show that the range defined by the maximum and minimum values increase for higher uncertainty in the component parameters. It seems that in case of maximum drift of these parameters from its best values, the solution of the single-point estimate of the minimal cut-set approach will be very far from the extreme points even . for highly stable components.

Tables XI and XII show the results from the approximate statistical approach. These tables present the variance and coefficient of dispersion of both reliability indices for different dispersions of the component parameters. The results indicate that both variances increase as the uncertainty of the parameters increases, the incidence being higher on the probability of system failure. It is important to remark that such analysis is not possible from a single-point solution. Table XIII presents the proportional error of both reliability indices for different proportional errors of the component parameter. The results indicate that the minimal cut-set approach has very low sensitivity to errors on the data and do not amplify such errors.

Table XIV presents 95% confidence limits for different coefficients of dispersion of the component data. It can be observed that the exact value calculated using the best value of the parameters is always inside the limits. It is also clear that the interval increases as the uncertainties of the parameters increase.

5.8.6 Test 3

The effect of the sample size of earlier testing on the performance of both approaches for the steady-state case was investigated within the high-reliability region. A failure rate of 2 f/year and a repair rate of 438 repairs/year were used for the tests.

The results from table XV show that the range provided by the deterministic approach improves as the sample size increases. The incidence of the sample size is greater in the frequency than in the probability of failure. For small samples the deterministic approach only provides maximum value.

The results in tables XVI and XVII indicate that the variance of both reliability indices is reduced as the sample size increases. These results are to be expected because the higher the sample size the smaller the variance of each component parameter. Table XVIII presents 95% confidence limits for both indices. The results show that for small samples the approach provides an upper limit which is reduced as the sample size increases. The incidence of the sample size for the lower limit is greater in the probability of system failure than in the frequency of system failure.

5.8.7 Test 4

The purpose of this test was to investigate the effect of the sample size on the best value of the reliability given by Eqn. (5.8) and its variance obtained by subtracting $\exp(-2t_m/\text{MTTF})$ from Eqn. (5.9).

At the same time an evaluation was made of the performance of the deterministic and statistical approaches for the non-repairable case. The MTTF of the components used was 4380 h.

Tables XIX and XX show the minimum and maximum values obtained from the deterministic solution for two relations of the mission time to the component mean time to failure of 0.50 and 0.05. These two cases give the possibility to analyze the performance in both reliability regions, low and high. In both cases the "best value" obtained from the exact solution is within the range defined by the minimum and maximum values of the deterministic approach. It is clear that for a given t_m /MTTF relationship, the greater the sample size the smaller the reliability, the effect being higher for mission times greater than 50% of the MTTF. The range provided for the deterministic solution is narrowest in the high reliability region.

Tables XXI and XXII present the same results but from the statistical approach. It is also clear that increasing the mission time reduces the system reliability for a definite MTTF of the components. It is interesting to note that the dispersion of the system reliability is extremely high for the low reliability region. This result agreed with the well-known statement that the minimal cut-set approach is good in the high reliability region. It is also observed that the dispersion of the system reliability in the high reliability region is smaller than the dispersion of a single component i.e., the minimal cut-set formulae do not amplify the uncertainty in the component estimates. For both reliability regions increasing the sample size reduces the dispersion of the system reliability estimate.

Table XXIII shows that in the high reliability region the uncertainties in the components data do not affect significantly the

estimate i.e., the c-coefficients obtained in the Taylor series expansion are much smaller than the b-coefficients. It can be observed from table XXIV that the Chebyshev's inequality provides a lower limit to system reliability which for this index, system reliability, is all that is needed.

5.8.8 Test 5

The objective of the test was to investigate the effect of different component mean time to failure and its uncertainty in the system reliability and the performance of the deterministic and statistical approaches in such conditions. The tests were performed for a mission time of 100 hrs., assuming a sample size of 32.

Table XXV indicates that for different component MTTF the best value obtained from the exact solution is always within the range defined from the deterministic solution. The results of Table XXVI show that the uncertainties of component data do not affect significantly the estimate of the system reliability compared with the singlepoint estimate. Such estimates improve for highly reliable components, and also the variances are reduced for an increasing component mean time to failure. Table XXVII shows that the greater the component mean time to failure the smaller the dispersion of the system reliability estimate. Table XXVIII presents a 95% lower confidence limit for the system reliability which improves for an increasing mean time to failure.

5.8.9 Test 6

This test was performed to analyse the effect of component uncertainties in the system reliability for different relations between the mission time to the component mean time to failure using the results given from the deterministic and statistical approaches. The parameters used were a component MTTF of 4380 h. and a sample size of 32.

The results of table XXIX indicate that the "best values" given by a single-point estimation from the exact solution can have more than 16% of deviation from the expected value when the variance of the component parameter estimates is taken into account for a mission time equal or greater than 50% of the component mean time to failure. The solution from the statistical approach allows us to appreciate such effects because the variance of the estimate increases under such conditions. The range defined from the deterministic approach and shown in table XXX always encloses the best value when in the high reliability region. The variance of the estimate is reduced for shorter mission times and also the dispersion of the estimate as shown in table XXXI. Table XXXII shows that Chebyshev's inequality gives important information about the system reliability for the high reliability region.

TABLE III

DETERMINISTIC SOLUTION FOR DIFFERENT COMPONENT AVAILABILITY

PROBABILITY OF FAILURE Component repair rate = 438 repairs/year Component dispersion coefficient = 5%

Failure	Component	Single	Determinis	tic range
f/year	availability	estimate	Minimum	Maximum
2	0.995455	0.415×10^{-4}	0.340×10^{-4}	0.507×10^{-4}
4	0•990959	0.165×10^{-3}	0.135×10^{-3}	0.202×10^{-4}
8	0.982063	0.654×10^{-3}	0.537×10^{-3}	0.799×10^{-4}
16	0.964758	0.256 x 10 ⁻²	0.211×10^{-2}	0.313×10^{-2}
32	0.931915	0,980 x 10 ⁻²	0.817×10^{-2}	1.20×10^{-2}
219	0.666667	0.243	0.254	0.344

TABLE IV

DETERMINISTIC SOLUTION FOR DIFFERENT COMPONENT AVAILABILITY FREQUENCY OF FAILURE

Component repair rate = 438 repairs/year Component dispersion coefficient = 5%

Failure rate	Component availability	Single point	Deterministic range	
f/year		estimate	Minimum	Maximum
2	0•995455	0.0364	0.031	0.042
4	0.990959	0.145	0.124	0.167
8	0.982063	0•578	0•537	0 •799
16	0.964758	2.28	1.95	2.63
32	0.931915	8.77	7.65	10.25
219	0.666667	202	261	323
			I	

TABLE V

APPROXIMATE STATISTICAL STEADY-STATE SOLUTION FOR DIFFERENT COMPONENT AVAILABILITY

PROBABILITY OF FAILURE

Failure rate f/year	Component availability	Single point estimate	Statistic Estimate	cal solution
2	0.995455	0.415×10^{-4}	0.417×10^{-4}	0.846×10^{-11}
4	0.990959	0.165×10^{-3}	0.166 x 10^{-3}	0.132 x 10 ⁻⁹
8	0.982063	0.654×10^{-3}	0.658×10^{-3}	0.200×10^{-8}
16	0.964758	0.256 x 10 ⁻²	0.258×10^{-2}	0.288×10^{-7}
32	0.931915	0.980×10^{-2}	0.994×10^{-2}	0.376 x 10 ⁻⁶
219	0.666667	0.243	0.297	0.128×10^{-3}

TABLE VI

APPROXIMATE STATISTICAL STEADY-STATE SOLUTION FOR DIFFERENT COMPONENT AVAILABILITY

Failure rate	Component availability	Single point	Statistical	solution
f/year		estimate	Estimate	Variance
2	0.995455	0.0364	0.0365	0.405×10^{-5}
4	0.990959	0.145	0.146	0.630×10^{-4}
8	0.982063	0.578	0.580	0.952×10^{-3}
16	0.964758	2.28	2.30	0.136×10^{-1}
32	0.931915	8.77	8.97	0.177
219	0.666667	202	292	64.5

FREQUENCY OF FAILURE

TABLE VII

95% CONFIDENCE LIMITS FOR THE PROBABILITY OF SYSTEM FAILURE

Failure rate	Component availability	Single point	Confidence	limits
f/year		estimate	Lower	Upper
2	0.995455	0.415×10^{-4}	0.287×10^{-4}	0.547×10^{-4}
4	0.990959	0.165×10^{-3}	0.115×10^{-3}	0.217×10^{-3}
8	0.982063	0.654×10^{-3}	0.458×10^{-3}	0.858×10^{-3}
16	0.964758	0.256×10^{-2}	0.182×10^{-2}	0.334×10^{-2}
32	0.931915	0.980×10^{-2}	0.720×10^{-2}	0.127×10^{-1}
219	0.666667	0.243		0.347
			<u> </u>	

TABLE VIII

Failure rate	Component availability	Single point	Confidence 2	limits
f/year		estimate	Lower	Upper
2	0.995455	0.0364	0.0275	0.0455
4	0.990959	0.145	0.1103	0.1813
8	0.982063	0.578	0.4422	0.7182
16	0.964758	2.28	1.774	2.818
32	0.931915	8.77	7.086	10.85
219	0.666667	20 2	-	327.8

95% CONFIDENCE LIMITS FOR THE FREQUENCY OF SYSTEM FAILURES

TABLE IX

DETERMINISTIC STEADY-STATE SOLUTION FOR DIFFERENT DISPERSION COEFFICIENTS OF COMPONENT PARAMETERS

Component parameter	Failures/year		
of dispersion	Minimum	Maximum .	
1	0.159×10^{-3}	0.172×10^{-3}	
2	0.153×10^{-3}	0.179×10^{-3}	
3	0.146×10^{-3}	0.186×10^{-3}	
4	0.141×10^{-3}	0.194×10^{-3}	
5	0.135×10^{-3}	0.202×10^{-3}	
6	0.130×10^{-3}	0.210×10^{-3}	
7	0.125×10^{-3}	0.218×10^{-3}	
8	0.120×10^{-3}	0.227×10^{-3}	
9	0.115×10^{-3}	0.237×10^{-3}	
10	0.111×10^{-3}	0.246×10^{-3}	

TABLE X

DETERMINISTIC STEADY-STATE SOLUTION FOR DIFFERENT DISPERSION COEFFICIENTS OF COMPONENT PARAMETERS-FREQUENCY OF SYSTEM FAILURE

Component parameter	Encounters/year		
% coefficient of dispersion	Minimum	Maximum	
1	0.141	0.149	
2	0.137	0.154	
3	0.132	0.158	
4	0.128	0.163	
5	0.124	0.167	
6	0.119	0.172	
7	0.115	0.176	
8	0.111	0.180	
9	0.106	0.185	
10	0.102	0.189	

TABLE XI

APPROXIMATE STATISTICAL STEADY-STATE SOLUTION FOR DIFFERENT DISPERSION COEFFICIENT OF COMPONENT PARAMETERS PROBABILITY OF SYSTEM FAILURE

Component failure rate = 4 f/year Component repair rate = 438

Parameter % coefficient of dispersion	Estimate	Variance	System % coefficient of dispersion
1	0.1653×10^{-3}	0.527×10^{-11}	1.39
2	0.1654×10^{-3}	0.211×10^{-10}	2.78
3	0.1656 x 10 ⁻³	0.474×10^{-10}	4.16
4	0.1658 x 10 ⁻³	0.843 x 10 ⁻¹⁰	5.54
5	0.1660×10^{-3}	0.132×10^{-9}	6.91
6 [.]	0.1664×10^{-3}	0.189×10^{-9}	8.28
7	0.1669×10^{-3}	0.258×10^{-9}	9.63
8	0.1674 x 10 ⁻³	0.337×10^{-9}	10.97
9	0.1679×10^{-3}	3.427×10^{-9}	12.31
10	0.1685 x 10 ⁻³	0.527 x 10 ⁻⁹	13.62

TABLE XII

APPROXIMATE STATISTICAL STEADY-STATE SOLUTION FOR DIFFERENT DISPERSION COEFFICIENT OF COMPONENT PARAMETERS

FREQUENCY OF SYSTEM FAILURE

Component failure rate = 4 f/year Component repair rate = 438 repairs/year

Parameter % coefficient of dispersion	Estimate	Variance	System % coefficient of dispersion
1	0.1454	0.25×10^{-5}	1.09
2	0.1455	0.10×10^{-4}	2.18
3	0.1456	0.23×10^{-4}	3.27
4	0.1457	0.40×10^{-4}	4.36
5	0.1458	0.63×10^{-4}	5.44
6	0.1459	0.91×10^{-4}	6.52
7	0.1461	0.12×10^{-3}	7.60
8	0.1463	0.16×10^{-3}	8.68
9	0.1466	0.20×10^{-3}	9•75
10	0.1468	0.25×10^{-3}	10.81

TABLE XIII

PROPORTIONAL ERROR IN THE PROBABILITY AND FREQUENCY OF SYSTEM FAILURE FOR DIFFERENT PROPORTIONAL ERROR IN THE COMPONENT PAPAMETERS

Component % proportional error	Probability of fail- ure % proportional error	Frequency of failure % proportional error
1	0.18	0.28
2	0.24	. 0.34
3	0.36	0.41
4	0.48	0.48
· 5	0.60	0.55
6	0.84	0.62
7	1.14	0.75
8	1.43	0.89
9	1.73	1.09
10	2.08	1.23
1		•
TABLE XIV

95% CONFIDENCE LIMITS

PROBABILITY AND FREQUENCY OF SYSTEM FAILURE

Parameter % coefficient	Probability of	Frequency of failure		
of dispersion	Lower	Upper	Lower	Upper
l	0.155×10^{-3}	0.176×10^{-3}	0.138	0.153
2	0.145×10^{-3}	0.185×10^{-3}	0.131	0.160
3	0.135×10^{-3}	0.196×10^{-3}	0.124	0.167
4	0.125×10^{-3}	0.207×10^{-3}	0.117	0.174
5	0.115×10^{-3}	0.217×10^{-3}	0.110	0.181
6	0.105×10^{-3}	0.228×10^{-3}	0.103	0.189
7	0.950×10^{-4}	0.239×10^{-3}	0.096	0.196
8	0.852×10^{-4}	0.250×10^{-3}	0.090	0.203
9	0.755×10^{-4}	0.260×10^{-2}	0.083	0.210
10	0.659×10^{-4}	0.271×10^{-3}	0.076	0.218

TABLE XV

DETERMINISTIC STEADY-STATE SOLUTION FOR DIFFERENT SAMPLE SIZES

Component failure rate = 2 f/year Component repair rate = 438 repairs/year Node 4 in Fig. 1

Sample	Probabi l ity o	f failure	Frequency of failure	
size	Minimum	Maximum	Minimum	Maximum
3	0.12×10^{-5}	0.14×10^{-2}	-	0.114
5	0.46×10^{-5}	0.37×10^{-3}	-	0.091
9	0.95×10^{-5}	0.18×10^{-3}	-	0.075
17	0.15×10^{-4}	0.11×10^{-3}	0.009	0.063
33	0.20×10^{-4}	0.85×10^{-4}	0.017	0.056
65	0.25×10^{-4}	0.68×10^{-4}	0.023	0.050
129	0.29×10^{-4}	0.59×10^{-4}	0.027	0.046
257	0.32×10^{-4}	0.53×10^{-4}	0.030	0.043

TABLE XVI

STATISTICAL STEADY-STATE SOLUTION FOR DIFFERENT SAMPLE SIZES

Sample	Parameters % coefficient	Frobability of failure			
size	of dispersion	Estimate	Variance	% dispersion	
3	70.7	0.825×10^{-4}	0.169×10^{-8}	49.8	
5	50.0	0.620×10^{-4}	0.847×10^{-9}	46.9	
9	35•3	0.518×10^{-4}	0.423×10^{-9}	39•7	
17	25.0	0.466×10^{-4}	0.212×10^{-9}	31.2	
33	17.7	0.441×10^{-4}	0.106×10^{-9}	23.3	
65	12.5	0.428×10^{-4}	0.529×10^{-10}	16.7	
129	8.8	0.422×10^{-4}	0.264×10^{-10}	12.2	

. Component failure rate = 2 f/year Component repair rate = 438 repairs/year

TABLE XVII

STATISTICAL STEADY-STATE SOLUTION FOR DIFFERENT SAMPLE SIZES

FREQUENCY OF FAILURE

Component failure rate = 2 f/year Component repair rate = 438 repairs/year

Sample	Parameters % coefficient	Frequency of failures			
size	of dispersion	Estimate	Variance	% dispersion	
3	70.7	0.0544	0.810×10^{-3}	52.3	
5	50.0	0.0454	0.405×10^{-3}	44.3	
9	35.3	0.0409	0.203×10^{-3}	34.8	
17	25.0	0.0387	0.101×10^{-3}	26.0	
33	17.7	0.0376	0.506×10^{-4}	18.9	
65	12.5	0.0370	0.253×10^{-4}	13.6	
129	8.8	0.0367	0.127×10^{-4}	9.7	

TABLE XVIII

95% CONFIDENCE LIMITS FOR DIFFERENT SAMPLE SIZES

Sample	Probability	of failure	Frequency of failure	
size	Lower	Upper	Lower	Upper
3	-	0.267×10^{-3}	-	0.182
5	-	0.192×10^{-3}	-	0.135
9	-	0.144×10^{-3}	-	0.105
17		0.112×10^{-3}	-	0.0837
33	-	0.901×10^{-4}	0.0057	0.0694
65	0.103×10^{-4}	0.753×10^{-4}	0.0145	0.0595
129	0.192×10^{-4}	0.651 x 10 ⁻⁴	0.0208	0.0526

Component failure rate = 2 f/year Component repair rate = 438 repairs/year

TABLE XIX

DETERMINISTIC SOLUTION FOR THE NON-REPAIRABLE SYSTEM

Component MTTF = 4380.0 hrs. Mission time = 0.50 MTTF

Sample	System reliability		
size	Best value	Minimum	Maximum
2	0.8613	0.161	1.0
4	0.7617	0.238	0.943
8	0.7143	0.323	0.845
16	0.6915	0.390	0.765
32	0.6804	0.440	0.707

TABLE XX

DETERMINISTIC SOLUTION FOR THE NON-REPAIRABLE SYSTEM

Component MTTF = 4380.0 Mission time = 0.05 MTTF

Sample	System	Reliability	
size	Best value	Minimum	Maximum
2	0.9987	0.9800	1.0
4	0.9972	0.9897	0.99996
8	0.9962	0.9918	0.9989
16	0.9956	0.9928	0.9977
32	0.9953	0.9934	0.9969

TABLE XXI

STATISTICAL SOLUTION FOR THE NON-REPAIRABLE SYSTEM

Component MTTF = 4380.0 hrs Mission time = 0.50 MTTF

Sample	Component Reliability			Syste	em Reliability	
size	Estimate	Variance	% dis- persion	Estimate	Variance	% dis persion
2	0.750	0.0753	109.8	0.84375	0.10358	38.1
4	0.670	0.0300	52.5	0.71017	0.5925×10^{-1}	34.3
8	0.637	0.0132	31.6	0.63968	0.2966×10^{-1}	26.9
16	0.621	0.0062	20.7	0.60412	0.1465×10^{-1}	20.0
32	0.614	0.0030	14.1	0.58633	0.7264×10^{-1}	14.5
1					<u> </u>	

TABLE XXII

STATISTICAL SOLUTION FOR THE NON-REPAIRABLE SYSTEM

Sample	Componer	nt Reliabilit	У	System	Reliability	
size	Estimate	Variance	% dis- persion	Estimate	Variance	% dis - persion
2	0.9750000	49.7×10^{-4}	7.23	0.99872	5.16×10^{-4}	2.27
4	0.9629668	10.4×10^{-4}	3.36	0.99716	1.63×10^{-4}	1.28
8	0.9570618	3.70×10^{-4}	2.01	0.99615	6.76×10^{-5}	0.83
16	0.9541366	1.60×10^{-4}	1.33	0.99560	3.14×10^{-5}	0.56
32	0.9526808	7.51×10^{-5}	0.91	0.99530	1.50×10^{-5}	0.39

Component MTTF = 4380.0 hrs.
 Mission time = 0.05 MTTF

TABLE XXIII

COMPARISON OF THE BEST VALUE AND THE ESTIMATE FROM THE STATISTICAL APPROACH FOR DIFFERENT SAMPLE SIZES

> Component MTTF = 4380.0 hrs. Mission time = 0.05 MTTF

Sample	System Reliability		
size	Best value	Estimate	
2	0.99872	0.99872	
4	0.99716	0.99716	
8	0.99617	0.99615	
16	0.99562	0.99560	
32	0.99533	0.99530	

TABLE XXIV

95% LOWER CONFIDENCE LIMIT FOR DIFFERENT

SAMPLE SIZES

Component MTTF = 4380.0 hrs. Mission time = 0.05 MTTF

Sample size	Best value	Lower confidence limit
2	0.99872	0.8972
4	0.99716	0.9400
8	0.99617	0.9594
16	0.99562	0.9705
32	0.99533	0.9779

TABLE XXV

DETERMINISTIC SOLUTION FOR DIFFERENT MEAN TIME TO FAILURE OF COMPONENTS

> Component MTTF = variable Mission time = 100 hrs. Sample size = 32

Component	Best	Determini	stic range
MTTF	value	Minimum	Maximum
438	0.9122	0.8674	0.9356
876	0.9762	0.9658	0.9838
1095	0 . 9847	0.9780	0.9896
2190	0.9961	0•9945	0•9974
4380	0.9990	0.9986	0•9994
17520	0 •99994	0.99991	0.99996

TABLE XXVI

STATISTICAL SOLUTION FOR DIFFERENT MEAN TIME TO FAILURE OF NON-REPAIRABLE COMPONENTS

Sample size = 32 Mission time = 100 hrs.

Component	System Reliability			
MTTF	Best value	Estimate	Variance	
438	0.9122	0.90498	11.2×10^{-4}	
876	0.9762	0.97570	1.67×10^{-4}	
1095	0.9847	0.97570	8.80×10^{-5}	
2190	0.9961	0.99609	1.17×10^{-5}	
4380	0.9990	0.99902	1.50×10^{-6}	
17520	0•99994	0.99994	2.39×10^{-8}	

TABLE XXVII

COMPARISON OF THE DISPERSION COEFFICIENTS OF THE COMPONENT AND SYSTEM RELIABILITIES FOR DIFFERENT MEAN TIME TO FAILURE OF NON-REPAIRABLE COMPONENTS

> Sample size = 32 Mission time = 100 hrs.

Component MTTF	Component % coefficient of dispersion	System % coefficient of dispersion
438	3.66	3.67
876	1.86	1.30
1095	1.46	0.95
2190	0.83	0.34
4380	0.42	0.12
17520	0.11	0.015

TABLE XXVIII

95% CONFIDENCE LIMIT FOR DIFFERENT MEAN TIME TO FAILURE OF NON-REPAIRABLE COMPONENTS

> Mission time = 100 hrs. Sample size = 32

Component MTTF	Best value	Lower limit
438	0.9122	0.75521
876	0.9762	0.91798
1095	0.9847	0.94245
2190	0.9961	0.98081
4380	0.9990	0.99355
17520	0.99994	0.99925

TABLE XXIX

EFFECTS OF COMPONENT DATA UNCERTAINTY IN THE SYSTEM RELIABILITY FOR DIFFERENT RELATION OF MISSION TIME TO MEAN TIME TO FAILURE OF NON-REPAIRABLE COMPONENTS

> Component MTTF = 4380.0 hrs. Sample size = 32

t _m /MTTF	Best value	Estimate	Variance
0.01	0.99981	0.99981	3.30×10^{-6}
0.05	0.99533	0 .99531	7.50×10^{-5}
0.10	0.98167	0.98132	1.14×10^{-4}
0.20	0.93144	0 .92658	7.89×10^{-4}
0.30	0.85775	0.83927	2.26×10^{-3}
0.40	0.77147	0.72413	4.48×10^{-3}
0.50	0.68043	0.58633	7.26×10^{-3}
0.60	0.59067	0.43103	1.03×10^{-2}
0.70	0.50611	0.2631	1.34×10^{-2}

TABLE XXX

DETERMINISTIC SOLUTION FOR DIFFERENT RELATION OF MISSION TIME TO MEAN TIME TO FAILURE OF NON-REPAIRABLE COMPONENTS

> Component mean time to failure = 4380.0 hrs. Sample size = 32

t _m /MTTF	Best value	Minimum	Maximum
0.01	0.999981	0.99973	0.99988
0.05	0.99533	0.99338	0.99690
0.10	0.98617	0.97368	0.98757
0.20	0.93144	0.89728	0.95048
0.30	0.85775	0.77728	0.88990
0.40	0.77147	0.62197	0.80787
0.50	0.68043	0.44001	0.70694
0.60	0.59067	0.23952	-
0.70	0.50611	0.27765	-

TABLE XXXI

DISPERSION COEFFICIENTS OF COMPONENT AND SYSTEM RELIABILITY OF DIFFERENT RELATIONS OF MISSION TIME AND MEAN TIME TO FAILURE Component MTTF = 4380.0 hrs

Sample size = 32

t _m /MTTF .	Component % coefficient of dispersion	System % coefficient of dispersion
0.01	0.18	0.036
0.05	0.91	0.39
0.10	1.57	1.06
0.20	3.60	3.03
0.30	5.38	5.66
0.40	7.14	9.25
0.50	8.89	14.54
0.60	10.63	23.58
0.70	12.36	44.03

TABLE XXXII

95% CONFIDENCE LIMITS OF SYSTEM RELIABILITY FOR DIFFERENT RELATIONS OF MISSION TIME TO MEAN TIME TO FAILURE OF NON-REPAIRABLE COMPONENTS

> Component MTTF = 4380.0 hrs. Sample size = 32

t _m /MTTF	Best value	Lower limit	Upper limit
0.01	0.99981	0.99821	1.0
0.05	0.99533	0.97786	1.0
0.10	0.98167	0.93353	1.0
0.20	0.93144	0.80100	1.0
0.30	0.85775	0.62676	1.0
0.40	0.77147	0.42466	1.0
0.50	0.68043	0.20516	1.0
0.60	0.59067	0.0	0.88559
0.70	0.50611	0.0	0.78116
1.00	0.30079	0.0	0.37026

5.9 Approximate confidence limits using Monte Carlo simulation

In the analytical method, the technique of combination of distribution functions is used to derive the sampling distribution of the system reliability indices from the sampling distributions of the component reliability estimates; confidence limits are then obtained from this sampling distribution.

For the Monte Carlo method, the numerical sampling distributions of component reliability estimates are generated. Sample values selected from these distributions are substituted into the mathematical relationship between system and components. This simulation is repeated until the numerical sampling distribution of the system reliability indices are generated. Confidence limits are then computed on this numerical distribution.

5.9.1 Non-repairable systems

The approach described here is similar to that of $\begin{bmatrix} 60 \end{bmatrix}$, except that in this case the system is represented by its minimal cutsets and $\begin{bmatrix} 60 \end{bmatrix}$ use the series and parallel reliability formulae to obtain the system reliability.

The probability of failure Q at time t is usually expressed as

$$Q(t) = \int_{0}^{t} f(t;\theta) dt$$

where $f(t;\theta)$ is the failure density model with parameter θ . If the parameter of the model is unknown and must be estimated from component test data, then the probability of failure must be expressed as an estimate $\hat{Q}(t)$. Such a model parameter has been extensively estimated by the method of maximum likelihood. The results of such estimations summarized in [60] are presented here for the exponential case, where t_{in} are the i-th order statistics from a sample of size n.

If the exponential density is written as

$$f(t;\theta) = 1/\theta \exp(-t/\theta) \quad \theta > 0, t \ge 0$$
$$= 0 \qquad \text{elsewhere}$$

Then the maximum likelihood estimator $\hat{\theta}$, for the mean time between failures θ , from a complete sample or censored after r failures is given by

$$\hat{\theta}_{r,n} = \left\{ \sum_{i=1}^{r} t_{in} + (n-r)t_{rn} \right\} / r$$
 (5.68)

where the numerator of Eqn. (5.68) is the total accumulated operating time T of the components under test. Further $\hat{\theta}_{r,n}$ is unbiased and $2r\hat{\theta}_{r,n}/\theta$ is distributed as chi-square with 2r degrees of freedom⁽⁴⁶⁾. For a two-sided confidence level $(1 - \alpha)$ we can write confidence limits making use of the $\alpha/2$ and $1-\alpha/2$ percentage points of the chi-square distribution

$$\frac{2 r \hat{\theta}}{\chi^{2}_{\alpha/2,2r}} \leqslant \theta \leqslant \frac{2 r \hat{\theta}}{\chi^{2}_{1-\alpha/2,2r}}$$
(5.69)

Since the parameter θ of the exponential failure model is unknown and must be estimated from component test data, then the probability of failure must be expressed as an estimate $\hat{Q}(t)$. By the invariant property of ML estimator of parameters⁽⁶⁴⁾, if a random value is taken from the distribution of the estimator for the parameter θ and this is substituted into the expression for $Q(t_m)$ in the exponential case, one obtains a sample probability of failure given by

$$\hat{Q}(t_m) = 1 - \exp(-t_m \chi^2_{2r} / 2r \hat{\theta})$$
 (5.70)

The Monte Carlo computer technique for the reliability of a non-repairable system represented by its minimal cut-sets involves use of a digital computer to obtain a large number of sample probability of failure values for each component from which one obtains a numerical sampling distribution for the system reliability $R_s(t_m)$ through the mathematical expression

$$R_{s}(t_{m}) = 1 - \sum_{i=1}^{NMC} Pr\{\overline{C}_{i}\} = 1 - \sum_{i=1}^{NMC} \prod_{j \in i} Q_{j}(t_{m}) (5.71)$$

If m(Monte Carlo sample size) values are obtained for $R_s(t_m)$ by the previous synthetic sampling and arranged in order of increasing magnitude, each value will represent a 1/m increase in the step cumulative distribution function. If m is taken as large as, say 1000 then the 99.% confidence limits for $Q_s(t_m)$ will be the largest and smallest

values of the ordered numerical distribution. In general, the twosided lower confidence limit is

$$R_{SL}(t_m) = R_S(i) \quad i = m \cdot \alpha/2$$
 (5.72)

and the upper confidence limit is

$$R_{SU}(t_m) = R_S(j) \quad j = m \cdot (1-\alpha/2)$$
 (5.73)

where m is the Monte Carlo sample size and α is the level of significance. When it is required that the system reliability does not go below a minimum value we use the procedure of one-sided confidence limit and Eqn. (5.72) corresponds to a lower one-sided confidence limit at $100(1-\alpha/2)$ confidence level.

5.9.2 Algorithm and an illustrative system

The main steps for the Monte Carlo simulation procedure can be summarised as follows:

- Step 1 : Generate the minimal cut-set matrix of the system.
- Step 2 : Generate component sample probability of failure values for the given mission time and sample size, using a random number generator from a chi-square distribution. Component sample values must not be ordered.
- Step 3 : Simulate minimal cut-set sample values and sample system reliability values.
- Step 4 : Sort the sample system reliability values in increasing order of magnitude.
- Step 5 : From the ordered values of system reliability obtain system reliability limits at any specified confidence level.

To illustrate the Monte Carlo simulation technique, we shall use the reliability network shown in Fig. 1. The problem is of course a hypothetical one, but it allowed us to compare the method of moments approach with the exact analytical result and now allows us to compare the approximate lower confidence limit from the Chebyshev's inequality with the present Monte Carlo solution. Components one through five were assumed to be equals, to follow the exponential failure density and tested until all components failed.

Using the previous Monte Carlo technique, one thousand samples of $Q(t_m)$ from Eqn. (5.70) were obtained for the components. These component-samples combined through the minimal cut-set matrix generated one thousand sample values for each minimal cut-set which were used to simulate the sample reliability values using Eqn. (5.71). The system reliability values were ordered to construct an approximate step distribution. Table XXXIII presents the results for 1000 experiments with **a** component mean-time-to-failure of 4380 hours, a relation of mission time to MTTF of 0.05 and a sample size of 32. Column 1 is the confidence level that the system reliability will fall between the given upper and lower limits of columns 2 and 3, respectively. Column 4 represents the one-sided confidence level that the system reliability is less than or equal to the values of column 2 or equal to or greater than the values of column 3.

The program was also run for 100, 500 and 1000 simulations with the same data as before except that a sample size of 16 was used. Comparison of the 1000-experiments of Tables XXXIII and XXXIV indicate that the higher the sample size the narrower the interval at any confidence level. This effect is obtained for any number of simulations used as shown in Table XXXV. This table also shows that the same effect is obtained when for the same sample size the number of trials is increased.

Table XXXVI indicates that the interval given by the Monte Carlo approach improves in the high reliability region. This effect is even more clear from Table XXXVII where the limits were obtained for different relations of the mission time to the component mean-time-tofailure.

5.9.3 Comparison with the moment approach

Table XXXVIII shows that one-hundred cycles of the simulation process (Monte Carlo method) for the sample system of Fig. 1 resulted in data samples for the system reliability having mean and variance very close to those computed using the equations of the moment approach. Hence, 100 cycles of the simulation process was deemed adequate.

Table XXXIX shows that the lower limit provided by the Chebyshev's inequality is always more pessimistic than the one from the Monte Carlo technique and obviously the former improves for more stable components, i.e., components with smaller variance. One of the major drawbacks of the Monte Carlo method is the large number of trials required to give a representative numerical distribution, and the time of the present computer approach is increased as the sample size and the number of trials is increased. Table XXXIX presents the computer times (seconds of CPU) for a selected number of sample sizes between 2 and 32 for 100, 500 and 1000 experiments. It must be emphasized that the computer time of the Chebyshev's approach is unaffected by the component sample size and is a one-experiment solution. The average computer time for the Chebyshev's inequality approach is 2.4 seconds of CPU in a CDC-6400 digital computer.

TABLE XXXIII

CONFIDENCE LIMITS FROM THE MONTE CARLO TECHNIQUE

FOR NON-REPAIRABLE SYSTEMS

MTTF = 4380 hrs. Sample size = $32 \text{ t}_{\text{m}}/\text{MTTF} = 0.05$ Monte Carlo size = 1000 experiments

Two-sided confidence level l	Upper limit 2	Lower limit 3	One-sided confidence level 4
99.0	0.996895	0.992634	99•5
98.0	0.996822	0.992818	99•0
97.0	0.996711	0.992902	98.5
96.0	0.996521	0.993022	98.0
95.0	0.996472	0.993165	97•5
90.0	0.996283	0.993558	95.0
80.0	0.996048	0.993938	90.0
70.0	0.995869	0.994129	85.0
60.0	0•995750	0•994294	80.0
50.0	0.995643	0•994486	75.0

TABLE XXXIV

CONFIDENCE LIMITS FROM THE MONTE CARLO TECHNIQUE FOR NON-REPAIRABLE SYSTEMS WITH DIFFERENT NUMBER OF TRIALS MTTF = 4380 hrs. Sample size = 16 tm/MTTF = 0.05

Two- sided	100-experiments		o- 100-experiments 500-e		500-expe	riments	1000-exper	iments
confi- dence level	Upper limit	Lower limit	Upper limit	Lower limit	Upper limit	Lower limit		
99.0	0.997479	0.991266	0.997325	0.991118	0.997600	0.991115		
98.0	0.997474	0.991274	0.997216	0.991349	0.997399	0.991406		
97.0	0.997161	0.992248	0.997197	0.991847	0.997269	0.991976		
96.0	0.997106	0.992488	0.997144	0.992111	0.997118	0.992109		
95.0	0.996947	0.992710	0.997032	0.992312	0.997094	0.992339		
90.0	0 •99 6 45 7	0.993058	0.996817	0 .992684	0.996826	0.992941		
80.0	0.996232	0.994099	0.996543	0 •9934 62	0.996514	0.993486		
70.0	0.995891	0.994674	0.996363	0 .9938 80	0.996294	0.993818		
60.0	0•995599	0.994924	0.995199	0.994182	0.996082	0.994072		
50.0	0.995308	0.995304	0.996073	0.994370	0.995903	0.994274		

TABLE XXXV

90% CONFIDENCE LIMITS FOR DIFFERENT SAMPLE SIZES

```
MTTF = 4380 Hrs. t_m/MTTF
```

0.989928

0.991739

0.992710

0.993456

100-exp	eriments	500-exper	iments	1000-expe	eriments
Upper limit	Lower limit	Upp er l imit	Lower limit	Upper limit	Lower limit
0.999337	0.987198	0.999120	0.988493	0.999082	0.988200

0.998179

0.997584

0.996817

0.996339

= 0.05

0.991117

0.992072

0.992684

0.993550

TABLE XXXVI

90% CONFIDENCE LIMITS FOR DIFFERENT MEAN TIME TO FAILURE

1000-experiments

0.998375

0.997548

0.996631

0.996602

Sample size

2

4

8

16

32

Sample size = 32

 $t_m = 100 \text{ hrs}$

0.998299

0.997431

0.996826

0.996283

Component	1000-experiments		
MTTF	Upper limit	Lower limit	
438 876 1095 2190	. 0.924528 0.980748 0.987608 0.996902	0.874458 0.967168 0.978845 0.994619	
43 80	0.999228	0.998650	

0.990204

0.991914

0.992941

TABLE XXXVII

90% CONFIDENCE LIMITS FOR DIFFERENT RELATION OF MISSION TIME TO COMPONENT MEAN TIME TO FAILURE

1000-experiments Sample size = 32 MTTF = 4380 hrs.

t _m /MTTF	Upper limit	Lower limit
0.01	0.999852	0•999741
0 .05	0.996283	0.993558
0.10	0.985170	0•974729
0.20	0.941678	0.902517
0.30	0.872867	0.790141
0.40	0.780800	0.644682
0.50	0.670542	0.474584

TABLE XXXVIII

ESTIMATE AND VARIANCE OBTAINED ANALYTICALLY AND BY MONTE CARLO SIMULATION FOR DIFFERENT NUMBERS OF TRIALS

Component MTTF = 4380 hrs. $t_m/MTTF = 0.5$ Sample size = 32 Analytical solution Estimate = 0.58633

Estimate = 0.58633Variance = 0.72643×10^{-2}

Monte Carlo size	Estimate	% diff.	Variance	% diff.
20	0.56197	4.3	0.57660×10^{-2}	26.0
30	0.56734	3.3	0.59543×10^{-2}	22.0
50	0.57673	1.7	0.60500×10^{-2}	20.1
100	0.58653	0.03	0.71288×10^{-2}	1.9

TABLE XXXIX

COMPARISON OF 90% LOWER CONFIDENCE LIMIT FOR

DIFFERENT SAMPLE SIZES

Monte Carlo	size =	100 experiments
MTTF = 4380	hrs.	$t_{\rm m}/\rm MTTF = 0.05$

Sample size	Chebyshev's inequality	Monte Carlo technique	% diff.
2	0.92690	0.98720	6.1
4	0.95675	0 . 9899 3	3.4
8	0.97015	0.99174	2.2
16	0.97787	0.99271	1.5
- 32	0•99945	0.99346	0.6
			<u> </u>

TABLE XL

COMPUTER TIME (SECONDS OF CPU) FOR DIFFERENT NUMBER OF TRIALS AND SAMPLE SIZES

Sample	Number o	f experiments	3
size	100	500	1000
2.	1.50	2.20	4.50
4	1.60	2.30	4.90
8	1,80	3.50	5.60
16	2.10	4.90	8.00
32	2.30	6.30	12.10
			[

5.9.4 Repairable systems

This section presents a new Monte Carlo technique for the probability and frequency of system failure which combines minimal cutset representation and Monte Carlo simulation. The individual components have been assumed to be statistically independent of each other, and repair of failed components is allowed. The method is a synthetic one, in that the steady-state probability that components are failed and the rate of departure from the failed state for each components are generated and then used to determine numerical distributions for the probability and frequency of system failure.

The probability that component i is failed in a steady state situation is given by

$$P_{d_{i}} = \frac{\lambda_{i}}{\lambda_{i} + \mu_{i}}$$

where λ_i is the component failure rate

 μ_i is the component repair rate.

The one-to-one correspondence between P_{d_i} and the ratio of its repair and failure rate μ_i/λ_i is obvious and will be used to define F-distributed variables from which P_d -sample values can be generated.

The usual sample estimate of P_{d_i} is

$$\hat{P}_{d_{i}} = \frac{1}{1 + \hat{\mu}_{i}/\hat{\lambda}_{i}}$$
 (5.74)

Let us assume that t_1 (time-to-failure) and t_2 (time-to-repair) are statistically independent random variables with exponential probability density functions

$$f_1(t_1) = e^{-\lambda t_1}$$
 (5.75)

$$f_2(t_2) = e^{-\mu t_2}$$
 (5.76)

then $\hat{\lambda}$, the sample estimate of λ , is obtained from

$$\hat{\lambda} = \frac{n_1}{\sum_{i=1}^{n_1} t_{1_i}}$$
(5.77)

where t₁ = time between the (i-1)st and the i-th failures

n, = number of failures

2

and $\hat{\mu},$ the sample estimate of $\mu,$ is obtained from

$$\mu = \frac{\frac{n_2}{n_2}}{\sum_{j=1}^{n_2} t_{2j}}$$
(5.78)

where ${}^{t}2_{j}$ = time-to-repair associated with the j-th failure n_{2} = number of repair actions initiated.

Consider a random sample of n_1 times-to-failure and n_2 timesto-repair drawn from the populations described by Eqns. (5.75) and (5.76) with sample means $\hat{\lambda}$ and $\hat{\mu}$ calculated from Eqns. (5.77) and (5.78) respectively. It is well known that $2n_1\lambda/\hat{\lambda}$ and $2n_2\mu/\hat{\mu}$ are chisquare distributed variables with $2n_1$ and $2n_2$ degrees of freedom respectively. Since they are independent due to the independence of the random variables t_1 and t_2 , it is possible to define a new variable

$$\frac{2 n_2 \mu/\hat{\mu}}{2 n_2} = \frac{\hat{\lambda} \mu}{\hat{\mu} \lambda}
 (5.79)
 \frac{2 n_1}{2 n_1}
 \frac{\lambda/\hat{\lambda}}{2 n_1}
 = \frac{\hat{\lambda} \mu}{\hat{\mu} \lambda}
 (5.79)
 \frac{2 n_1}{2 n_1}$$

which is F-distributed with $2n_2$, $2n_1$ degrees of freedom. Similar sample estimators for t_1 and t_2 of those given by Eqns. (5.77) and (5.78) can be formulated for the gamma and Weibull probability density function⁽⁶⁰⁾. Therefore a similar z-variable can be defined also for such distributions.

The variable z_1 can be used to obtain a lower confidence limit for P_{d_i} as follows:

$$\Pr\left\{\frac{\mu \hat{\lambda}}{\lambda \hat{\mu}} \ll F_{1-\alpha} \quad (2n_2, 2n_1)\right\} = 1 - \alpha$$

$$\Pr\left\{\frac{\mu}{\lambda} \notin \frac{\hat{\mu}}{\hat{\lambda}} F_{1-\alpha} (2n_2, 2n_1)\right\} = 1 - \alpha$$

$$\Pr\left\{1+\frac{\mu}{\lambda} \leqslant 1+\frac{\hat{\mu}}{\hat{\lambda}} F_{1-\alpha}\left(2n_{2},2n_{1}\right)\right\} = 1-\alpha$$

$$\Pr\left\{\frac{1}{1+\frac{\hat{\mu}}{\hat{\lambda}}} F_{1-\alpha}\left(2n_{2},2n_{1}\right) \ll \frac{1}{1+\frac{\mu}{\lambda}}\right\} = 1-\alpha \quad (5.80)$$

In most practical cases $n_1=n_2$ and Eqn. (5.80) becomes

$$\Pr\left\{\frac{1}{1+\frac{\hat{\mu}}{\hat{\lambda}}}F_{1-\alpha}(2n,2n)} \leqslant P_{d}\right\} = 1-\alpha \quad (5.81)$$

and the $(1 - \alpha)$ lower confidence limit is found from

$$P_{d_{L}} = \frac{\hat{\lambda}}{\hat{\lambda} + \hat{\mu} F_{1-\alpha}(2n,2n)}$$
 (5.82)

Equation (5.83) is used to generate sample steady-state probability of failure values for each component

$$P_{d_{i}} = \frac{\hat{\lambda}_{i}}{\hat{\lambda}_{i} + \hat{\mu}_{i} F_{2n_{i}}, 2n_{i}}}$$
(5.83)

where $F_{2n_i}, 2n_i$ represents random values drawn from an F-distribution with $(2n_i, 2n_i)$ degrees of freedom.

As mentioned before $2n_2 \mu/\hat{\mu}$ is chi-square distributed with $2n_2$ degrees of freedom. Hence, a $(1 - \alpha)$ lower confidence limit for μ can be obtained as follows :

$$\Pr\left\{\begin{array}{c}\frac{2n_{2}\mu}{\hat{\mu}} \ll \chi^{2} \\ \frac{1-\alpha}{2n_{2}}\end{array}\right\} = 1 - \alpha$$

$$\Pr\left\{ \mu \ll \frac{\hat{\mu}}{2n_2} \times \left\{ \begin{array}{c} 2 \\ 1-\alpha, 2n_2 \end{array} \right\} = 1 - \alpha \right\}$$

and the $(1 - \alpha)$ lower confidence limit for μ with $n_2 = n$ is found from

$$\mu_{\rm L} = \frac{\hat{\mu}}{2n} \chi_{1-\alpha,2n}^{2}$$
(5.84)

Equation (5.85) is used to generate sample values of the rate of departure from the failed state for each component, thus

$$\mu_{i} = \frac{\hat{\mu}_{i}}{2n_{i}} \chi^{2}$$
(5.85)

where $\chi^2_{n_i}$ represents random values drawn from a chi-square distribution with $2n_i$ degrees of freedom.

The Monte Carlo technique involves use of a digital computer to obtain a large number of component sample values of the steadystate probability of being in the down state P_{d_i} and for the rate of departure from the failed state μ_i . These values are combined through Eqns. (5.86) and (5.87) to obtain sample values for P_f and f_f respectively.

$$P_{f} = \sum_{i j \in i} \prod_{j \in i} P_{dj} ; i = 1, NMC$$
 (5.86)

$$\mathbf{f}_{\mathbf{f}} = \sum_{\mathbf{i}} \prod_{\mathbf{j} \in \mathbf{i}} \mathbf{P}_{\mathbf{d}_{\mathbf{j}}} \cdot \sum_{\mathbf{j} \in \mathbf{i}} \boldsymbol{\mu}_{\mathbf{j}}$$
(5.87)

Use of the same ordering procedure mentioned for the non-repairable system, allows us to form numerical distributions from which approximate confidence limits for both reliability indices at any confidence level can be obtained.

5.9.5 Algorithm and an illustrative system

The main steps of the Monte Carlo technique can be summarised as follows:

- Step 1 : Generate the minimal cut-set and cut-node incidence matrices of the system.
- Step 2 : Generate sample values for the steady-state probability of being in the down state and for the rate of departure from the failed state for each component using random number generators from an F- and Chi-square distribution respectively. Sample values must not be ordered.
- Step 3 : Simulate sample minimal cut-set values using the component sample values from step 2 and the minimal cut-set matrix of step 1.

- Step 4 : Simulate sample values for the probability and frequency of failure of the i-th node of the system using the information from the cut-node incidence matrix.
- Step 5 : Sort the probability and frequency of i-node failure arrays in increasing order of magnitude.
- Step 6 : From the ordered values of step 5 obtain confidence limits for the probability and frequency of i-node failure at any specified confidence level. If reliability indices are required for all or for special nodes of the system repeat 4, 5 and 6 for such nodes.

For illustrative purposes we use the reliability network of Fig. 1 to compute approximate confidence limits for the probability and frequency of failure of node 4. Components one through five were assumed to be equal, to follow the failure and repair exponential densities and were tested until all components failed.

One thousand samples of P_d and μ from Eqns. (5.83) and (5.85) respectively were generated for the components. These values combined through the minimal cut-set and cut-node incidence matrices simulated 1000-samples of the probability and frequency of node 4 failure. The node samples were ordered to construct an approximate step distribution. Tables XLI and XLII present the results for 1000 experiments using a component failure rate of 2 f/year and a mean-time-to-repair of 20 hours for a component sample size of 10. Columns 1 to 4 in both tables have the same interpretation as given in the non-repairable case.

Results were also obtained for 80, 500 and 1000 experiments with the same data as before but the sample size was assumed to be 16.

Tables XLIII and XLIV show the results for the probability and frequency of failure respectively. Comparison of the results of the 1000 experiments for sample sizes of 10 and 16 show that the larger the sample size the narrower the reliability intervals at any confidence level. It is observed that both the sample size and the number of trials, increase the computer time.

5.9.6 Comparison with the Moment approach

Table XLV shows that eighty cycles of the simulation process for the sample system of Fig. 1 resulted in data samples for the probability and frequency of system failure having mean and variance very close to those computed using the equations of the analytical approach. Hence, 80 cycles of the simulation process was deemed adequate.

Tables XLVI and XLVII compare the two-sided 80 and 90% confidence limits, for different sample sizes, obtained from the Monte Carlo with the same results provided by the Chebyshev's inequality for the probability and frequency of failure respectively. The upper bound provided by the Chebyshev's inequality is more conservative than the limit from the Monte Carlo simulation. They approach each other when the sample size increases.

Table XLVIII presents the computer time for the Monte Carlo approach for different sample sizes and 500 experiments. It is observed that the larger the sample size i.e., the more stable is the component data, the longer the computer time for the Monte Carlo technique. Obviously the time will be even longer still if we increase the number of trials. The Chebyshev's approach, although more pessimistic, has the advantage of being a one-experiment solution and its time solution is independent of the component sample size. The average computer time of the Chebyshev's approximation for the same cases of table XLVIII was 0.625 s. As the variance of the reliability indices decrease for highly stable components (see tables XVI and XVII) the bounds from the Chebyshev's inequality will improve as the component sample size increase.

TABLE XLI

APPROXIMATE CONFIDENCE LIMITS FROM THE MONTE CARLO TECHNIQUE FOR THE PROBABILITY OF FAILURE

Component	failure rate	2	2 f/year
Component	repair rate	-	438 repairs/year
Component	sample size	=	10
Number of	trials	Ħ	1000

Two-sided confidence level 1	Upper limit 2	Lower limit 3	One -sided confidence level 4
99.0	0.11354 x 10 ⁻³	0.18715 x 10 ⁻⁴	99•5
98.0	0.99245 x 10 ⁻⁴	0.21985 x 10 ⁻⁴	99.0
97.0	0.89982 x 10 ⁻⁴	0.24545 x 10 ⁻⁴	98•5
96.0	0.85320 x 10 ⁻⁴	0.26095 x 10 ⁻⁴	98.0
95.0	0.78378 x 10 ⁻⁴	0.28334 x 10 ⁻⁴	97•5
90.0	0.74842×10^{-4}	0 .2987 6 x 10 ⁻⁴	95.0
80.0	0.68322 x 10 ⁻⁴	0.32120 x 10 ⁻⁴	90.0
70.0	0.63162 x 10 ⁻⁴	0.34686 x 10 ⁻⁴	85.0
60.0	0.59447×10^{-4}	0.37234 x 10 ⁻⁴	80.0
50.0	0 -55 660 x 10 ⁻⁴	0.39377 x 10 ⁻⁴	75.0

Computer time for tables XLI and XLII = 14.9 s. of CPU

TABLE XLII

APPROXIMATE CONFIDENCE LIMITS FROM THE MONTE CARLO TECHNIQUE FOR THE FREQUENCY OF FAILURE

Component fai	lure rate =	2 f/year
Component rep	air rate 😑	438 repairs/year
Number of tri	als =	1000
Component sam	ple size 📟	10

Two-sided confidence level l	Upper limit 2	Lower limit 3	One-sided confidence level 4
99.0	0.106118	0.015220	99•5
98.0	0.091146	0.018642	99.0
97.0	0.081142	0.020209	98.5
96.0	0.073153	0.022182	98.0
95.0	0.068939	0.024022	97•5
90.0	0.066956	0.025227	95.0
80.0	0.060108	0.027326	90.0
70.0	0.055070	0.029681	85.0
60.0	0.051658	0.032263	80.0
50.0	0.048486	0.034276	75.0

TABLE XLIII

PROBABILITY OF FAILURE

APPROXIMATE CONFIDENCE LIMITS FROM THE MONTE CARLO TECHNIQUE FOR DIFFERENT NUMBER OF SIMULATIONS

 $\lambda_i = 2 f/year$

µ_i = 438 repairs/year

sample size = 16

2-sided	80 - exp	eriment s	500 - expe	periments 1000-experiment		iments
confid. level	Upper limit	Lower limit	Upper limit	Lower limit	Upper limit	Lower limit
99.0	0.816×10^{-4}	0.233×10^{-4}	0.844×10^{-4}	0.221×10^{-4}	0.860×10^{-4}	0.225×10^{-4}
98.0	0.700×10^{-4}	0.251×10^{-4}	0.766×10^{-4}	0.253×10^{-4}	0.777×10^{-4}	0.256×10^{-4}
97.0	0.659×10^{-4}	0.264×10^{-4}	0.711×10^{-4}	0.274×10^{-4}	0.722×10^{-4}	0.276×10^{-4}
96.0	0.639×10^{-4}	0.278×10^{-4}	0.682×10^{-4}	0.296×10^{-4}	0.677×10^{-4}	0.295×10^{-4}
95.0	0.614×10^{-4}	0.286×10^{-4}	0.658×10^{-4}	0.311×10^{-4}	0.642×10^{-4}	0.311×10^{-4}
90.0	0.602×10^{-4}	0.300×10^{-4}	0.636×10^{-4}	0.323×10^{-4}	0.619×10^{-4}	0.322×10^{-4}
80.0	0.559×10^{-4}	0.319×10^{-4}	0.605×10^{-4}	0.336×10^{-4}	0.577×10^{-4}	0.346 x 10 ⁻⁴
70.0	0.520×10^{-4}	0.332×10^{-4}	0.567×10^{-4}	0.353×10^{-4}	0.544×10^{-4}	0.362×10^{-4}
60.0	0.500×10^{-4}	0.362×10^{-4}	0.534×10^{-4}	0.365×10^{-4}	0.514×10^{-4}	0.379×10^{-4}
50.0	0.477×10^{-4}	0.382×10^{-4}	0.508×10^{-4}	0.394×10^{-4}	0.492×10^{-4}	0.396 x 10 ⁻⁴
Time*	3.10		3.10 8.60		19.40	

* Computer time in seconds of CPU for tables XLIII and XLIV

TABLE XLIV

FREQUENCY OF FAILURE

APPROXIMATE CONFIDENCE LIMITS FROM THE MONTE CARLO TECHNIQUE FOR DIFFERENT NUMBER SIMULATIONS

 $\lambda_i = 2 f/year$

μ_i = 438 repairs/year

sample size = 16

2-sided	80 - experiments		500 - exp	eriments	1000 - experiments	
confid. level	Upper limit	Lower limit	Upper limit	Lower limit	Upper limit	Lower limit
99.0	0.068789	0.018970	0.075004	0.018019	0.078305	0.019258
98.0	0.059203	0.021769	0.068794	0.021400	0.068699	0.021614
97.0	0.057627	0.023003	0.064587	0.022943	0.064372	0.023395
96.0	0.057186	0.024773	0.061666	0.024746	0.060134	0.024654
95.0	0.053620	0.025260	0.059069	0.026049	0.057546	0.026579
90.0	0.051733	0.026061	0.057234	0.027037	0.055374	0.027761
80.0	0.047365	0.027598	0.054259	0.028512	0.051075	0.030033
70.0	0.045595	0.029213	0.050662	0.030060	0.048230	0.031573
60.0	0.043263	0.031013	0.047824	0.031571	0.046354	0.033388
50.0	0.039641	0.032914	0.045105	0.034078	0.043638	0.034634

TABLE XLV

COMPARISON OF ESTIMATE AND VARIANCE OBTAINED ANALYTICALLY AND BY MONTE CARLO FOR DIFFERENT NUMBER OF

TRIALS

- $\lambda = 2 f/year$
- μ = 438 repairs/year
- Sample size = 32

Analytical	solution	:	P_{f}	#	0.44158×10^{-4}
		Var	(P_f)	8	0.10917×10^{-9}
			f_{f}	=	0.03760
		Var	$(\mathbf{f}_{\mathbf{f}})$	8	0.52265×10^{-4}

Monte	Pro	Probability of failure				Frequency of failure		
Size	Estimate	% Diff.	Variance	% Diff.	Estimate	% Diff.	Variance	% Diff.
20 40 50 80	0.40790x10 ⁻⁴ 0.46482x10 ⁻⁴ 0.43105x10 ⁻⁴ 0.44241x10 ⁻⁴	8.3 5.0 2.4 0.2	0.13461x10 ⁻⁹ 0.12722x10 ⁻⁹ 0.10225x10 ⁻⁹ 0.11325x10 ⁻⁹	18.9 14.2 6.8 5.5	0.03990 0.03591 0.03703 0.03777	5.8 4.7 1.5 0.5	0.94401x10 ⁻⁴ 0.80533x10 ⁻⁴ 0.70594x10 ⁻⁴ 0.66276x10 ⁻⁴	44.6 35.1 26.0 16.6

TABLE XLVI

COMPARISON OF CONFIDENCE LIMITS OBTAINED FROM THE CHEBYSHEV'S INEQUALITY AND BY MONTE CARLO SIMULATION FOR THE SAMPLE SYSTEM IN FIGURE 1

PROBABILITY OF FAILURE

 $\lambda = 2 f/year$

٠.

 μ = 438 repairs/year

·	limit		
Analytical Chebyshey La App	Monte Carlo	% diff.	An Chob

Sample	Confi-	Lower limit		Upper limit			
size	level	Analytical Chebyshev's App.	Monte Carlo	% diff.	Analytical Chebyshev's App.	Monte Carlo	% diff.
2	80	0.0	.24406 x 10 ⁻⁴	-	.25367 x 10 ⁻³	.15766 x 10 ⁻³	60.9
	90	0.0	$.19387 \times 10^{-4}$.30755 x 10 ^{−3}	.17049 x 10 ⁻³	80.4
J	80	0.0	.28339 x 10 ⁻⁴	-	1.43972×10^{-4}	.92113 x 10 ⁻⁴	36.0
4	90	0.0	.26578 x 10 ⁻⁴	-	1.75081×10^{-3}	.10395 x 10 ⁻³	68.4
0	80	0.04069×10^{-4}	$.30804 \times 10^{-4}$	86.8	1.02401×10^{-4}	.69031 x 10 ⁻⁴	48.3
0	90	0.0	$.28057 \times 10^{-4}$	-	1.22766×10^{-4}	$.77275 \times 10^{-4}$	58.9
76	80	0.13396×10^{-4}	$.31873 \times 10^{-4}$	58.0	80568 x 10 ⁻⁴	•55947 x 10 ⁻⁴	44.0
10	90	0.0	$.30019 \times 10^{-4}$	-	$.94480 \times 10^{-4}$	$.60237 \times 10^{-4}$	56.8
70	80	0.20795×10^{-4}	•35286 x 10 ⁻⁴	41.1	.67521 x 10 ⁻⁴	.52114 x 10 ⁻⁴	29.6
)2	90	0.11117×10^{-4}	$.33220 \times 10^{-4}$	66.5	$.77199 \times 10^{-4}$.54512 x 10 ⁻⁴	41.6
61	80	0.26424×10^{-4}	$.36788 \times 10^{-4}$	28.2	$.59202 \times 10^{-4}$	$.48297 \times 10^{-4}$	22.6
04	90	0.19636×10^{-4}	$.36098 \times 10^{-4}$	45.6	.65980 x 10 ⁻⁴	$.49113 \times 10^{-4}$	34.4
1.09	80	0.30613×10^{-4}	$.37017 \times 10^{-4}$	17.3	•53699 x 10 ⁻⁴	.45039 x 10 ⁻⁴	19.2
170	90	0.25832 x 10 ⁻⁴	•35538 x 10 ⁻⁴	27.3	.58480 x 10 ⁻⁴	$.46817 \times 10^{-4}$	24.9

TABLE XLVII

COMPARISON OF CONFIDENCE LIMITS OBTAINED FROM THE CHEBYSHEV'S INEQUALITY AND BY MONTE CARLO SIMULATION FOR THE SAMPLE SYSTEM IN FIGURE 1

FREQUENCY OF FAILURE

 $\lambda \doteq 2 \text{ f/year}$

i.

 $\mu = 438$ repairs/year

Sample	Confi-	Lower limit		Upper limit			
Size	dence level	Analytical Chebyshev's ineq.	Monte Carlo	% diff.	Analytical Chebyshev's ineq.	Monte Carlo	% diff.
2	80	0.0	.0170057	-	.162316	. 102333	58.6
2	90	0.0	.0138928	-	. 199597	.187130	6.7
1.	80	0.0	.0242289	-	.1003648	.0857474	17.0
4	90	0.0	.0215687	-	.1218895	.0942943	29.3
Q	80	.0075509	.0220612	65.8	.0755891	.0635672	18.9
0	90	.0065403	.0202890	67.8	.0896803	.0742623	20.8
16	80	.0156010	. 0275988	43.5	.0620790	•0473646	31.1
	90	.0059751	.0260608	77.1	.0717049	.0517327	38.6
70	80	.0214345	•029 7 346	27.9	.0537655	•0474789	13.2
22	90	.0147385	• 0 282148	47.8	.0604615	.0 48 8776	· 23.7
61	80	.024151	.0 <u>32</u> 6170	26.0	•049869	.0418941	19.0
04	90	.0209732	.0314058	33.2	.0530468	. 0434 75 8	22.0
1.28	80	.0287431	•0322662	10.9	.0447169	. 0397986	12.4
120	90	.0254349	.0315736	19.4	.0480251	.0407258	17.9

TABLE XLVIII

COMPUTER TIME OF 500-EXPERIMENT MONTE CARLO TECHNIQUE FOR THE PROBABILITY AND FREQUENCY OF NODE FAILURE

Sample size	Computer Time (seconds of CPU)
2	4.10
4	4.50
8	5.98
16	8.60
32	14.40
64	25.90
128	49.30

5.10 Distributions of the reliability indices

Determination of the exact distribution of the probability and frequency of system failure does not seem practicable. Firstly, the probability transformations involved in the minimal cut-set formulae with the exponential failure and repair models do not result in a tractable final distribution. Secondly, different distributional forms for the system components would seem to prevent general solutions for the required distributions.

The approach adopted here was to use the data produced by Monte Carlo simulation (section 5.9) for the sample network in Fig. 1 to discover the distributions of the probability and frequency of system failure. All components of the sample network were assumed to follow the exponential failure and repair models. The normal distribution was tested by statistical goodness-of-fit tests and by probability plotting. In both cases, repairable and nonrepairable, the normal distribution was found to provide a good fit to the data produced by the Monte Carlo simulation for the sample system when the component data had been obtained from high sample sizes i.e. when the data is more stable. Figures 2 and 3 illustrate the fit of the normal distribution to the data produced by Monte Carlo simulation for the sample system in the nonrepairable and repairable case respectively. Tables XLIX and L show the results of 4 selected tests of the Chi-square test for different sample sizes in the nonrepairable and repairable cases respectively.

Table LI to LIII compare limits for two-sided 80 and 90 per cent confidence intervals on the reliability indices when repair is or not permitted respectively. The agreement between lower and upper confidence limits obtained by the normal approximation and by Monte Carlo simulation



Fig.2. Probability plotting of the reliability of the system in Fig. 1


Fig. 3. Probability plotting of the reliability indices for the system in Fig. 1

TABLE XLIX

GOODNESS-OF-FIT TEST FOR NORMAL DISTRIBUTION . CASE OF NONREPAIRABLE COMPONENTS

> Component MTTF = 4380 hrs. t_m/MTTF = 0.5

No. of trials = 100

Sample size	Test	Degrees of freedom	Theoretical $\chi^2_{0.90}$	$rac{2}{\chi}^2$
	1	3	6.25	22.22
2	2	2	4.61	11.87
2	3	3	6.25	14.71
	4	4	7.78	20.85
	1	4	7.78	11.43
	2	4	7.78	10.48
4	3	4	7.78	15.32
	4	4	7.78	9,83
	<u>ן</u>	5	9.24	8.94
•	2	4	7.78	4.46
Ö	3	4	7.78	3.97
	4	5	9.24	4.23
	1	5	9.24	8.13
	2	3	6.25	3.34
16	3	4	7.78	4.68
	4	4	7.78	7.06
	1	4	7.78	5.89
70	2	4	7.78	2.12
<i>3</i> 2	3	3	6.25	2.42
	4	4	7.78	2.37
	1	3	6.25	.2.56
64	2	4	7.78	3.02
	3	3	6.25	2.50
	4	5	9.24	2.22

TABLE L

GOODNESS-OF-FIT TEST FOR NORMAL DISTRIBUTION CASE OF REPAIRABLE COMPONENTS

 λ = 2 f/year μ = 438 repairs/year No. of trials = 80

Sample size	Test	Degrees of freedom	Theoretical $\chi^2_{_{0.90}}$	χ^2	
	P	ROBABILITY O	F FAILURE		
8	1 2 3 4	3 4 2 4	6.25 7.78 4.61 7.78	13.52 19.09 14.76 23.15	
16	1 2 3 4	4 5 4 4	7.78 9.24 7.78 7.78 7.78	7.81 10.83 16.05 21.19	
64	1 2 3 4	<u>4</u> <u>4</u> <u>4</u> <u>4</u>	7.78 7.78 7.78 7.78 7.78 7.78	4.35 3.92 6.80 2.67	
FREQUENCY OF FAILURE					
8	$ \frac{1}{2} \overline{3} \overline{4} $	2 4 3 4	4.61 7.78 6.25 7.78	29.80 32.08 19.44 16.62	
16	1 2 3 4	4 5 4 4	7.78 9.24 7.78 7.78 7.78	9.22 9.71 10.83 16.18	
64	1 2 3 4	5 3 4 4	9.24 6.25 7.78 7.78 7.78	7.77 6.07 4.22 3.66	

•

TABLE LI

COMPARISON OF CONFIDENCE LIMITS OBTAINED FROM THE NORMAL APPROXIMATION AND BY MONTE CARLO SIMULATION

Component MTTT = 4380 hrs.

No. of trials = 100

Sample	Confidence	Lower limit		Upper limit			
size	level	Normal approximation	Monte Carlo	% Diff.	Normal approximation	Monte Carlo	% Diff.
8	80	0.41921	0.48536	13.6	0.86015	0.74109	16.1
0	90	0.35635	0.40294	11.6	0.92301	0.76766	20.2
16	80	0.44927	0.50163	10.4	0.75915	0.68755	10.4
10	90	0.40508	0.46206	12.3	0.80334	0.69772	15.1
70	80	0.47723	0.51588	7.5	0.69543	0.65505	6.2
<u>)</u> 2	90	0.44613	0.48427	7.9	0.72653	0.67297	8.0

1

COMPARISON OF CONFIDENCE LIMITS OF THE PROBABILITY OF FAILURE OBTAINED FROM THE NORMAL APPROXIMATION AND BY MONTE CARLO SIMULATION

 λ = 2 f/year

μ = 438 repairs/year

No. of trials = 80

Sample Confidence		Lower limit			Upper limit		
size	level	Normal approximation	Monte Carlo	% Diff.	Normal approximation	Monte Carlo	% Diff.
	80	.30784 x 10 ⁻⁴	.35286 x 10 ⁻⁴	12.8	.57532 x 10 ⁻⁴	•52114 x 10 ⁻⁴	10.4
32	90	.26970 x 10 ⁻⁴	$.33220 \times 10^{-4}$	18.8	.61346 x 10 ⁻⁴	.54512 x 10 ⁻⁴	12.5
	80	•33432 x 10 ⁻⁴	.36788 x 10 ⁻⁴	9.1	.52194 x 10 ⁻⁴	.48297 x 10 ⁻⁴	8.1
64	90	.30756 x 10 ⁻⁴	$.36098 \times 10^{-4}$	14.8	.54870 x 10 ⁻⁴	.49113 x 10 ⁻⁴	11.7
	80	•35549 x 10 ⁻⁴	.37017 x 10 ⁻⁴	4.0	.48763 x 10 ⁻⁴	.45039 x 10 ⁻⁴	8.3
128	90	$.33664 \times 10^{-4}$	•35538 x 10 ⁻⁴	5.3	.50648 x 10 ⁻⁴	$.46817 \times 10^{-4}$	8.2

LII

TABLE LIII

COMPARISON OF CONFIDENCE LIMITS OF THE FREQUENCY OF FAILURE OBTAINED FROM THE NORMAL APPROXIMATION AND BY MONTE CARLO SIMULATION

> λ = 2 f/year μ = 438 repairs/year

No. of trials = 80

Sample	Sample Confidence		Lower limit		Upper limit		
size level	level	Normal approximation	Monte Carlo	% Diff.	Normal approximation	Monte Carlo	% Diff.
	80	0.0283463	0.0297346	4.7	0.0468537	0.0474789	1.3
32	90	0.0257075	0.0282148	8.9	0.0494925	0.0488776	1.3
61	80	0.0305187	0.0322662	5.4	0.0435013	0.0397986	9.3
04	90	0.0286677	0.0315736	9.2	0.0453523	0.0407258	11.4
1.00	80	0.0296683	0.0326170	9.9	0.0435012	0.0418941	3.8
120	90	0.0308543	0.0314058	1.8	0.0426057	0.0434758	2.0

is seen to be almost equal for the nonrepairable condition in table LI. The same comparison for the repairable condition in tables LII and LIII indicates that the agreement is better for the frequency of failure than for the probability of failure. In both cases for all tests made the analytical results (normal approximation) are more optimistic (lower limit) or pessimistic (upper limit) than those obtained from the simulation process. For small samples sizes (8 and 16) the upper limit of the analytical method for the nonrepairable case would appear to give differences slightly greater than the lower confidence limit. For high sample size both have almost equal deviations from the simulation results. A similar tendency was observed for the repairable condition but for both cases the agreement improves as the sample size increases.

5.11 Conclusion

This chapter has presented two different approaches for taking into account the effect of uncertain component failure and repair rates on the probability of failure of nonrepairable and repairable systems and on the frequency of failure of repairable systems. This is accomplished by computing approximate confidence limits for these reliability indices. The interval estimates define upper and lower bounds such that these random variables lie between these limits with a specified probability.

The analytical approach for the computation of confidence limits is based on the low order moments of the variables and uses Chebyshev's inequality to bound the probabilities that these random variables lie within a certain interval. The formulae for the estimate and variance of minimal cut-sets up to the third order have been presented.

The Monte Carlo simulation approach presented for the repairable case uses the one-to-one correspondence between the steady state probability of an element being down and the ratio of its repair and failure rates to define F-distributed variables from which sample values can be generated. These sample values for each component are combined through the minimal cut-set relationship between system reliability and components to generate system samples. A simulation process is made to generate the numerical distribution from which confidence limits can be computed.

The results obtained have proved that the minimal cut-set approach do not amplify the uncertainty in the component data. The tests performed show that the performance of both approaches improve as the component data is more stable and bounds from the Chebyshev's inequality are more conservative than those obtained from the Monte Carlo simulation approach. A case study has indicated that the reliability indices from the minimal cut-set approach for systems with highly stable components seem to obey a normal distribution to a reasonable approximation when the components follow the exponential failure and repair models.

CHAPTER VI

SENSITIVITY ANALYSIS AND RELIABILITY CALCULATIONS IN DIGITAL PROTECTION SYSTEMS

6.1 Introduction

The object of sensitivity analysis of system reliability indices is to improve the reliability of the system through knowledge of the component parameters that have a critical effect. Indeed, without performing such a study it is quite difficult to determine the effect of fractional changes in component parameters on fractional changes of the system reliability indices.

The aim of the present chapter is to present a sensitivity analysis approach based on the minimal cut-set approximation which combines data on the variation of the component reliability parameters with information of the sensitivity coefficients of the system reliability indices. It also presents a practical application of the methods given in preceding chapters by examining a proposed digital hardware configuration⁽⁷⁾ for integrated protection of one corner of a mesh type high voltage substation used typically on the 400 kV system of the C.E.G.B. General alternative configurations are investigated and a sensitivity analysis of the reliability indices is done for a particular application.

6.2 Sensitivity coefficients

The sensitivity coefficient of function u with respect to parameter x, designated by S_x^u , is defined as the limit of the fractional change in u, also called proportional error, divided by the fractional change in x as Δx and Δu become small, i.e.

$$S_{x}^{u} = \lim_{\Delta x \to 0} \frac{\Delta u/u}{\Delta x/x} = \frac{du/u}{dx/x} = \frac{x}{u} \frac{du}{dx}$$
(6.1)

Solving Eqn. (6.1) for the fractional change in u yields

$$\frac{\mathrm{d}u}{\mathrm{u}} = S_{\mathrm{x}}^{\mathrm{u}} \cdot \frac{\mathrm{d}x}{\mathrm{x}}$$
(6.2)

If u is a function of several variables, we can write the total differential of u as

$$du = \frac{\partial u}{\partial x_1} dx_1 + \frac{\partial u}{\partial x_2} dx_2 + \dots + \frac{\partial u}{\partial x_n} dx_r$$

Dividing by u yields

$$\frac{\mathrm{d}u}{\mathrm{u}} = \frac{\partial \mathrm{u}}{\partial \mathrm{x}_{1}} \frac{\mathrm{d}\mathrm{x}_{1}}{\mathrm{u}} + \frac{\partial \mathrm{u}}{\partial \mathrm{x}_{2}} \frac{\mathrm{d}\mathrm{x}_{2}}{\mathrm{u}} + \dots + \frac{\partial \mathrm{u}}{\partial \mathrm{x}_{n}} \frac{\mathrm{d}\mathrm{x}_{n}}{\mathrm{u}}$$
(6.3)

If we replace du/dx with $\partial u/\partial x$ in Eqn. (6.1) because u is a function of several variables and substitute into Eqn. (6.3), we obtain

$$\frac{du}{u} = S_{x_1}^{u} \frac{dx_1}{x_1} + S_{x_2}^{u} \frac{dx_2}{x_2} + \dots + S_{x_n}^{u} \frac{dx_n}{x_n}$$
(6.4)

This equation is extremely useful since each sensitivity coefficient appears as a weighting factor which tells us the effect of fractional changes in x_i on the fractional change in u. Such an equation enables the designer to evaluate his design and choice of component in terms of parameter variations, so that he will be able to identify which component improvements will result in the best system improvement.

6.3 Application to system reliability indices

A general procedure for a sensitivity analysis of the system reliability indices found from the minimal cut-set approach can be summarised as follows :

- 1. Define the reliability network of the system.
- 2. Find its minimal cut-set and cut-node incidence matrices.
- 3. Develop an equation for the system reliability indices.
- 4. Obtain the first partial derivatives of the index equations with respect to each independent component reliability measure or parameter.
- 5. Solve the index equation for the estimated values of each component reliability measure or parameter.
- 6. Solve the partial derivatives for the estimated values.
- 7. From the values obtained in 5) and 6), determine the sensitivity coefficient matrices for the system reliability indices.

The first-partial derivatives of the indices equations with respect to each independent component parameter were used in chapter V when the variance equations of such indices were formulated. For each minimal cut-set these derivatives were identified as b-coefficients with the parameter being indicated as a subindex of the b's, e.g., $b_{p\lambda_i}$ was the first-partial derivative of the steady-state probability of failure with respect to the failure rate of the i-th component. The total sensitivity coefficient of any one of the indices with respect to each component parameter or reliability measure will be the summation of the respective b-coefficients from all the minimal cut-sets which include the particular component.

For the steady-state probability of failure of a system with NMC minimal cut-sets, the sensitivity coefficients for node k are given by

$$S_{P_{f,\lambda,j}}^{n_{k}} = \frac{\lambda_{j}}{P_{f}} \sum_{i=1}^{NMC} b_{p\lambda_{j}}^{i,n_{k}}$$
(6.5)

$$S_{P_{f,j}}^{n_{k}} = \frac{\mu_{j}}{P_{f}} \sum_{i=1}^{NMC} S_{p\mu_{j}}^{i,n_{k}}$$
(6.6)

where $b_{p\lambda_j}^{i,n_k}$ means the first-partial derivative of the steady-state probability of cut i with respect to the failure rate of component j, being summated if and only if cut i is a minimal cut for node k. From Eqns. (6.5) and (6.6) it can be shown that

$$S_{P_{f},\lambda_{j}}^{n_{k}} = -S_{P_{f},\mu_{j}}^{n_{k}}$$
(6.7)

implying that the fractional changes of the failure and repair rates of the j-component have the same weight on the fractional changes of the steady-state probability of failure of node k but that the same fractional change of each parameter produces the opposite effect on the node index.

For the steady-state frequency of failure the sensitivity coefficients for node k are

$$S_{f_{j},\lambda_{j}}^{n_{k}} = \frac{\lambda_{j}}{f_{f}} \sum_{i=1}^{NMC} S_{f\lambda_{j}}^{i,n_{k}}$$
(6.8)

$$s_{f_{f},\mu_{j}}^{n_{k}} = \frac{\mu_{j}}{f_{f}} \sum_{i=1}^{NHC} b_{f\mu_{j}}^{i,n_{k}}$$
 (6.9)

Use of the minimal cut-set and cut-node incidence matrices allow us to compute efficiently sensitivity coefficients of all nodes in the system with respect to each component parameter.

The sensitivity coefficients for the probability of system failure $Q_s(t_m)$ in the interval from 0 to t_m are given by

$$S_{Q_{j}}^{Q_{s}} = \frac{Q_{j}(t_{m})}{Q_{s}(t_{m})} \sum_{i=1}^{NMC} b_{Q_{j}}^{i}(t_{m})$$

$$(6.10)$$

6.4 <u>Illustrative example</u>

Tables I and II show examples of the sensitivity coefficient matrices for the nodes of Fig. 1 in chapter V, where all elements have been assumed to be equal with a failure rate of 2 f/year, a mean-timeto-repair of 20 hours and a sample size of 257. The information in both tables shows the extent to which the node reliability index is sensitive to changes in each component parameter. For example, Table I indicates that the probability of failure of node 2 is greatly reduced with a reduction of the failure rates of components 1 and 2 but is almost unaffected even for large reductions of the failure rates of all the other components. Node 3 shows the same effects as node 2 although component 2 has more influence than component 1, being the reverse for node 2. Node 4 is equally affected by all component changes except for component 3, which has a negligible effect on its probability of failure. For all nodes, no sizable improvement of the reliability indices is obtained from an improvement of component 3. Similar conclusions are obtained from the upper-half of Table II concerning the effects of component failure changes on node frequency of failure changes. The lower half of the same table shows the opposite effect on the frequency of failure from changes in the component repair rate. The example is a trivial one but illustrates the help that the designer can obtain from the sensitivity coefficients.

Tables III and IV present the effects of different component availability on the sensitivity coefficient of node 4 for its probability and frequency of failure. For all components except component 3, the higher the availability, the greater the node sensitivity to component parameter changes.

Tables V and VI present the sensitivity coefficients of node 4 for different sample sizes of a component failure rate estimate of 2 f/year and a repair rate estimate of 438 repairs/year. Both tables indicate that for all components but component 3 the more stable are the components i.e., the lower their variances, the greater the sensitivity coefficient for both reliability indices. This tendency is the same as saying that the estimates of the probability and frequency of failure are influenced by the variances of the component parameters.

6.5 Computer-operated protection reliability

As power systems grow in complexity, protective relays play a more and more important role in the detection and removal of faulted equipment. Traditionally the protection functions in the power system substation has been performed by analog and wired-logic circuitry. Continuing rapid advances in digital computer technology have prompted a re-evaluation of protective devices and techniques with the result that digital processors are beginning to be used for the protection and control of high voltage systems.

Quantitative determinations of reliability allow comparison of different alternatives of protection system configurations to evaluate the relative reliability of different schemes. For analog relays only a few groups of papers (66-70) were found related to the subject of protective relay reliability. There is now considerable literature describing the application of digital computers to protection tasks in substations. Most papers deal with proposals for the application of digital computers to transmission line protection, busbar protection and transformer protection. The use of digital computers for protection purposes is economically more attractive if it can be applied to a number of protection and control functions simultaneously. A group of papers (67), (71-75)have considered the concepts of integration of control and protection

TABLE I

NODE PROBABILITY OF FAILURE SENSITIVITY MATRIX

 $\lambda = 2 \text{ f/year}$ $\mu = 438 \text{ repairs /year}$ Sample size = 257

Node	Failure rate sensitivities			
Com- ponent	2	3	4	
1 2 3 4 5	0.987808 0.978909 0.008899 0.004450 0.004450	0.978909 0.987808 0.008899 0.004450 0.004450	0.493912 0.493912 0.004470 0.493912 0.493912	

* Repair rate sensitivities are equal with opposite sign

TABLE II

NODE FREQUENCY OF FAILURE SENSITIVITY MATRIX

 $\lambda = 2 f/year$

μ = 438 repairs /year

Sample size = 257

Node	Failure rate sensitivities			
Com- ponent	2	3	4	
1 2 3 4 5	0.991617 0.978277 0.013340 0.006670 0.006670	0.978277 0.991617 0.013340 0.006670 0.006670	0.495821 0.495821 0.006715 0.495821 0.495821	
Node	Rej	pair rate sensitivi	ities	
Com- ponent	2	3	4	
1 2 3 4 5	-0.495778 -0.486905 -0.008873 -0.004437 -0.004437	-0.486905 -0.495778 -0.008873 -0.004437 -0.004437	-0.247341 -0.247341 -0.004467 -0.247341 -0.247341	

TABLE III

PROBABILITY OF FAILURE SENSITIVITY COEFFICIENTS FOR DIFFERENT COMPONENT FAILURE RATES . $\mu = 438$ repairs/year Sample size = 257

Failure	Component	Failure rate sensitivity*		
rate		Component 1,2,4,5	Component 3	
2	0•995455	0.493912	0.004470	
4	0.990959	0.491720	0.008820	
8	0.982063	0.487394	0.017177	
16	0.964758	0.478966	0.032610	
. 32	0.931915	0.462953	0.059022	
219	0.666667	0.332684	0.166342	

* Repair rate sensitivities are equal with opposite sign

TABLE IV

FREQUENCY OF FAILURE SENSITIVITY COEFFICIENTS FOR DIFFERENT COMPONENT AVAILABILITY

 $\mu = 438 \text{ repairs/year}$

Sample size = 257

Failure	Component	Failure r	ate	Repair r	ate
rate	availabi- lity	Component 1,2,4,5	Component 3	Component 1,2,4,5	Component 3
2	0.995455	0.495821	0.006715	-0.247341	-0.004467
4	0.990959	0.493608	0.013222	-0.245662	-0.008774
8	0.982063	0.489242	0.025637	-0.242328	-0.016935
16	0.964758	0.480738	0.048275	-0.235758	-0.031595
32	0.931915	0.464591	0.086102	-0.223024	-0.055304
219	0.666667	0.333526	0.222351	-0.111175	-0.111175
			L		<u> </u>

TABLE V

PROBABILITY OF FAILURE SENSITIVITY COEFFICIENTS FOR DIFFERENCE SAMPLE SIZES

 $\lambda = 2 f/year$

 $\mu = 438 \text{ repairs/year}$

Sample	Failure rate*		
size	Components 1,2,4,5	Components 3	
2	0.250286	0.002265	
4	0.333080	0.003014	
8	0.399089	0.003612	
16	0.442984	0.004009	
32	0.468763	0.004242	
64	0.482811	0.004369	
128	0.490156	0.004434	

* Repair rate sensitivities are equal with opposite sign

TABLE VI

FREQUENCY OF FAILURE SENSITIVITY COEFFICIENTS FOR DIFFERENT SAMPLE SIZES

 $\lambda = 2 f/year$

 $\mu = 438 \text{ repairs/year}$

Sample	Failure	rate	Repair rate	
size	Components 1,2,4,5	Components 3	Components 1,2,4,5	Components 3
2	0.333593	0.004518	-0.166413	-0.003005
4 .	0.399457	0.005410	-0.199270	-0.003599
8	0.443210	0.006003	-0.221096	-0.003993
16	0.468890	0.006351	-0.233906	-0.004224
32 [·]	0.482878	0.006540	-0.240884	-0.004350
64	0.490190	0.006639	-0.244532	-0.004416
128	0.493930	0.006690	-0.246397	-0.004450
32 64 128	0.482878 0.490190 0.493930	0.006540 0.006639 0.006690	-0.240884 -0.244532 -0.246397	-0.004350 -0.004416 -0.004450

in a computer-based substation. Most of the pertinent papers present the possible benefits in terms of speed, flexibility and cost that could potentially be offered by digital computers in these applications. In general, some form of redundancy is implicit in achieving high reliability but no explicit and systematic procedure has been used for the reliability evaluation of different schemes.

In general any protection system has to satisfy two requirements: (i) it must operate in the presence of a fault and (ii) it must be selective i.e., it must not produce false operation. Existing relaying systems are dedicated devices which usually do not operate unless a fault occurs. Since faults are relatively rare, the system spends almost its entire life in a passive state. This is all right, of course, because the system needs to operate only when a fault occurs. However, it is not known if the system will operate correctly until the next fault occurs and the system is called upon to operate. Thus the probability of failure depends upon the frequency of faults occurring within the protected zone i.e., is a function of the number of operations the relay performs and is not a function of real time. Proposed computer based protection systems or substation control systems should be continually in an active state and have a probability of failure which is independent of the frequency of fault occurrence on the protected item. This allows convenient prediction (via self-testing) of successful operation should a fault occur.

A failure in a computer system can occur either as a hardware failure or as a software failure. Whereas the former can be determined from the hardware reliability figures, software reliability is extremely difficult to predict. In this context modularity of programs and of self-checking procedures have been suggested to make software related failure arbitrarily small. The reliability measures provided by the techniques presented in this thesis are based on the steady state values

of the probability and frequency of system failure. For a computer system which is intended to operate for 20 years or so, it is evident that all hardware failures must be repaired so steady state indices can be used to measure the hardware performance. Other quantities of interest such as availability, mean down time, mean up time, etc., are easily evaluated from probability and frequency of failure indices. The sensitivity coefficients described in section (6.2) are useful when deciding where the improvements must be located. The ability of computer-based protection schemes, among others, to anticipate power system transients and swings, to change protection characteristics and to provide flexible compensation for mutual coupling effects would effectively increase the security of the protection system against false operation.

Only one of the transmission line techniques ⁽⁷⁷⁻⁷⁸⁾ is known to have been implemented in a practical field trial in the protection of a 230 kV line. Therefore lack of service experience prevents a meaningful assessment to be made of the reliability of any form of computerbased protection and control scheme. It remains to be demonstrated that such computer schemes ensure a level of reliability equal to that achieved in conventional schemes. Nevertheless any system with selfchecking facilities which allows prediction of its own operation, detects faults and provides alternative solutions, modifies the reliability concept used for conventional schemes where the faults are usually detected by failure of the equipment to operate.

6.6 <u>A computer-based protection scheme</u>

In order to study different techniques for digital protection of high voltage substations a model has been designed for one corner of a mesh substation. The details of the proposed hardware configuration including the model details were published in [7].

Fig. 1 shows the proposed hardware configuration for one mesh corner with the following blocks :

- (i) Data acquisition interface (DAI) : Digital signals (switch positions) and analog signals (current, voltage) will be fed through an appropriate interface from the power system (substation equipment).
- (ii) Data validation processor (DVP) : A small digital processor will control data collection in digital form, perform appropriate validation checks upon it and control data output to an intermediate store.
- (iii) Data intermediate storage (DIS) : At the end of each data acquisition cycle the resulting words of validated data will be transferred to the protection processor by an interface which operates in a first-in, first-out (FIFO) mode.
- (iv) Control and protection processor (CPP) : The data block from the DIS is transferred to the data base (DB) which is a portion of the CPP core store. Using the contents of the DB the CPP will perform the computations necessary for control and protection functions and provide suitable control strategy to the substation equipment.

6.7 Alternative configurations - Reliability evaluation

The general alternative configurations presented in Fig. 2 were investigated from a reliability point of view. Specifically excluded are current and voltage transformers and circuit breakers.

If all the component-blocks are assumed to be identical, with λ and μ as the failure and repair rate respectively, table VII lists the probability and frequency of failure for each system configuration.



Figure 1. A hardware configuration for a digital protection system

 $P_{\rm d}$ corresponds to the steady-state probability that a component is failed and is given by $\lambda/\lambda+\mu$.

The five different availability expressions $(1 - P_f)$ derived from table VII are compared in Figure 3. Note that the duplex system with parallel operation of the processors in charge of the data validation and protection functions will offer the best solution. The two-out-ofthree system is better than the single processor-single interface as long as $0.916 < 1 - P_d < 1.0$ and becomes worse than the single system for $0 < 1 - P_d < 0.916$, where $(1 - P_d)$ is the component-block availability. Table VIII presents a numerical comparison of the reliability indices of the five configurations.

Tables IX and X present the probability and frequency of failure of the duplex system for selected values of the component failure and repair rate respectively. Obviously both reliability indices increase as the component availability decrease. An increase of 20 times the MTTR of the components will have greater consequences on the system reliability than an increase of 20 times of its failure rate. Hence, the duplex system with parallel operation of the processors will have greater availability if themean-time-to-repair of each component is kept as low as possible i.e., the system must have a self-checking and faultlocating facility. This facility should continuously monitor the ability of the hardware to function correctly and in the event of a failure alarm, enables personnel to quickly locate the trouble.

6.8 Reliability analysis in a digital protection system

The lack of proper data for each single component prohibits a particular reliability evaluation, so an experimental sensitivity analysis was conducted on the system shown in Fig. 4 to obtain the

(1) Single processor - single interface



(2) Single processor - duplicate interface



(3) Duplex system-independent operation



(4) Duplex system - parallel operation of processors



(5) Two-out-of-three configuration



Figure 2. Alternative system configurations

TABLE VII

PROBABILITY AND FREQUENCY OF FAILURE FOR DIFFERENT SYSTEM CONFIGURATIONS

System configura- tion	Probability of failure	Frequency of failure
1	4 P _d	4 Ρ _d μ
2	$2P_{d} + 2P_{d}^{2}$	$(2P_{d} + 4P_{d}^{2}) \mu$
3	16 P ² d	32 Ρ <mark>2</mark> μ
4	6 P ² d	12 Ρ <mark>2</mark> μ
5	48 P ² d	96 Ρ <mark>2</mark> μ

TABLE VIII

COMPARISON OF RELIABILITY INDICES FOR DIFFERENT SYSTEM CONFIGURATIONS

 $\lambda = 2 f/year$

μ	=	876	repairs/	'year
---	---	-----	----------	-------

System config- uration	Probability of failure	Frequency of failure	Up-time p/unit	Down-time (hrs)
1	0.911×10^{-2}	7.98	0.124	10.00
2	0.457×10^{-2}	3.99	0.25	10.03
3	0.830×10^{-4}	0.145	6.9	5.0
4	0.311×10^{-4}	0.055	18.2	4.95
5	0.249×10^{-3}	0.436	2.2	5.0



TABLE IX

RELIABILITY INDICES OF THE DUPLEX SYSTEM FOR DIFFERENT FAILURE RATES Component MTTR = 5 hrs

Failure rate f/year	Component availability	Probability of failure	Frequency of failure
0.1	0.999943	0.194×10^{-7}	0.680×10^{-4}
0.2	0.999886	0.779×10^{-7}	0.273×10^{-3}
0.3	0.999829	0.175×10^{-6}	0.613×10^{-3}
0.4	0.999772	0.313×10^{-6}	0.109×10^{-2}
0.5	0.999715	0.487×10^{-6}	0.171×10^{-2}
0.6	0.999658	0.702×10^{-6}	0.246×10^{-2}
0.7	0.999601	0.955×10^{-6}	0.335×10^{-2}
0.8	0.999544	0.125×10^{-5}	0.438×10^{-2}
0.9	0.999487	0.158 x 10 ⁻⁵	0.554×10^{-2}
1.0	0.999430	0.195 x 10 ⁻⁵	0.683×10^{-2}
1.5	0.999145	0.439×10^{-5}	0.154×10^{-1}
2.0	0.998860	0.780 x 10 ⁻⁵	0.273×10^{-1}

TABLE X

RELIABILITY INDICES OF THE DUPLEX SYSTEM FOR DIFFERENT REPAIR TIMES

Component failure rate = 2 f/year

MTTR	Component	Probability	Frequency of
(hrs)	availability	of failure	failure
1	0.999772	0.313×10^{-6}	0.00546
2	0.999544	0.125×10^{-5}	0.01093
4	0.999086	0.501×10^{-5}	0.02195
8	0.998177	0.199×10^{-4}	0.04367
10	0.988713	0.764×10^{-3}	1.34
20	0.977679	0.299×10^{-2}	2.62

variation of the probability and frequency of system failure from a wide range of the reliability component parameters, taking into account different grades of data dispersion.

Regardless of which system configuration is chosen for a particular system, it will be quite important to determine :

(i) how the estimate and variance of the reliability indices would be affected by the failure and repair rate of the components and for the dispersion of the component data,

(ii) what will be the approximate confidence limits of the probability and frequency of failure and how the uncertainty of the data will affect these confidence limits

(iii) what component improvement will result in the maximum improvement of the system reliability indices.

A sensitivity analysis of this nature will indicate the accuracy of the estimates and will show which components have a critical effect. Indeed, without performing such a study it is quite difficult to determine how data uncertainty will affect the estimate and variance of the reliability . indices and how much confidence should be placed in the estimates.

The present sensitivity analysis was also done to see how the reliability indices were affected when components of lower availabilities such as current transformers, are included in the reliability evaluation of the scheme.

The system configuration selected is a simple representation of the digital equivalent of an overcurrent scheme for a distribution feeder. The three-phase currents are fed into the system by different channels and a programmed device (element 10) acts as main protection. Simultaneously a back-up facility (element 11) incorporated in the system to prepare a data base through which the decision logic (element 12)



- 1 3 Current transformers
- 4 6 Sample-hold circuits
- 7 9 Analog/digital converters
- 10 Programmed device
- 11 Back-up processor
- 12 Decision logic

Figure 4. A simplified digital overcurrent protection scheme

provides an additional check on the correct operation of the main protection. The scheme in Fig. 4 is a simple application of the dedicated and integrated concepts introduced in [76] and is used to illustrate the reliability analysis that can be done with the methods presented in this thesis. The minimal cut-set algorithm of section (4.7) was used to find the 29 minimal cut-sets between nodes 1 and 10 of Fig. 4. The system considered has twelve components which are identified as digital (components 4 to 12) and non-digital (components 1 to 3) in the next sections.

6.8.1 Failure rate

The failure rate of each type of component was given selected values between 0.1 to 2 f/year for the digital components and between 0.5 and 10 f/year for the non-digital elements. In both cases the relation between minimum and maximum values assigned was 20.

The sensivitivy analysis assumed an average failure rate of 4 f/year for the non-digital components. This is an average calculated from [8] . The MTTF of digital components is approximately 8000 hours i.e., an average failure rate of 1 f/year.

The estimates and variances for both reliability indices for different failure rates of the digital components are listed in table XI. It is observed that an increase of 20 times the failure rate of the digital components would cause an increase of 2 and 20 times of the probability and frequency of failure estimates respectively. A similar analysis for the failure rate of the non-digital components gave almost constant values for the estimate and variance of both reliability indices. These values were equal to the ones presented in table XI for a failure rate of 1 f/year. The higher effect of the digital components, despite its smaller failure rate, is due to the greater influence of the firstorder cut-set produced by component 12. Similar results were found when the failure rate of both types of components were varied simultaneously.

6.8.2 Repair rate

In a test similar to that for failure rate, the mean time to repair of the components were given selected values between 1 and 20 hours. The average values are the same as used in the previous test.

The estimates and variances are presented in table XII. In this test the repair rate of both digital and non-digital components were varied simultaneously. It is observed that an increase of 20 times the repair time causes an increase in the probability of failure but has no influence in the frequency of failure. This index presents almost a constant value of approximately 1 encounters/year with a standard deviation of 0.07 encounters/year.

Tables XI and XII indicate that a variation of the mean time to repair of the components has greater effect on the probability of failure than a similar variation of its failure rate, the effect being reversed on the frequency of failure.

6.8.3 Coefficient of dispersion

This parameter refers to the quotient of the standard deviation and the estimate of the reliability indices. The test was performed to see what influence different grades of data uncertainty has on the dispersion of the reliability indices. The variable parameter was the sample size from which the component data has been obtained. Selected values between 4 and 32 were used for this parameter.

Table XIII shows the coefficient of dispersion of the estimates

of both reliability indices for different sample sizes of both components. For both indices, the more stable the component data the smaller the dispersion of the estimates, the effect being greater for the frequency of failure.

6.8.4 Confidence limit

An upper confidence limit was found for both reliability indices using the technique presented in section (5.8). The sample size of the components was varied between 4 and 32 for both types of components.

Table XIV presents a 90% upper confidence limit for the probability and frequency of failure for different sample size of the digital and non-digital components. Both results indicate that the greater the sample size the smaller the upper limit defined by Chebyshev's inequality.

6.8.5 Sensitivity coefficients

The results presented in tables XI to XIV give estimate, variance and an upper confidence limit for both reliability indices but they do not give clear information about the effect of an improvement in component rates on system reliability indices.

Table XV lists the sensitivity coefficients of each reliability index with respect to each component parameter of the system. These coefficients were calculated using the technique presented in section 6.3. It is observed in table XV that both reliability indices are increased if the failure rate of any one of the components is increased, the effect being greater from component 12. The probability of failure is reduced by increasing the repair rate i.e., reducing the mean-time-torepair of any one of the components, especially 10 to 12. The frequency of system failure is reduced by increasing the repair rate of components 1 to 11 but is increased if component, 12 increases its repair rate.

6.9 Conclusion

From the different general configurations considered here as alternatives to the single processor-single interface proposed in [7] for the hardware of an integrated digital protection of one corner mesh substation, the duplex system with both processors operating in parallel offers the best solution based on a reliability criterion. A selfchecking and fault-locating facility must be incorporated for maximum availability of the system.

A complete reliability and sensitivity analysis can be done for any proposed configuration using the techniques presented in this thesis. Such analysis should examine the incidence of data uncertainty in the estimates and variances of the indices and find how much confidence should be placed in these estimates. Similarly, it will identify what component improvements result in the best system reliability improvement.

TABLE XI

ESTIMATES AND VARIANCES OF THE SYSTEM RELIABILITY FOR DIFFERENT FAILURE RATES OF THE DIGITAL COMPONENTS $\lambda_{nd} = 4 \text{ f/year}$ Sample size = 16

$r_d = r_{nd} = 5 hrs$

Failure rate	Component availabil-	Probability of failure		Frequency of failure	
1/year	lty	Estimate	Variance	Estimate	Variance
0.1 0.2 0.3 0.4 0.5 0.6 0.7 0.8 0.9 1.0 1.5 2.0	0.999943 0.999886 0.999829 0.999772 0.999715 0.999658 0.999601 0.999544 0.999487 0.999487 0.999480 0.999145	0.609x10 ⁻⁴ 0.122x10 ⁻³ 0.183x10 ⁻³ 0.244x10 ⁻³ 0.304x10 ⁻³ 0.365x10 ⁻³ 0.426x10 ⁻³ 0.487x10 ⁻³ 0.548x10 ⁻³ 0.609x10 ⁻³ 0.913x10 ⁻³	0.434×10^{-9} 0.174×10^{-8} 0.391×10^{-8} 0.694×10^{-8} 0.109×10^{-7} 0.156×10^{-7} 0.278×10^{-7} 0.351×10^{-7} 0.433×10^{-7} 0.974×10^{-7} 0.173×10^{-6}	0.10009 0.20012 0.30016 0.40021 0.50028 0.60036 0.70045 0.80056 0.90068 1.00081 1.50166 2.00284	0.667×10^{-3} 0.267×10^{-2} 0.600×10^{-2} 0.107×10^{-1} 0.167×10^{-1} 0.240×10^{-1} 0.326×10^{-1} 0.426×10^{-1} 0.539×10^{-1} 0.665×10^{-1} 0.150 0.265
2.0	0.990000	0.122810		2.00204	0.209

TABLE XII

ESTIMATE AND VARIANCE OF THE SYSTEM RELIABILITY FOR DIFFERENT MEAN TIME TO REPAIR OF THE COMPONENTS

λ _{nd}	=	4 f/year	r _{nd} = 5 hrs	
λ _d	•	l f/year	sample size = 1	.6

Component	Probability of failure		Frequency of failure	
(hrs)	Estimate	Variance	Estimate	Variance
1 2 5 10 20	0.122×10^{-3} 0.244×10^{-3} 0.609×10^{-3} 0.122×10^{-2} 0.244×10^{-2}	0.174×10^{-8} 0.694×10^{-8} 0.433×10^{-7} 0.173×10^{-6} 0.689×10^{-6}	1.00024 1.00038 1.00081 1.00156 1.00321	0.666×10^{-1} 0.666×10^{-1} 0.665×10^{-1} 0.664×10^{-1} 0.661×10^{-1}

TABLE XIII

COEFFICIENT OF DISPERSION OF SYSTEM RELIABILITY INDICES FOR DIFFERENT SAMPLE SIZES OF THE COMPONENT DATA

$$\lambda_d = 4 \text{ f/year}$$
 $r_{nd} = r_d = 5 \text{ hrs}$

$$\lambda_{nd} = 1 \text{ f/year}$$

Sample size	Probability of failure % c.d.	Frequency of failure % c.d.
4	61 .2	57.7
8	46.7	37•7
16	34.2	25.8
32	24.6	17.9

TABLE XIV

90% UPPER CONFIDENCE LIMIT OF RELIABILITY INDICES FOR DIFFERENT SAMPLE SIZES

$$\lambda_{nd} = 4 f/year$$
 $\lambda_d = 1 f/year$

 $r_{nd} = r_d = 5 hrs$

Sample size	Probability of failure	Frequency of failure
4	0.284×10^{-2}	3.580
8	0.202×10^{-2}	2.689
16	0.154×10^{-2}	2.154
32	0.124×10^{-2}	1.803

TABLE XV

SENSITIVITY COEFFICIENTS FOR THE PROBABILITY AND FREQUENCY OF FAILURE FOR THE SYSTEM IN FIG. 4

> $\lambda_{nd} = 4 \text{ f/year}$ $\lambda_d = 1 \text{ f/year}$ $r_{nd} = r_d = 5 \text{ hrs}$ sample size = 16

Component	Probability of failure		Frequency of failure	
	Failure rate	Repair rate	Failure rate	Repair rate
1-3	0.000044	-0.000044	0.000140	-0.000093
4-9	0.000011	-0.000011	0.000035	-0.000023
10-11	0.000534	-0.000534	0.001139	-0.000569
12	0.936424	-0.936424	0.998054	0.000570

CHAPTER VII

CONCLUSIONS AND SCOPE FOR FUTURE RESEARCH

7.1 Use of data validation techniques in digital relaying systems

Power system protection with on-line digital computers requires a maximum availability of the data used for protection decisions. Protection statistics have shown that the second highest cause of faultclearance equipment can be attributed to the presence of bad data. Data validation techniques possess the ability to detect and identify missing data and large unexpected errors in data points. These properties can easily be achieved by checking the measurements themselves, their differences against chosen bounds and cross-checks between analogue (current or voltage) and digital (switch position) values. Once errors have been detected and identified, a simple estimation routine based on any type of available redundancy can be used to improve the reliability of the relaying data base.

7.1.1 Zero data detection

Loss of data is a critical factor on the reliability of the relaying data base. Zero data detection by software requires consideration of the conversion tolerance of the analog-digital converters. The time taken by such checks will consume most of the time interval available between successive samples. Zero data detection by hardware is shown to be essential in the practical implementation of a relaying data validation algorithm.

7.1.2 Doubtful data identification

The use of a logic 1 as zero and error flags incorporated in the data-word structure was found to be extremely useful in the validation
procedure to optimise the time spent in data identification. Such a facility adds one and two extra bits to the word structure of switch and analogue data respectively. These flag informations ease the development of a systematic estimation procedure for the best use of the redundancy available in the measurement set.

7.1.3 Application to one-corner of a mesh substation

The critical factor in the design of a relaying data validation algorithm is the short time available between samples. If high sampling rates e.g. 32 samples per cycle, are required in a scheme similar to the computer-substation model described in chapter II, no valuable checks can be implemented and missing data may deteriorate the performance of the protection scheme. However the data validation algorithm and the estimation procedure proposed in the same chapter can be successfully applied in such a scheme using a maximum sampling rate of 16 samples/cycle. Other problems such as core storage is not critical. Use of assembly language was found to be essential for the practical implementation of any relaying data validation algorithm.

Special care must be taken in the design of the interface structure (sample-hold circuits, A/D converters and multiplexers) to minimize the input and conversion time, maximizing the time available for validation. Packaging group of switch positions in one data-word also contributes to this objective.

Off-line and on-line test results indicate that the check on two complementary bits representing each switch position provide a reliable method for the detection of any doubtful switch indication. The use of a sample-switch routine proved to allow correction of such doubtful indications. On-line results confirmed the reliable performance of the

zero data detection by hardware. Test results also showed that the use of a residual value and a residual counter work satisfactorily in the detection of missing data. The tolerance limit to be used requires special consideration of any unbalance present either in the power system or in each data acquisition link.

7.2 Reliability evaluation of complex system

The reliability evaluation of complex systems with unreliable branches and nodes is only feasible using the tie-set or cut-set approaches described in chapter III. For systems with an average node degree (elements incident on the node) greater than four and/or components of high reliability the cut-set approach is more efficient than the tie-set technique. Bounds obtained using the inclusion-exclusion principle have smaller errors than those obtained from the disjoint or independence approximation for both tie-set and cut-set methods.

7.3 Minimal cut-set enumeration

The enumeration of the minimal cut-sets of all nodes in a separable, nonseparable or acyclic directed graph can be solved with one application of the appropriate algorithm described in chapter IV. The combination of the novel concept of the cut-node incidence matrix and sparsity techniques produces very efficient algorithms (fast and small storage requirements) for complex systems.

7.4 Node failure problem in power systems

Unreliable nodes were shown to have considerable influence on the reliability indices of each load point in an electric power system, even for highly reliable nodes. The branch-node cut-set algorithm

described in chapter IV is computationally efficient and combined with the minimal cut-set formulae developed by Billinton and Grover⁽⁷⁹⁾ provides a sequential method for performing complete reliability calculations in distribution and transmission systems.

7.5 Probability and frequency of system failure

The new statistical formulation of the probability and frequency of system failure provides extra information (variances and confidence limits) which enables the system designer to evaluate the goodness of his design. The variance equations developed in chapter V can be an aid to the designer in determining the causes of excessive reliability index variance since he can study these equations and determine which $(\partial u/\partial x_i)^2 \sigma_{x_i}^2$ terms are contributing most heavily to index variance. He can decide whether to reduce the large terms by using more stable components (reduction of $\sigma_{x_i}^2$) or by modifying the system configuration in such a way that the sensitivity of the reliability index to a particular component parameter is reduced (reduction of $\partial u/\partial x_i$). Similarly the sensitivity coefficients presented in chapter VI provide useful information about where to locate redundancy for the greatest increase of the system reliability.

The moment approach and the use of Chebyshev's inequality presented in Chapter V always give more conservative results than the Monte Carlo technique but the agreement improves for highly stable components. The advantage of the former method is in a one-experiment constant-time solution whilst the computation time of the simulation procedure increases with the component sample size and the number of trials.

7.6 Reliability considerations in the design of digital protection systems

The double nature of protection functions requires the evaluation of two performance indices, the probability of failure to operate in the presence of a fault and the security of the system against false operation. Moreover a proper evaluation also involves the determination of the frequency of such incidents. At the moment it appears that the only meaningful approach for the reliability assessment of any computer-based protection scheme is on the average value of failure and repair rates available for different types of digital components. In this sense the methods presented in this thesis represent a systematic approach for the comparative evaluation of the reliability of different alternatives. The same techniques are applicable if failure and repair rates are available for each type of incident and the analysis is based on the steady state values of the indices. However it remains to be demonstrated that the solution adopted for the hardware and software structure of a computer-based protection scheme will have a level of reliability and security better or worse than that achieved in conventional schemes.

7.7 Future work

1. Data validation

Research is required on the feasibility of a data validation structure in a digital micro-processor based relay.

2. Minimal cut-set enumeration

An algorithmic procedure is required for the enumeration of the minimal cut-sets of complex systems with an r-out-of-n configuration.

3. Variance and confidence limits

Further investigations into the Moment method approach is needed for the case of component data obtained from small sample sizes.

4. Sensitivity coefficients

Research on the extension of the method is needed so that when the system reliability indices are at a certain level the overall system cost is minimized.

5. Digital protection reliability

Further investigation is required on the determination of the resultant reliability of a computer-based protection system taking into account both the reliability of the operation in the presence of a fault and the reliability of the system against false operations.

7.8 Original contributions

- i. the development of a data validation algorithm for the detection and identification of gross errors (missing or erroneous data) and the on-line testing of its performance using a laboratory computer-based substation model.
- ii. design of an estimation procedure to provide replacements
 for loss data based on the results of the data validation
 algorithm and using either the inherent redundancy of
 electrical networks or redundant measurements.
- iii. development and testing of algorithms for the enumeration of the set of minimal cuts of separable, nonseparable and acyclic directed graphs based on graph theory concepts.
 - iv. inclusion of the node failure problem in the reliability evaluation of distribution and transmission systems using a new tested branch-node minimal cut-set algorithm.
 - v. development and testing of two new approaches for the variance and confidence limits of the probability and frequency of failure of repairable systems using a Moment method or Monte Carlo simulation.

vi. formulation of the sensitivity of the probability and frequency of system failure to component failure and repair rates by means of sensitivity coefficients to be used in the redundancy location problem.

APPENDIX 1

1.0 <u>Minimal path examples of table II - Chapter III</u> : All sets of paths have been randomly generated.

Notuonk	Minima:l pat	hs
Network	Binary representation	Base 10
1	0 1 0 1 0 1 1 0 1 0 1 0	5 6 10
2	0 0 1 0 1 0 1 0 1 0 1 0 0 1 0	5 10 18
3	1 1 0 1 0 0 1 1 0 0 1 0 1 0 1 0 1 0 1 0 1 0	52 50 42 41
4	1 0 1 0 0 0 1 1 0 0 1 1 0 1 1 0 0 1 0 1 0 1 0 1 1 0 0 1 0 1 0 0 1 0 1 0 1 0 0 1 0 1 0 1 0 0 0 1 0	81 77 74 57 37 34
5	0 0 1 0 1 0 0 1 0 1 0 1 0 1 0 0 1 1 0 0 1 0 1 0 0 0 1 1 1 1 0 1 0 1 0 0 0 0 1 1 1 0 1 1 1 0 0 0 1 0 0 1 0 0 1 1	20 85 25 35 106 14 113 19
6	1 0 0 1 0 0 0 1 0 1 0 1 0 0 1 0 1 0 1 0 0 1 0 1 0 0 1 0 1 0 1 0 0 0 1 0 1 1 1 0 0 0 0 1 0 0 1 0 0 0 1 0 0 0 1 0 0 1 0 0 0 0 1	72 84 82 81 56 36 34 33

2.0 Edge listings of graphs in Table IV - Chapter IV

Graphs 1 and 4 have been taken from [32] and [30] respectively. Graphs 5, 6 and 7 are complete graphs. Graph 9 corresponds to the AEP 14 busbar test-system. All other graphs have been randomly generated.

	<u>Graph 1</u>		G	caph 2	
Е	SB	EB	E	SB	EB
1	1	3	1	1	2
2	3	4	2 3	1	23
3	4	7	4	2	4
4	7	8	5	2 3	4 4
5	1	2	7	4	5
6	2	5	8	4	5
7	5	6	10	4	7
8	6	8	11	4	6
9	3	5	13	4. 6	8
10	5	7	14	6	8
	-	·	15 16	7	9
			17	8	10
			18 19	8 9	12
			20	9	13
			21	10	13
			22	11	14
			25	13	15
			25	14	15
			26	15	16
			27	16	20
	•		28	15	17
			29	15	20
			50 31	18 18	20
			J± 32	15	19
			33	19	20

Е	SB	EB
1 2 3 4 5 6 7 8 9 10	1 1 2 2 3 4 5 4	22 22 33 44 45 57 77
11	4	6 8
13	6	8
14	6	8
15	7	9
10	1 8	ל חו
18	8	11
19	9	12
20	9	13
21	10	13
22	11	14
23	12	15
24	13	15
25	14	- 15

Gı	aph 4	
E	SB	EB
1 2 3 4 5 6 7 8 9 10 11 12 13 14 15 16	1112234454666477	2 2 3 4 4 5 5 6 6 8 8 7 8 8 7 8
	•	-

	•	
	Graph 5	·
E	SB	EB
1 2 3 4 5 6 7 8 9 0 1 1 2 3 4 5 6 7 8 9 0 1 1 2 3 4 5 6 7 8 9 0 1 2 3 4 5 6 7 8 9 0 1	1 1 1 1 1 2 2 2 2 2 2 3 3 3 3 4 4 4 5 5 6	2 34 56 7 3 4 56 7 4 56 7 56 76 77
म	<u>Graph 6</u> SB	EB
1	1	2
1234563890112345678901222222222222222222222222222222222222	111112222233334445561234567	234567345674567567677888888888

<u>Graph 7</u>

Е	SB.	EB
1234567890112345678901123456789012222222222223333333333333333333333333	11111122223333444556123456712345678	234567345674567567677888888889999999999

· <u>·</u>	raph o	
E	SB	EB
- 1234567890112345678901223456789012334567890123444444444	11122344544466778899011234565775859000222222222222222222222222222222222	2 2 3 4 4 4 5 5 7 7 6 8 8 8 9 9 10 11 2 3 3 4 4 5 5 7 7 6 8 8 8 9 9 10 11 2 3 3 4 4 5 5 7 7 6 8 8 8 9 9 10 11 2 3 3 4 4 5 5 10 2 17 2 18 2 19 20 12 2 2 4 4 5 8 3 8 8 2 6 7 8 2 10 2 1 2 2 2 4 4 5 8 3 8 8 6 7 8 2 10 2 1 2 2 2 4 4 5 8 3 8 8 6 7 8 2 10 2 10 2 1 2 2 2 4 4 5 8 3 8 8 6 7 8 2 10 2 10 2 1 2 2 2 4 4 5 8 3 8 8 6 7 8 2 10 2 10 2 1 2 2 2 4 4 5 8 3 8 8 6 7 8 2 10 2 10 2 10 2 10 2 10 2 10 2 10 2
41	41	20

	Graph 9	
E	SB	EB
1	1	2
2	· 1	3
3	2	3
4	2	6
5	. 3	4
6	3	5
7	3	6
8	4	5
9	5	6
10	5	8
11	5	9
12	6	7
13	7	11
14	7	12
15	7	13
16	8	·9
17	9	10
18	9	14
19	10	11
20	12	13
21	13	14

.

.

.

۰.

3.0 Edge listing of graphs in table VIII and XVI - Chapter IV

Graphs 10 and 11 have been randomly generated. Graph 12 corresponds to the West-Venezuelan 115 kV system.

	Graph 10	<u>)</u>		Graph 1	1
E	SB	EB	E	SB	EB
1	1	10	1	1	2
2	1	5	2	1	3
3	10	7	3	2	4
4	10	2	4	2	10
5	5	2	5	3	10
6	2	7	6	10	4
7	-2	8	7	10	ģ
8	-2	.8	8	10	9
9	-7	9	9	-4	5
10	-7	3	10	4	7
11	-8	6	11	9	7
12	-9	4	12	-5	6
	-		13	-7	8
			14	-8	11

Graph 12

Е	SB	EB
1	1	2
2	1	4
3	1	5
4	1	7
5	1	13
6	2	3
7	3	á
8	á	5
ģ	5	6
10	, ,	7
1 1 1	2	8
10	0	
12	0	7
1)	-9	10
14	9	11
15	-11	12
16	11	13
17	-2	14
18	-14	15

APPENDIX 2

This appendix includes the detailed discussion and proof of each algorithm presented in chapter IV. First two general theorems are introduced whose proofs can be found in the reference quoted in each case. Then it follows the proofs specially developed by the author for the new minimal cut-set generation algorithms presented in sections 4.5, 4.6 and 4.7 of chapter IV.

A2.1 General theorems

<u>Theorem 1⁽³²⁾</u>: The removal of the components in a minimal cutset between nodes s and t separates the graph into exactly two connected subgraphs G_s and G_t where G_s contains node s and G_t contains node t.

<u>Theorem 2⁽⁴²⁾</u>: The ring sum of any two cut-sets in a connected graph is either a third cut-set or an edge disjoint union of cut-sets.

A2.2 Discussion and proof of the algorithm for non-separable graphs of section 4.5

This section justifies the steps of the algorithm and proves that it generates the complete set of minimal cuts between the reference node and all other nodes of a nonseparable graph.

The procedure begins generating the basic minimal cut-set with the set of edges incident to the reference node. Each time a new node is scanned the algorithm automatically generates a minimal cut-set with the set of edges incident to the node being scanned. Steps 1 and 4 of the algorithm are justified by : <u>Theorem 3</u> : In a nonseparable graph G the set of edges incident on each node of G is a minimal cut-set.

Proof : To prove this lemma one should note that the removal of all edges incident to node i breaks all paths from the reference node r to node i. Then this removal separates the graph into at least two connected subgraphs, one contains node i and the other containsnode r. Now we prove, by contradiction, that the removal of the edges incident to node i separates G into exactly two connected subgraphs G_1 and G_2 where G_1 contains only node i and G_2 contains all other nodes. Assume that there are more than two connected subgraphs. Then there must be a connected subgraphs G_{χ} that includes neither i nor r. As G is connected there must be at least one path in G between r and each node in G3. Since the only edges removed are the ones incident to node i, if G_3 is disconnected from G_2 all paths from G_2 to G_3 must contain one of the removed edges. Then the original graph is separable because all paths from the reference node to any node in ${\rm G}^{}_{\rm J}$ pass through node i i.e., graph G has node connectivity one. But this is contrary to the original assumption that G is nonseparable. Then the set of edges incident to node i separates the graph into exactly two connected subgraphs G1 and G2. Hence by Theorem 1 the Theorem is proved.

Corollary 1 : The minimal cut-set formed by the set of edges incident to node i is a minimal cut-set only for node i.

Corollary 2 : The minimal cut-set formed by the set of edges incident to the reference node is a minimal cut-set for all other nodes of the graph.

Corollary 1 and 2 justify the construction of the cut-node vectors generated by steps 4 and 1 of the algorithm respectively.

So far we have generated only a set of minimal cuts constituted by the proper minimal cuts of all nodes of the graph including the reference node. Step 5 suggests additional cut-sets are obtained from the ring sum operations between two or more of these cut-sets. Theorem 2 indicates that the ring sum of any two cut-sets in a graph is either a third cut-set or an edge-disjoint union of cut-sets. However we need to prove that the set of proper cuts is a fundamental set of minimal cuts from where all other cut-sets of a graph can be obtained.

<u>Theorem 4</u> : The set of minimal cuts corresponding to the set of (n - 1) proper minimal cuts of a graph of n nodes forms a fundamental set for the generation of the complete set of minimal cuts between the reference node and all other nodes of the graph. This set includes the proper cut of the reference node.

Proof: By Theorem 1 the problem of generating the set of minimal cuts between the reference node r and all other nodes of a graph is the same problem that generates all the partitions of the set N of nodes into the subsets Y and \overline{Y} . The set Y must define a connected subgraph that includes node r. The set \overline{Y} must define a connected subgraph that includes at least one of the other nodes of the graph. There are 2^n ways to partition N into two mutually exclusive subsets Y and \overline{Y} in a connected graph. Since r is restricted to be a member of Y the number of partitions is reduced to 2^{n-1} . Every such partition of a complete graph is a minimal cut-set. Then we shall prove that the set of (n - 1)proper cut-sets is a basis for the complete set of minimal cuts of a graph by showing that every partition of the set N in which node r and at least one of the other nodes are separated can be obtained from the set of subgraphs representing (n - 1) proper cut-sets of the graph.

By Corollary 1 of Theorem 3 each proper cut-set j is a subgraph g_j which can be represented by an n-tuple(an ordered array of n elements)

$$x_{j} = (x_{1}, x_{2}, \dots, x_{n})$$

such that

 $x_i = 1$ if node i is in g_j and $x_i = 0$ if node i is not in g_j

Then the set of subgraphs representing the proper cut-sets of a connected graph of n nodes are

 $X_{1} = 100.^{n}.0$ $X_{2} = 010.^{n}.0$ $X_{n} = 000.^{n}.1$

Consider an arbitrary partition N_1 of the set N such that

 $N_1 = O_1 I_2 \cdots O_j I_k O_l I_m O_n$

where nodes 2, k and m form the subset Y of N and all other nodes form subset \overline{Y} . The n-tuple representing N₁ can be seen to correspond to the ring sum of the n-tuples X₂, X_k and X_m. Then any of the 2ⁿ⁻¹ partitions which define subsets Y and \overline{Y} can be obtained by the ring sum of the n-tuples representing the proper cut-sets of the nodes in subset Y. Besides the n-tuple X_n define a partition equivalent to the one given by the ring sum of all other (n - 1) n-tuples therefore only (n - 1) proper cut-sets are necessary to generate the complete set of 2^{n-1} partitions. Hence the theorem is proved and justifies step 2 of the algorithm. Obviously only for a complete graph is it necessary to enumerate the 2^{n-1} partitions because each one corresponds to a minimal cut-set. Then an efficient algorithm will not enumerate 2^{n-1} cases unless, of course, there are 2^{n-1} minimal cuts.

From the definitions of ring sum operation of two graphs and edgedisjoint events the ring sum of two cut-sets with no edges in common is always an edge-disjoint union of cut-sets. This property can be used to discard all the ring sum operations which positively produce an edge-disjoint union of cut sets. This reduction is achieved if only the ring sums between two cuts that have at least one edge in common are considered during the cut-set generation procedure. It has been provided in step 3 of the algorithm by the index $CI_k(i)$ associated with the ith cut-set when scanning node k.

However when scanning node k the ring sum of the proper cut-set with any of the cut-sets with non-zero index may also give, by Theorem 2, an edge-disjoint union of cut-sets. The algorithm must provide the steps to identify these edge-disjoint events.

Lemma 1 : The possible edge-disjoint union of cut-sets generated by the algorithmic procedure when scanning node k is always contained in one of the minimal cut-sets generated during the same scan.

Proof: Note that the search of the nodes is made starting by the reference node r and once the basic minimal cut-set is generated we move to an adjacent node v. At v we form the proper cut-set of v and make the ring sums with all previous cut-sets which have at least one edge in common. Once we do all ring sums we move then to one of the unscanned nodes closest to r. This process is continued till (n - 1) nodes in the graph are considered. This method of searching is the one responsible that only

(n - 1) nodes need to be scanned because the proper cut-set of the nth node has been automatically generated from the ring sum of the (n - 1) proper cuts. At the same time this way of searching guarantees that each time a ring sum of the kth proper cut is made with a cut-set of non-zero index, the resulting set of edges (cut-set or edge-disjoint union of cut-sets) have at least one edge in common with the kth proper cut-set which is not contained in any of the minimal cut-sets generated when scanning the previous (k - 1) nodes. Then an edge-disjoint union of cut-sets generated from the ring sums of the kth proper cut contain one or more edges in common with this cut-set which are not present in any of the cut-sets generated before the scan of node k. Since by definition any edge-disjoint union of cut-sets can be decomposed into two mutually exclusive cut-sets, the one which includes the edges in common with the kth proper cut must correspond to one of the minimal cut-sets generated during the scan of node k. Hence the lemma.

This check is provided by step 5 of the algorithm and reduces the number of comparisons to a minimum, increasing the efficiency (by reducing the computation time) of the algorithm.

So far, only the incidence of the basic minimal cut-set and the proper cut-sets on the nodes of the graph have been defined. By definition the cut-node vector associated with each cut-set tells us which nodes are affected for that cut i.e., it contains the nodes that we have associated with subset \overline{Y} . The cut-node vector for all other cut-sets is generated as follows:

Lemma 2 : A cut-set generated from the ring sum of the basic minimal cut-set and the jth proper cut-set is not a cut-set for node j.

Proof: From Corollary 2 of Theorem 3 the subsets Y and \overline{Y} from the basic cut-set are $Y_1 = 1$ and \overline{Y} contain all other nodes. Now, the basic

cut-set and the jth proper cut-set are combined (ring sum) only if they have at least one edge in common. Since the ring sum operation discard all the edges in common the resulting set of edges cannot be anymore a minimal cut-set for node j because the edge(s) discarded provide a path(s) from the reference node to node j. For the new set of edges Y = 1,j and \overline{Y} all other nodes.

Lemma 3 : A cut-set generated from the ring sum of two proper cut-sets is a cut-set for both nodes.

Proof: From Corollary 1 of Theorem 3 the jth and kth proper cut-sets are cuts only for nodes j and k respectively i.e., $\overline{Y}_j = j$ and $\overline{Y}_k = k$. In the procedure they are combined only if they have at least one edge in common. Since each edge has only 2 terminals, any edge in common between the two cut-sets must have j and k as terminal nodes. Then the set of edges from their ring sum still breaks all paths from the reference node to nodes j and k. For the new set of edges $\overline{Y} = j$ and k and Y contain all other nodes.

Therefore the algorithmic procedure presented in this section does represent an efficient and systematic way to generate the complete set of minimal cuts between the reference node and all other nodes of a nonseparable graph.

A2.3 Discussion and proof of the algorithm for separable graphs of Section 4.6

The algorithm for separable graphs with some of its subgraphs with tree configuration results from slight modifications introduced to the algorithm presented for nonseparable graphs. However some of the Theorems and Lemmas presented in section (A2.2) are not directly applicable

for separable graphs. This section will justify the steps of the algorithm and prove that it does generate the complete set of minimal cuts between the reference node and all other nodes of separable graphs as defined in section (4.6.1).

As before the process starts generating the basic minimal cut-set with the set of edges incident to the reference node. Since the systems analysed by the algorithm are considered to be connected, the reference node as defined in this thesis cannot be an articulation point. Then the removal of the edges incident on the reference node r must separate the graph into two connected subgraphs, one contains r and the second all other nodes of the system. Otherwise node r is an articulation point. Hence by Theorem 1 the set of edges incident to node r is a minimal cut-set and affects all other nodes of the graph. Hence step 1 of the algorithm is justified. All other proper cut-sets enumerated in step4. of the process are justified below.

<u>Theorem 5</u> : In a separable graph the complete set of edges on an articulation point are not a minimal cut-set.

Proof: To prove the Theorem let A be the set of all edges incident on the articulation point v of graph G. If A is a minimal cut-set then by Theorem 1 of Appendix 2 the removal of A from G must separate the graph into exactly two connected subgraphs G_1 and G_2 . Since A contains all edges incident on v, if subgraph G_1 contains only node v then G_2 must contain all other nodes. Now, by definition a connected graph is said to be separable if there exists a subgraph g in G such that \overline{g} (the complement of g in G) and g have only one node in common. Then all paths between any node in g and a node in \overline{g} must pass through

this common node. Such common node is what is called an articulation point. Since G_1 contains only node \mathbf{v} , if \mathbf{v} is an articulation point the removal of A must define a third subgraph G_3 such that all paths from G_2 to G_3 pass through \mathbf{v} . Then by Theorem 1, the set A is not a minimal cut and the Theorem is proved.

From Theorem 5 it can be concluded that only a subset of the edges incident to an articulation point will be a minimal cut-set between that point and the reference node.

Let G1 and G2 the two subgraphs of G defined by an articulation point v. If subgraph G_1 contains the reference node then we have assumed that G₂ has a tree configuration where the set of edges with v as terminal will form the root of the tree. Then all cut-sets between the tree nodes (internals or pendants) and the reference node will also be cut-sets between the articulation point and the tree nodes. Since there is one and only one path between every pair of nodes in a tree, there is only one path (see Theorem 6) between the articulation point and each one of the other nodes of the tree. This one-to-one correspondence suggests to solve the cut-set enumeration problem by representing each subgraph G₂ by a directed subgraph whose edges are oriented moving away from the articulation point. With this convention the subset of edges in the tree connected to the articulation point will be oriented incident out of the articulation point. Then only the set of undirected edges incident on the articulation point will form a cut-set for that point. Similarly only the set of edges incident into a tree node will separate the node from the articulation point dividing the subgraph into two connected subgraphs. Then this set of edges is a minimal cut-set for the node tree. This explains the generation of the proper cut-sets by step 4 of the algorithm. Moreover all cut-sets

of this tree will be disjoint events respective of any other cut-set of the graph, then there is no need to make the ring sums with the previous cut-sets as indicated in step 4. This disjoint property also justifies step 2. However if one restricts that the last unscanned node will never be a tree node, only NN - 1 nodes need to be scanned by the procedure.

The subgraph which includes all nodes between the reference point and each articulation point is a nonseparable graph therefore for all these nodes the algorithms steps were justified in section (A2.2). The only difference is when constructing the cut-node vectors of the cut-sets between the reference node r and an articulation point w. In this case, as mentioned before, each cut-set will separate the graph into two connected subgraphs. One subgraph will include the reference node and the other will always include w and all nodes whose paths from r pass through w. Then the cut-node vector generation follows from Lemmas 2 and 3 if we note that the presence or absence of a path between r and w implies also the connection or disconnection respectively of all tree nodes connected to the articulation point.

<u>Theorem 6</u> : In a tree every node with degree greater than one is an articulation point.

Proof: A node with a degree greater than one will have at least two incident edges. Then these two incident edges will have one common node and two non-adjacent nodes. A tree by definition is a connected graph without circuits. The union of two different paths between two nodes will always form a circuit. Then there is one and only one path between every pair of nodes in a tree. Therefore the removal of a node tree with degree greater than one will disconnect every pair of non-

adjacent nodes of the edges incident to that node and therefore will disconnect the tree. Hence the theorem is proved.

By Theorem 6 the construction of the cut-node incidence vector for each internal node of a tree is justified as mentioned for the case of an articulation point. Clearly the proper cut-set of a pendant node will affect only that node.

The fundamental set of minimal cuts used by the algorithm is the group of proper cut-sets of the nodes of the graph. Then by Theorem 4 the set of (n - 1) proper cut-sets in an n-nodes graph does generate the complete set of minimal cuts. As mentioned before this fundamental set must include all proper cut-sets of any tree node in the graph.

A2.4 Discussion and proof of the algorithm for directed acyclic graphs of section 4.7

The method is based on a combinational approach to the problem of enumerating minimal cut-sets rather than a purely mathematical one. The requirement for a combination of edges to be identified as an s-t directed cut-set is that this set of edges must break all directed paths between s and t. However the procedure instead of enumerating all directed paths and after solving the problem by a combinational approach (30), it searches directly for all the edges combinations that break the s-t paths as it moves from node to node.

It is evident that if we must satisfy that each combination of edges break all s-t paths, one needs to start the combination procedure with one set that holds that property. Since all directed paths between an origin node s and a terminal node t must include at least one edge incident out of s and one edge incident into t, the set of edges incident out of s is an s-t minimal cut-set. Step 1 of the algorithm constructs this cut-set and uses it as the basis to generate all other

combinations as follows. Once this cut-set is formed the procedure moves to an unscanned node w adjacent to the origin. This is achieved by step 3 that uses an index cut to search for adjacent nodes. Once a node w we scan all edges incident on w. Each cut-set already in the list will form a new cut-set replacing the edges incident into w that are contained in the cut by the set of edges going out of w. To be sure that all nodes nearest than w to the origin have been scanned all the edges incident into w must be already scanned by the algorithm and therefore contained in at least one of the cut-sets previously generated. This also ensures that all possible directed paths passing through w have been implicitly constructed by the algorithm so any combination with the edges leaving w will break all s-t paths. Once all combinations have been made with the edges incident out of w, we move to an adjacent node v. At v we scan again all edges connected to v and make new combinations with the edges going out of v. This process is continued until all edges incident out of any node in the graph are scanned. Since node t has only edges leading into t the procedure stops when (n - 1)nodes of a graph of n nodes have been searched (step 2).

Each new combination of edges is a cut-set because starting from a cut-set all that it does is to replace edges for those that generate the directed paths that pass through the node under scan. Unfortunately no all cuts generated by the procedure need to be minimal therefore if a new combination of edges is contained or contains one of the previous cut-sets, only the combination with minimum number of elements is kept as a cut-set. Note that each new combination contains all the edges incident out of the node under scan and none of these edges can be present in any one of the cuts present in the list before the actual scan is started. Otherwise it is incident out from two different nodes.

This reduces the comparisons at minimum because each new combination of edges needs to be compared only with the cut-sets generated when scanning the same node (step 4).

REFERENCES

- HECHT, H. and GRONBERG, G. "The use of on-line minicomputers in Swedish municipal power distribution systems", Symposium on implementation of real-time control by digital computer, paper M, Imperial College, London, September 1973.
- 2. WIMMEL, K.H. and ZOLLENKOPH, K. "Experience in the use of on-line digital computer for switching a 380/110 kV system", Symposium on implementation of real-time control by digital computer, paper N, Imperial College, London, September 1973.
- 3. COUCH, G.H. and MORRISON, I.F. "Data validation and topology determination for power system monitoring and control", IEEE Conference paper No. C74 361-2, presented at the Summer Power Meeting, July 1974.
- 4. BREWER, G.L. and TAM, C.C. "The computational aspects of the CEGB new national control centre", 3rd PSCC Proceedings, paper STY-5, June 1969.
- PULSFORD, H.E., LAST, F.H., and LOWE, B. "On-line facilities in the national control centre of the Central Electricity Generating Board", CIGRE paper No. 32-12, 1972.
- 6. MURRAY, B.E. and DROMEY, G. "Practical design considerations affecting the use of digital computers in high voltage substations", 4th IFAC/IFIP International conference on digital computer applications to process control, Part II pp. 292-303, March 1974.
- 7. HORNE, E. and CORY, B.J. "Digital processors for substation switching and control", IEE Conference on developments in power system protection, Conference publication No. 125, March 1975.
- 8. EDGLEY, R.K. "Back-up protection and overall protection of power system networks" Electra-CIGRE, No. 34, May 1974.

- 9. WARRINGTON, A.R. Van C. "Protective relays: Their theory and practice", Volume 1, Chapman and Hall Ltd., London 1968.
- 10. HARDER, E.L. and CUNNINGHAM, J.C. "Relay and circuit breaker application". Electrical transmission and distribution reference book, Chapter 11, Westinghouse Electric Corporation, 1950.
- 11. SEYMOUR, C.R. "Statistics of CEGB protection performance 1968-73", IEE Conference on developments in power system protection", Conference publication No. 125, March 1975.
- 12. GONZALEZ, G.E. "Relaying data validation". Digital protection of power systems - Post experience course, Imperial College, London, March 1975.
- 13. MINIESY, S.M. "Integrated digital protection and control of high voltage substations", Power system report No. 94, Imperial College, London September 1973.
- 14. HORNE, E. "Interface reference manual", Unpublished report, Imperial College, London, November 1974.
- 15. MOORE, E.F. and SHANNON, C.E. "Reliable circuits using less reliable relays", Journal of the Franklin Institute, Vol. 262, 1956, pp. 191-208 and pp. 281-297.
- 16. SINGH, C. and BILLINTON, R. "A new method to determine the failure frequency of a complex system", IEEE Trans. on Reliability, Vol. R-23 No. 4, October 1974.
- 17. BAZOVSKY, I. "Reliability theory and practice", Prentice-Hall Space Technology series, New Jersey, 1961.
- 18. SHOOMAN, M.L. "Probabilistic reliability : An engineering approach", McGraw-Hill, New York, 1968.

- 19. DUBES, R. "Two algorithms for computing reliability" IEEE Trans. on Reliability, December 1963.
- 20. WING, O. and DEMETRIOUS, P. "Analysis of probabilistic networks", IEEE Trans. on Commun. Technol., Vol. COM-12, September 1964.
- 21. MISRA, K.B. "An algorithm for the reliability evaluation of redundant networks", IEEE Trans. on Reliability, Vol. R-19, No. 4, November 1970.
- 22. MOSKOWITZ, F. "The analysis of redundancy networks", AIEE Trans. Commun. Electron. Vol. 77, November 1958.
- 23. HENLEY, F.J. and WILLIAMS, R.A. "Graph theory in modern engineering", Academic Press, New York, 1973.
- 24. LEE, C.Y. "Analysis of switching networks", Bell Syst. Techn. J., November 1955.
- 25. BROWN, D.B. "A computerized algorithm for determining the reliability of redundant configurations" IEEE Trans. on Reliability, Vol. R-20, March 1971.
- 26. FRATTA, L. and MONTANARI, U.G. "A Boolean algebra method for computing the terminal reliability in a communication network" IEEE Trans. on Circuit theory, Vol. CT-20, May 1973.
- 27. LIN, P.M., LEON, B.J. and HUANG, T.C. "A new algorithm for symbolic system reliability analysis" IEEE Trans. on Reliability, Vol. R-25, No. 1, April 1976.
- 28. PIERCE, A.R. "Bibliography on algorithms for shortest path, shortest spanning tree, and related routing problems 1956-1974", Networks, Vol. 5, No. 2, April 1975.
- 29. OKADA, S. "Algebraic determination of loop and paths in graphs", the fourth international conference on operations research -Pre-prints of proceedings, International Federation of Operational Research Society, Boston 1966.

- 30. NELSON, A.C. Jr., BATTS, J.R. and BEADLES, R.L. "A computer program for approximating system reliability" IEEE Trans. on Reliability, Vol. R-19, No. 2, May 1970.
- BATTS, J.R. "Computer program for approximating system reliability
 Part II", IEEE Trans. on Reliability, Vol. R-20, No. 2, May 1971.
- 32. JENSEN, P.A. and BELLMORE, M. "An algorithm to determine the reliability of a complex system", IEEE Trans on Reliability, Vol. R-18, No. 4, November 1969.
- 33. BELIMORE, M. and JENSEN, P.A. "An implicit enumeration scheme for proper cut generation", Technometrics, Vol. 12, No. 4, November 1970.
- 34. HANSLER, E., MCAULIFFE, G.K. and WILKOV, R.S. "Exact calculation of computer network reliability" Networks, Vol. 4, 1974.
- 35. BARLOW, R.E. and PROSCHAN, F. "Mathematical theory of reliability", John Wiley and Sons Inc., New York, 1965.
- 36. BARLOW, R.E. and PROSCHAN, F. "Statistical theory of reliability and life testing: Probability models", Holt, Rinehart and Winston Inc., New York, 1975.
- 37. FRANK, H. and FRISCH, I.T. "Communication, transmission and transportation networks", Addison-Wesley, Reading, Mass., 1971.
- 38. WILLIAMS, G. "The design of survivable communication networks", IEEE Trans. Commun. Syst., Vol. CS-11, June 1963.
- 39. MESSINGER, M. and SHOOMAN, M.L. "Reliability approximations for complex structures", Proc. 1967 Ann. Symp. Reliability IEEE, New York.
- 40. ESARY, J.D. and PROSCHAN, F. "Coherent structures of non-identical components", Technometrics, Vol. 5, No. 2, May, 1963.

- 41. FELLER, W. "An introduction to probability theory and its applications", Vol. I, 3rd Ed., John Wiley and Sons, New York, 1968.
- 42. DEO, N. "Graph theory with applications to engineering and computer science", Prentice-Hall Inc., Englewood Cliffs, N.J., 1974.
- 43. GONZALEZ-URDANETA, G.E. "Reliability of redundant configurations", Power System Report No. 97, Imperial College, London, May 1975.
- 44. CHRISTIAANSE, W.R. "A technique for the analysis of repairable redundant systems", IEEE Trans. on Reliability, Vol. R-19, No. 2, May 1970.
- 45. ELMAGHRABY, S.E. "Activity networks and their generalizations", Lecture notes in operations research and mathematical systems, No. 29, Springer-Verlag, 1970.
- 46. EPSTEIN, B. and SOBEL, M. "Life testing", American Statistical Association Journal, pp. 486-501, September 1953.
- 47. BORG, P.W. ter "Field reporting system for reliability analysis on telecommunication equipment", Proceedings of the NATO conference on reliability testing and reliability evaluation, paper III-A, September 1972.
- 48. DRENICK, R.F. "The failure law of complex equipment", J. Soc. Ind. and Appl. Math., Vol. 8, pp. 680-690, 1960.
- 49. EL-SAYYAD, G.M. "Estimation of the parameter of an exponential distribution", Journal of the Royal Statistical Society, Series B, Vol. 29, pp. 525-532, 1967.
- 50. GNEDENKO, B.V., BELYAYEV, Yu.K. and SOLOVYEV, A.D. "Mathematical methods of reliability theory", Academic Press, 1969.
- 51. NAGELE, G.G. and SELLINSCHEGG, D.W. "Test cost minimization for reliability assessment", Proc. 1972 Annual Symposium on Reliability.

- 52. LLOYD, D.K. and LIPOW, M. "Reliability, management, methods and mathematics", Englewood Cliffs, N.J., Prentice Hall, 1962.
- 53. SATHE, Y.S. and VARDE, S.D. "On minimum variance unbiased estimation of reliability", The Annals of Mathematical Statistics, Vol. 4D, pp. 710-714, April 1969.
- 54. FUGH, E.L. "The best estimate of reliability in the exponential case", Operation Research, Vol. 11, pp. 57-61, 1963.
- 55. KOPAL, Z. "Numerical Analysis, Chapman and Hall, Ltd., 1961.
- 56. MANN, N.R. "A survey and comparison of methods for determining confidence bounds on system reliability from subsystem data", Proceedings of the NATO conference on reliability testing and reliability evaluation, paper I-B, September 1972.
- 57. ROSENBLATT, J.R. "Confidence limits for the reliability of complex systems", Statistical Theory of Reliability (Ed. M. Zelen), The University of Wisconsin Press, Madison, Wisconsin, 1963.
- 58. MURCHLAND, J.D. and WEBER, G.G. "A moment method for the calculation of a confidence interval for the failure probability of a system", Proc. 1972 Annual Reliability and Maintainability Symposium.
- 59. BURNETT, T.L. and WALES, B.A. "System reliability confidence limits", Proc. 7th National Symp. on Reliability and Quality Control, pp. 118-128, 1961.
- 60. LEVY, L.L. and MOORE, A.H. "A Monte Carlo technique for obtaining system reliability confidence limits from component test data", IEEE Trans. on Reliability, Vol. R-16, No. 2, pp. 69-72, September 1967.
- 61. KAMAT, S.J. and RILEY, M.W. "Determination of reliability using eventbased Monte Carlo simulation", Trans. on Reliability, Vol. R-24, No. 1, pp. 73-75, April 1975.

- 62. MEYER, P.L. "Introductory probability and statistical applications", Addison-Wesley Publishing Co. Inc., 1965.
- 63. SINGH, C. "Reliability modelling and evaluation in electric power systems", Ph.D. thesis, University of Saskatchewan, Saskatoon, Canada, August 1972.
- 64. LARSON, H.J. "Introduction to probability theory and statistical inference", Wiley series in Probability and Mathematical Statistics, 1969.
- 65. GONZALEZ-URDANETA, G.E. and CORY, B.J. "A minimal cut-set algorithm for the reliability evaluation of transmission and distribution systems". 3rd Annual Symposium on Uncertainty in Risk and Reliability, University of Bradford, England, 1976.
- 66. SMIRNOV, E.P. "An approach to the evaluation of the reliability of relay protection gear", Electric Technology USSR, Vol. 3, No. 9-10, 1965.
- 67. ZUL', N.M. and KULIEV, F.A. "On the dependability of relay control gear in electrical networks", Electric Technology USSR, Vol. 3, No. 9-10, 1965.
- 68. FABRIKANT, V.L. "Applying reliability theory to appraisal of the performance of relay protection gear", Electric Technology USSR, Vol. 3, No. 9-10, 1965.
- 69. GRIMES, J.D. "On determining the reliability of protective relay systems", IEEE Trans. on Reliability, Vol. R-19, No. 3, August 1970.
- 70. NAYLOR, J.H. "A mathematical model for the prediction of protection reliability" IEE Conference - Developments in Power System Protection, London, March 1975.
- 71. MURRAY, B. and HUGHES, G. "An approach to on-line substation control", Fourth Power Systems Computation Conference, Grenoble, September 1972, Paper 3.2/1.

- 72. ARNOTT, I.A., PITMAN, J.W., BATES, I.P., MILLER, I.H. and SCOTT, A.J.M. "Development of system control facilities for the Victorian transmission network", CIGRE SC31-32 combined meeting, Australia, March 1973, paper D3.
- 73. CONNELLY, K., SCOTT, A.J.M., CAMPAIN, A.J. and MORRISON, I.F. "Integrated protection and control in substations", CIGRE Paper 34-07, August 1974.
- 74. DROMEY, G. and RASMUSSEN, J.P. "Integrated control concepts in power systems", CIGRE Paper 34-08, August 1974.
- 75. LUCKETT, R.G., MUNDAY, P.J. and MURRAY, B.E. "A substation based computer for control and protection", IEE Conference-Developments in power system protection, London March 1975.
- 76. CORY, B.J., DROMEY, G. and MURRAY, B.E. "Digital systems for protection", CIGRE Paper No. 34-08, August 1976.
- 77. GILCREST, G.B., ROCKEFELLER, G.D. and UDREN, E,A. "High speed distance relaying using a digital computer - Part I : System description" IEEE Trans. on Power App. and Systems, Vol. PAS-91, No. 3, 1972.
- 78. GILCREST, G.B., ROCKEFELLER, G.D. and UDREN, E.A. "High speed distance relaying using a digital computer - Part II : Test results" IEEE Trans. on Power App. and Systems, Vol. PAS-91, No. 3, 1972.
- 79. BILLINTON, R. and GROVER, M.S. "Reliability evaluation in distribution and transmission systems", Proc. IEE, Vol. 122, No. 5, May 1975.